

Scuola Internazionale Superiore di Studi Avanzati

A thesis submitted for the degree of Doctor of Philosophy in Geometry and Mathematical Physics

# The geometry of holomorphic submersions, their deformations and moduli

Annamaria Ortu

Supervisors: Prof. Jacopo Stoppa & Dr. Ruadhaí Dervan

Academic Year 2022-23



<sup>6</sup>Ως τοῦ ἀεὶ ὄντοσ γνώσεως, ἀλλὰ οὐ τοῦ ποτέ τι γιγνομένου καὶ ἀπολλυμένου. Εὐομολόγητον, ἔφη, τοῦ γὰρ ἀεὶ ὄντοσ ἡ γεωμετρικὴ γνῶσίς ἐστιν.

"Geometry is the knowledge of that which always is, and not of something which at some time comes into being and passes away." "That is readily admitted, for geometry is the knowledge of the eternally existent."

Plato, Republic, VII, 527

#### ABSTRACT

Proper holomorphic submersions can be viewed as both generalising holomorphic vector bundles and as a way of studying families of smooth projective varieties. We consider submersions whose fibres are analytically K-semistable, thus they each admit a degeneration to a Kähler manifold with constant scalar curvature. On such holomorphic submersions, we introduce and study certain canonical relatively Kähler metrics, called *optimal symplectic connections*, which generalise Hermite-Einstein connections for vector bundles and are defined as solutions to a geometric partial differential equation.

Using optimal symplectic connections, we first give a general construction of extremal metric on the total space, in adiabatic classes, generalising results of Dervan-Sektnan, Fine, Hong. We then construct an analytic moduli space of holomorphic submersions admitting an optimal symplectic connection. To do so, we develop a deformation theory of holomorphic submersions and we combine techniques from geometric invariant theory with the study of the analytic properties of the optimal symplectic connection. We also show that the moduli space is a Hausdorff complex space which admits a Weil-Petersson type Kähler metric.

# Contents

In	Introduction				
1	Bacl	kgroun	d	1	
	1.1 Extremal Kähler metrics			2	
	1.2	Moment maps and GIT-stability			
		1.2.1	Hamiltonian actions	4	
		1.2.2	Geometric Invariant Theory	5	
	1.3	Scalar	curvature as a moment map	7	
		1.3.1	Complexified orbits and Kähler potentials	13	
	1.4	Deform	nation theory of cscK metrics	15	
		1.4.1	Reduction to the finite dimensional problem	17	
	1.5	The m	oduli space of cscK manifolds	19	
2	Holomorphic submersions				
	2.1	1 Splitting of the function space		24	
	2.2	Optimal symplectic connections		25	
		2.2.1	The relatively cscK case	26	
		2.2.2	Optimal symplectic connections in general	27	
	2.3	Deform	mations of fibrations	29	
		2.3.1	Families of holomorphic submersions	32	
		2.3.2	Relative Kuranishi's Theorem	34	
3	Extr	emal m	etrics on the total space	41	
	3.1	Expan	sion of the scalar curvature	41	
	3.2	Linearisation of the fibrewise map $\nu$			
	3.3	Linear	isation of the optimal symplectic connection equation	44	
	3.4 Automorphisms of the optimal symplectic connection equation .			48	
	3.5	Special Kähler metrics on the total space			
		3.5.1	Approximate solutions in the case of discrete automorphism g	roup 51	
		3.5.2	Approximate solutions in the presence of automorphisms	55	
		3.5.3	Solution to the non-linear equation	58	

# CONTENTS

4	The moduli space of holomorphic submersions			
	4.1	Openness of the setting	61	
	4.2	Openness of the space of optimal symplectic connections	66	
	4.3	The moduli space of optimal symplectic connections	73	
	4.4	A Weil-Petersson type Kähler metric	75	
		4.4.1 The determinant line bundle for the Weil-Petersson metric	80	
Bibliography				

#### **AKNOWLEDGMENTS**

First and foremost, I owe many thanks to my supervisors, Ruadhaí Dervan and Jacopo Stoppa. I thank Ruadhaí for patiently guiding me through the project of this thesis and for always explaining with great clarity the underlying motivations behind the various geometric problems I encountered. I thank Jacopo for consistently presenting me with fresh and captivating perspectives on Kähler geometry and my own research and for his constant encouragement and support. I visited Ruadhaí a few times in Cambridge and Glasgow and I thank him for his hospitality and for funding my travels through his Royal Society University Research Fellowship. My short visit to the University of Cambridge, in particular, has been a most exciting and memorable experience, and for that I am truly grateful.

I thank Eloise Hamilton, Masafumi Hattori, Eveline Legendre, Julius Ross, Cristiano Spotti and especially Lars Martin Sektnan and Gábor Székelyhidi, reviewers of this thesis, for many inspiring conversations and for their interest in my work. I thank my academic siblings Michael Hallam and John McCarthy, my fellow PhD students at SISSA and particularly Muhammad Sohaib Khalid for sharing our common interests and perspectives throughout our PhD journeys. I thank Chiara Ballarino for the many days spent learning math together and for sharing with me the most gratifying experience of witnessing the spark of an "a-ha moment" in each other's eyes.

Special thanks go to Carlo for listening to my confused ramblings about things I struggled to understand and for helping me turn them into tangible mathematical concepts, for surviving the pandemic together and for sticking with me from across the ocean.

Lastly, I wish to thank my friend Chiara, whose great enthusiasm has always been a source of inspiration, my friend Alessandra, who walked with me through all stages of education since middle school, and my brother Francesco, who moved to Trieste just at the right time.

# Introduction

A fundamental result in the study of holomorphic vector bundles is the Hitchin-Kobayashi correspondence, which establishes an equivalence between the slopestability of the vector bundle and the existence of a Hermite-Einstein connection. While the former is a purely algebro-geometric notion, the latter is a condition in the form of a geometric PDE involving the curvature of a connection. For vector bundles over a curve, the Hitchin-Kobayashi correspondence is a classical result of Narashiman and Seshadri [61], and it was extended to higher dimensional bases by Donaldson [18], Uhlenbeck and Yau [80], Lübke [52], Kobayashi [48].

Both slope stability and Hermite-Einstein metrics can be used to construct moduli spaces of vector bundles. Seshadri [72] and Newstead [62, 63] gave the first construction of a moduli space of stable vector bundles over a curve, while Mumford [58] introduced semistability in the sense of Geometric Invariant Theory (GIT) to study this moduli space and established its structure as a global GIT quotient. Fujiki and Schumacher [29] later directly constructed the moduli space of Hermite-Einstein vector bundles over a fixed compact Kähler manifold using analytic techniques. These moduli spaces remain a central object of study to this day, and we refer to Greb-Sibley-Toma-Wentworth [33] for recent work containing a discussion of analytic and algebraic compactifications.

Motivated by the Hitchin-Kobayashi correspondence, the Yau-Tian-Donaldson conjecture [81, 79, 21] predicts that an algebro-geometric notion of stability for polarised varieties, *K-stability*, should be equivalent to the existence of Kähler metrics with constant scalar curvature. While still open in full generality, the conjecture is known to be true for Fano varieties, due to Chen-Donaldson-Sun [9, 10, 11]. The fact that the existence of constant scalar curvature Kähler (cscK) metrics implies K-stability is also a theorem of Donaldson [21], Stoppa [74] and Berman-Darvas-Lu [2]. Both K-stability and cscK metrics lead to the existence of moduli spaces. An analytic moduli space of constant scalar curvature Kähler manifolds was constructed by Fujiki and Schumacher [31] in the discrete automorphism case, and extended by Dervan and Naumann in the presence of automorphisms [13] and by Inoue [45] to Fano manifolds with Kähler-Ricci solitons. On the other hand, moduli spaces of K-stable varieties, known as K-moduli, are an active area of research in algebraic geometry [76].

Our work falls into the general framework of studying how these two pic-

#### Introduction

tures interact in the context of proper holomorphic submersions. In this thesis, we focus on the analytic point of view and we study a generalisation of Hermite-Einstein connections on more general fibrations, called *optimal symplectic connections*. Throughout, by a fibration we always mean a proper holomorphic submersion  $\pi_X : (X, H_X) \rightarrow (B, L)$  of a relatively polarised compact Kähler manifold onto a compact polarised base, and we assume that the fibres of  $\pi_X$  are analytically K-semistable. We explore the implications of having an optimal symplectic connection on the existence of special Kähler metrics on the total space and we construct the analytic moduli space of fibrations admitting an optimal symplectic connection. Indeed, one of the main goals of moduli theory besides parametrising certain geometric objects is to study their behaviour in families. From this point of view, our construction of the moduli space of holomorphic submersions is a step towards understanding how projective varieties vary in families.

The easiest and most instructive case to understand the ingredients involved is indeed the case of projectivised vector bundles. Hong [39] related the existence of a solution to the Hermite-Einstein equation on the vector bundle with the existence of a Kähler metric with constant scalar curvature on the total space. More precisely, any Hermitian metric on the vector bundle induces a fibrewise Fubini-Study metric on the projectivisation, and these metrics differ by an automorphism of the fibres. However, if the Hermitian metric satisfies the Hermite-Einstein condition, then it is uniquely determined, so that there is a canonical choice of Fubini-Study metric on the fibres of the projectivisation. This choice allowed Hong to construct constant scalar curvature Kähler metrics on the total space.

In the more general case of a polarised fibration with constant scalar curvature Kähler fibres,  $\pi_X : (X, H_X) \rightarrow (B, L)$ , Dervan and Sektnan [14] introduced optimal symplectic connections as analogous to the Hermite-Einstein connections for projectivised vector bundles. A relatively symplectic form  $\omega$  on X is called a *symplectic connection* in the language of symplectic fibrations because it determines a splitting of the tangent bundle of X into a vertical and a horizontal part, where the horizontal vector bundle is defined using orthogonality with respect to  $\omega$ . If one assumes that the fibres each have a cscK metric, then these metrics can be used to construct a relatively cscK metric  $\omega$  on X, but such an  $\omega$  is not unique if the fibres have non-trivial automorphisms. An optimal symplectic connection is then a *canonical* choice of  $\omega$ , defined in terms of a solution to a second-order elliptic PDE.

We further extend their definition to the following setting. Let  $(Y, H_Y) \rightarrow (B, L)$  be a holomorphic submersion and assume that the fibres are *analytically K*semistable, i.e. they each admit a degeneration to a cscK manifold. We assume also that these degenerations vary holomorphically in *B*, so that we have a degeneration  $(X, \mathcal{H}) \rightarrow (B, L) \times S$  of  $(Y, H_Y) \rightarrow (B, L)$  to a fibrewise cscK fibration  $(X, H_X) \rightarrow$ (B, L) parametrised by  $S \subseteq \mathbb{C}$ . Using a relative version of Ehresmann's theorem (Proposition 2.19) we take the perspective of varying the complex structure of the underlying symplectic fibration, from a relatively cscK complex structure *I* to small compatible deformations  $J_s$  which keep  $\pi$  holomorphic. We say that  $\omega$  is an *optimal symplectic connection* on  $(Y, H_Y)$  if it satisfies the geometric PDE

$$p_E(\Delta_{\mathcal{W}}\left(\Lambda_{\omega_B}(\gamma^* F_{\mathcal{H}})\right) + \Lambda_{\omega_B}\rho_{\mathcal{H}}\right) + \frac{\lambda}{2}\nu = 0.$$
(0.1)

In this expression  $\gamma^* F_{\mathcal{H}}$  and  $\rho_{\mathcal{H}}$  are curvature quantities which depend on  $\omega$ ,  $\nu$  is a curvature quantity that depends on the infinitesimal change in the complex structure and  $\lambda > 0$  is a constant. The left-hand side is a smooth function on Y, and the map  $p_E$  is the projection onto the global sections of the vector bundle  $E \rightarrow B$  of fibrewise holomorphy potentials with respect to the relatively cscK complex structure of X. The vanishing of the first term is the condition for an optimal symplectic connection in the sense of [14], i.e. where all the fibres are cscK, so our notion generalises their notion.

In the following, we consider only integral Kähler classes, although this is not essential. Indeed all our results hold if  $c_1(H_X)$  and  $c_1(L)$  are replaced respectively by a relative Kähler class and a Kähler class that do not come from holomorphic line bundles. Moreover, the base *B* is considered fixed.

# Summary of results

We make use of optimal symplectic connections to prove the existence of cscK and extremal metrics on the total space *Y* in *adiabatic classes* 

$$c_1(H_Y) + kc_1(L) \quad \text{for } k \gg 0.$$

To this end, we need to be able to choose an appropriate metric on the base manifold as follows. Let  $\mathcal{M}^{cscK}$  be the moduli space of cscK manifolds and let  $q : B \to \mathcal{M}^{cscK}$ be the moduli map induced by the central family  $(X, H_X) \to B$ , whose fibres are cscK. The space  $\mathcal{M}^{cscK}$  can be endowed with a *Weil-Petersson type* Kähler metric, and we denote by  $\alpha_{WP}$  the pull-back of it via q. This is a smooth semi-positive (1, 1)-form on B.

We first consider the case where the group of automorphisms of  $(Y, H_Y)$  and of (B, L) which preserve the map q are discrete. Thus we require that the base admits a *twisted cscK metric* with twisting form  $\alpha_{WP}$ :

$$\operatorname{Scal}(\omega_B) - \Lambda_{\omega_B} \alpha_{WP} = constant.$$

**Theorem 0.1.** Assume that the automorphisms of  $(Y, H_Y)$  and of q are discrete. Let  $\omega$  be an optimal symplectic connection and  $\omega_B$  be a twisted cscK metric with twisting  $\alpha_{WP}$ . Then there exists a constant scalar curvature Kähler metric on Y in the class  $c_1(H_Y) + kc_1(L)$  for all  $k \gg 0$ .

If we allow the moduli map q of the central fibration and the total space (Y,  $H_Y$ ) to have automorphisms, the adiabatic limit method produces *extremal metrics* on

#### Introduction

the total space. In this case, we have to modify our hypotheses on  $\omega$  and  $\omega_B$  as follows: we require that  $\omega_B$  is *twisted extremal*, i.e.

$$Scal(\omega_B) - \Lambda_{\omega_B} \alpha_{WP}$$

is a holomorphy potential on *B* and that  $\omega$  is an *extremal symplectic connection*, i.e.

$$p_E\left(\Delta_{\mathcal{V}}(\Lambda_{\omega_B}(\gamma^*F_{\mathcal{H}})) + \Lambda_{\omega_B}\rho_{\mathcal{H}}\right) + \frac{\lambda}{2}\nu$$

is a holomorphy potential on *Y*. We also need some technical assumptions which we will explain in Section 3.5.2: the group of automorphisms of  $\pi_Y$  acts equivariantly on the family  $X \rightarrow B \times S$  and the extremal symplectic connection  $\omega$  is invariant under the flow of the extremal vector fields.

**Theorem 0.2.** Suppose that (B, L) admits a twisted extremal metric  $\omega_B$  and  $(Y, H_Y)$  admits an extremal symplectic connection  $\omega$ . Suppose also that all automorphisms of the moduli map q lift to  $(Y, H_Y)$ . Then there exists an extremal metric on Y in the class  $c_1(H_Y) + kc_1(L)$ for all  $k \gg 0$ .

Our results generalise previous works by many authors who consider more special situations: we already mentioned Hong's paper [39] about cscK metrics on the projectivisation of stable holomorphic vector bundles, in the case of a discrete group of automorphisms. In the presence of automorphisms of the vector bundle, Brönnle [7] proved the existence of extremal metrics on the projectivisation of vector bundles given as direct sums of stable bundles. Fine [24] proved the existence of cscK metrics on the total space of a fibration where all the fibres and the base are Riemann surfaces of genus  $g \ge 2$ . In this case, the choice of a relatively Kähler metric on the total space falls naturally on the hyperbolic metric, and the optimal symplectic connection condition is vacuous. Dervan and Sektnan [14] proved that the optimal symplectic connection condition reduces to the Hermite-Einstein condition on projectivised vector bundles, thus being a genuine generalisation. Moreover, they prove Theorem 0.1 and Theorem 0.2 in the case of a relatively cscK fibration. Similarly, McCarthy [55] proved that on isotrivial fibrations, the optimal symplectic connection condition becomes the Hermite-Yang-Mills condition on an associated principal bundle.

The proof of Theorems 0.1 and 0.2 is carried out using the *adiabatic limit* technique, a strategy which originates in Kähler geometry in the work of Fine [24]. It consists of expanding the scalar curvature of  $\omega + k\omega_B$  in inverse powers of k, with the idea that if k is large the base becomes very large and the curvature is concentrated in the vertical direction. In the easiest case of discrete automorphisms, the optimal symplectic connection condition and the twisted cscK equation on the base allow one to find a relatively Kähler metric which is constant to order  $k^{-1}$ . Then one proceeds inductively, adding at each step r a potential  $i\partial \bar{\partial} \varphi_r$  in order to make the scalar curvature constant up to the  $k^{-r-1}$ -term. The implicit function theorem then allows one to deform the approximate solution to a genuine solution. Our approach is a version of the one just described, except with *two* parameters. We consider a degeneration  $X \to B \times S$  of the fibration  $Y \to B$  to the relatively cscK fibration  $X \to B$  and we expand the scalar curvature of the Kähler metric  $(\omega + k\omega_B, J_s)$  in inverse powers of *k* and powers of *s*. Then we relate the parameters *k* and *s* by imposing  $\lambda k^{-1} = s$ , for some  $\lambda > 0$ .

Our notion of an optimal symplectic connection on a relatively K-semistable fibration should be the most general condition to ask in order to produce cscK or extremal metrics in adiabatic classes, provided all data is smooth and the aforementioned hypotheses on the lifting of the automorphism groups hold.

Allowing K-semistable fibres is essential for the construction of the moduli space of fibrations with an optimal symplectic connection. Indeed, when deforming a fibration with cscK fibres, one cannot expect that the fibres remain cscK. Analytic K-semistability, on the other hand, is an open condition, and this allows us to study the local behaviour of families of fibrations with an optimal symplectic connection. We prove the following result.

**Theorem 0.3.** There exists a moduli space  $\mathcal{M}$  that parametrises holomorphic submersions over a fixed base, with discrete relative automorphism group and which admit an optimal symplectic connection. The space  $\mathcal{M}$  is a Hausdorff complex space which carries a Weil-Petersson type Kähler metric.

We construct the moduli space  $\mathcal{M}$  by gluing local charts around fibrations that admit an optimal symplectic connection. If  $(Y, H_Y) \rightarrow (B, L)$  is such a fibration, the local moduli space around Y is given as the quotient

$$W_Y/\operatorname{Aut}(\pi_Y)$$
,

where  $W_Y$  is the complex space of all the deformations of Y that also admit an optimal symplectic connection, and we quotient by the action of the discrete group Aut( $\pi_Y$ ) of relative automorphisms, which is finite.

We explain in some more detail the definition of  $W_Y$ , which essentially involves two steps. The first step, explained in §4.1, consists of finding a locally closed analytic space which parametrises all small deformations of the complex structure of Y that admit a degeneration to a fibration with cscK fibres. To do so, we use the theory of deformations of cscK manifolds of Székelyhidi [77] and Brönnle [6] to develop a theory of deformations of fibrations. More precisely, we prove a fibration version of Kuranishi's deformation theorem (Theorem 2.21) that allows us to parametrise the compatible vertical deformations of the complex structure of X with a complex space  $V_{\pi}$  of harmonic (0, 1)-forms with values in the (1, 0)-vertical tangent bundle. Working locally in B, we then establish that the deformations of Y which degenerate to a relatively cscK fibration form a locally closed analytic subvariety  $V_{\pi}^+$  of  $V_{\pi}$ . We explicitly construct the relatively cscK degeneration using techniques from GIT; although we do not use it directly, our construction is related to the Byałinicki-Birula decomposition [4, 46]. This is the key new step in our construction, not present in other constructions of moduli spaces.

#### Introduction

The second step, which is the topic of §4.2, consists of proving that, in a small open neighbourhood  $W_Y$  of the point associated to Y in  $V_{\pi}^+$ , all fibrations admit an optimal symplectic connection. The proof of this essentially relies on the implicit function theorem and employs the linearisation of the equation with respect to the complex structure. It is here that we use the assumption on the discreteness of the relative automorphism group.

In the definition of an optimal symplectic connection and in the construction of the space  $W_Y$  it is essential to assume that the connected component of the identity of the groups of automorphisms  $Aut_0(X_b, H_b)$  of the fibres of the relatively cscK degeneration are all isomorphic. This assumption is considered to be a smoothness assumption for our setting and a fixed datum in our construction of the moduli space.

# Outlook

We have presented optimal symplectic connections as canonical choices of relatively Kähler metrics on relatively K-semistable fibrations. To genuinely call them *canonical*, optimal symplectic connections should be proven to be unique. In the relatively cscK case this is a theorem of Dervan, Sektnan [17] and Hallam [34]. We expect that uniqueness holds also in the relatively K-semistable case, up to the action of the group of automorphisms of the projection  $\pi_Y$ .

While our work concentrates on the analytic aspects of optimal symplectic connections and their moduli space, there are different algebro-geometric notions of stability that can be defined on fibrations and related to the existence of optimal symplectic connections. In particular, a fibration version of K-stability was developed by Dervan, Sektnan [16] and further studied by Hallam [34]. They prove that, on projectivised vector bundles, fibration semistability implies slope-semistability of the vector bundle and that the existence of an optimal symplectic connection implies stability, thus establishing first results in the direction of generalising the Hitchin-Kobayashi correspondence. Although they work on fibrations with Kpolystable fibres, their definition of stability also makes sense in the relatively K-semistable case we treat.

It is natural to ask if it is possible to give an algebro-geometric construction of the moduli space of fibrations based on stability. Such a construction would lead to the structure of a variety on the moduli space rather than the structure of a complex space, and would naturally allow singular fibres. In particular, it would parametrise certain stable fibrations which degenerate to a fibration whose general fibre is K-polystable. Moreover, the automorphism group of the general fibre of the degeneration should be fixed.

Hattori later introduced two different notions of stability: f-stability [38], related to Dervan-Sekntan stability of fibrations, and adiabatic K-stability [38, 37]. The latter is a condition defined on a fibration for adiabatic classes and involves also the K-stability of the base. Hashizume and Hattori [36] have constructed a moduli

space of adiabatically K-stable fibrations over a curve where the generic fibre is Calabi-Yau, using algebro-geometric techniques. In the special case where the fibres are all smooth, our moduli space constitutes an alternative construction of Hashizume-Hattori's moduli space.

In the construction of special Kähler metrics on the total space, open questions remain in the presence of singularities. In [25, §9] Fine explains a possible way to construct special Kähler metrics on the total space of holomorphic Lefschetz fibrations, where a finite number of fibres are singular, but the problem is still mostly open. Moreover, a key assumption in Theorem 0.2 is that all automorphisms of the moduli map on the base lift to the total space. When this assumption does not hold, existence results for special Kähler metrics were proved by Hong [40] in the case of projectivised vector bundles and extended by Lu-Seyyedali [51], but the problem is open on a general fibration, and even for projective bundles sharp results are not known.

Finally, Sektnan and Spotti [71] prove a similar result to our Theorem 0.2 on the total space of certain compactified test configurations, where the central fibre is cscK and the general fibre is just K-semistable. Such a test configuration can be viewed as a deformation of a compactified product test configuration for the central fibre. Their proof, however, does not require the extremal symplectic connection condition but requires that the vector bundle *E* of relatively cscK metrics is trivial. It is reasonable to expect that this in fact implies that the extremal symplectic connection condition is satisfied, thus relating the two constructions.

# Outline

We describe briefly the contents of each chapter. In Chapter 1 we give preliminary definitions and results about Kähler geometry. In particular, in §1.1 we collect some basic properties of the scalar curvature equation and in §1.3 we describe the moment map interpretation of the scalar curvature. Then in §1.4 we describe the relevant definitions and results on deformations of a cscK manifold.

Chapter 2 is a description of the Kähler geometry of holomorphic submersion. We describe relative Kähler metrics and the curvature quantities they induce. Then we discuss the notion of an optimal symplectic connection in the relatively cscK case following [14] and we extend it to the relatively K-semistable case. In §2.3 we extend the theory of deformations of a cscK manifold to the fibration setting and we prove a relative version of Kuranishi's Theorem.

In Chapter 3 we prove the existence of a cscK metric on the total space of the relatively K-semistable fibration: we derive the optimal symplectic connection equation by expanding the scalar curvature and we study its linearisation. Then we use the adiabatic limit strategy to prove the existence of cscK and extremal metrics on the total space.

In Chapter 4 we construct the moduli space of holomorphic submersions admitting an optimal symplectic connection. We then describe a Weil-Petersson type

## Introduction

Kähler metric on the moduli space, along with a natural line bundle. Chapters 2 and 3 are part of the author's article [66]. The results of Chapter 4 are contained in the author's preprint [65].

# Chapter 1 Background

In this chapter, we describe the theory of special Kähler metrics and of compatible deformations of the complex structure. We then explain the infinite-dimensional moment map picture for the scalar curvature and a finite-dimensional reduction in the case of constant scalar curvature.

We begin by fixing the notation. Throughout, we always work with *projective* Kähler manifolds, that is smooth projective varieties endowed with an ample line bundle and denoted by (M, L). We call the pair (M, L) a *polarised Kähler manifold*, and the ample line bundle *L* a *polarisation* of *M*. We consider the polarisation to be a fixed datum of the various problems we describe.

Let  $\omega$  be a Kähler form on M in the first Chern class of L and let J be the complex structure of M. We denote by  $g = g(\omega, J)$  the Riemannian metric on M induced by J and  $\omega$ , i.e.

$$g(\cdot, \cdot) = \omega(\cdot, J \cdot).$$

We will often call either the pair  $(\omega, J)$  or the Kähler form  $\omega$  alone a *Kähler metric*.

**Definition 1.1.** The *Ricci curvature* of  $\omega$  is the two-form

$$\operatorname{Ric}(\omega, J) = -\frac{i}{2\pi} \partial_J \bar{\partial}_J \log \omega^n.$$

The *scalar curvature* of the Kähler metric  $(\omega, J)$  is a smooth function on *M* defined as the contraction of the Ricci curvature:

$$\operatorname{Scal}(\omega, J) := \Lambda_{\omega} \operatorname{Ric}(\omega, J).$$

We are interested in *special* Kähler metrics, where the scalar curvature is subject to certain constraints. Among those, we often consider Kähler metrics with constant scalar curvature, where the constant is given by the intersection product

$$\widehat{S} = \frac{n c_1(M) \cdot c_1(L)^{n-1}}{c_1(L)^n}$$

In particular,  $\widehat{S}$  is a *topological constant* fixed by the polarisation.

# **1.1 Extremal Kähler metrics**

In this section, we recall some basic definitions and results on Kähler manifolds and the scalar curvature map; we refer to [78, Chapter 4] for an exhaustive discussion. Let (M, L) be a Kähler manifold and let  $(\omega, J)$  be a Kähler metric. A smooth function h on M is called a *holomorphy potential* if the (1, 0)-part of the Riemannian gradient of h, denoted by  $\nabla_{\alpha}^{1,0}h$ , is a holomorphic vector field.

**Definition 1.2.** A Kähler metric  $(\omega, J)$  on *M* is *extremal* if

$$\bar{\partial} \nabla_g^{1,0} \operatorname{Scal}(\omega, J) = 0,$$

i.e. if its scalar curvature is a *holomorphy potential*.

In particular, constant scalar curvature metrics are extremal. In the study of the existence of extremal and cscK metrics, it is essential to understand the linearisation of the scalar curvature, which can be written in terms of a differential operator called the *Lichnerowicz operator*. When linearising the scalar curvature function, we can either fix the complex structure *J* and vary the Kähler form  $\omega$  or fix  $\omega$  and vary *J*. In this section, we consider the complex structure *J* as fixed, and we describe the linearisation of Scal( $\omega$ , *J*) when we vary  $\omega$  in the cohomology class  $c_1(L)$ . In the next section we will describe the linearisation in the *J*-variable and the relation between the two. To avoid any confusion, in this section we write the scalar curvature as a function of  $\omega$  alone, Scal( $\omega$ ). The set of Kähler metrics in the same Kähler class of  $\omega$  with respect to *J* is

$$\mathcal{K}_{J}(\omega) = \left\{ \omega' \in c_{1}(L) \mid \omega' = \omega + i\partial_{J}\bar{\partial}_{J}\varphi \text{ for some } \varphi \in C^{\infty}(M, \mathbb{R}) \right\}.$$
(1.1)

Fixing a reference Kähler metric  $\omega \in c_1(L)$ , we can then describe the linearisation of  $Scal(\omega)$  in the direction of a Kähler potential  $\varphi$ .

**Definition 1.3.** Let  $\mathcal{D} : C^{\infty}(M, \mathbb{C}) \to \Omega^{0,1}(T^{1,0}M)$  be the operator

$$\mathcal{D}(\varphi) = \bar{\partial} \nabla_{\varphi}^{1,0} \varphi.$$

The Lichnerowicz operator is the composition  $\mathcal{D}^*\mathcal{D}$ , where  $\mathcal{D}^*$  is the adjoint defined with respect to the  $L^2(g)$ -inner product.

It can be written explicitly as follows:

$$\mathcal{D}^*\mathcal{D}(\varphi) = \Delta_g^2(\varphi) + \langle \operatorname{Ric}(\omega), i\partial\bar{\partial}\varphi \rangle + \frac{1}{2} \langle \nabla \operatorname{Scal}(\omega), \nabla \varphi \rangle.$$

The Lichnerowicz operator is a 4th-order elliptic operator. Its kernel, which by compactness is the kernel of  $\mathcal{D}$ , coincides with the space of real holomorphy potentials on M. In particular, it is clear from the definition that  $\omega$  is an extremal metric on M if and only if the scalar curvature of  $\omega$  is in the kernel of  $\mathcal{D}$ . The

#### 1.2 Moment maps and GIT-stability

linearisation of the scalar curvature in the direction of a Kähler potential  $\varphi$  can be written in terms of the Lichnerowicz operator as

$$-\mathcal{D}^*\mathcal{D}(\varphi) + \frac{1}{2} \langle \nabla \text{Scal}(\omega), \nabla \varphi \rangle.$$

In particular, the linearisation at a constant scalar curvature metric is given exactly by the Lichnerowicz operator.

We next describe the linearisation of the scalar curvature at an extremal metric. We denote by  $\overline{\mathfrak{h}}$  the space of holomorphy potentials and by  $\mathfrak{h}_0$  the space of holomorphic vector fields which admit a holomorphy potential. Solving the extremal equation means finding a Kähler metric  $\omega$  such that

$$\operatorname{Scal}(\omega) - f = 0$$

for some holomorphy potential f. If we change  $\omega$  to  $\omega + i\partial \bar{\partial} \varphi$ , then the holomorphy potential f changes to  $f + \frac{1}{2} \langle \nabla f, \nabla \varphi \rangle$ . Therefore, an extremal metric in the Kähler class of  $\omega$  is a zero of the operator

$$C^{\infty}(M, \mathbb{R}) \times \mathfrak{h} \to C^{\infty}(M, \mathbb{R})$$
  
(\varphi, h) \mapsto Scal(\varphi + i\delta\bar{\phi}\varphi) - \frac{1}{2} \lap{\nabla f}, \nabla \varphi \rangle - f. (1.2)

The linearisation  $\mathcal{G}$  of this operator at a solution is given again by the Lichnerowicz operator itself:  $\mathcal{G}(\varphi, 0) = -\mathcal{D}^* \mathcal{D} \varphi$ .

We end this brief overview of extremal and cscK metrics with a description of their automorphism group. Let Aut(M, L) be the group of automorphisms of M which lift to L and let  $\mathfrak{h}$  be its Lie algebra. Let  $Isom(M, \omega)$  be the group of holomorphic isometries of the Kähler metric  $(\omega, J)$  and let  $\mathfrak{t}$  be its Lie algebra. A well-known result of Matsushima and Lichnerowicz states that when  $\omega$  is cscK, the group Aut(M, L) is reductive [54, 50].

**Theorem 1.4.** *Suppose that there exists a constant scalar curvature Kähler metric on M. Then* 

$$\mathfrak{h}_0 = \mathfrak{k}_0 \oplus J\mathfrak{k}_0.$$

The above theorem is known as the *Matsushima criterion* or the *Cartan decomposition*. For an extensive discussion on the interplay between the existence of special Kähler metrics and the groups of automorphisms of the complex, Riemannian and symplectic structure we refer to [32, §3.4].

# **1.2** Moment maps and GIT-stability

In this section, we briefly describe the definition of a Hamiltonian action of a compact Lie group on a symplectic manifold. We explain how moment maps are used to take a quotient of a symplectic manifold with what is called the *symplectic* 

*reduction*. We also describe some features of Geometric Invariant Theory (GIT), which also allows us to take a quotient of a projective variety with respect to the action of a complex Lie group. The Kempf-Ness Theorem 1.12 relates the two constructions.

### 1.2.1 Hamiltonian actions

Let  $(M, \omega)$  be a symplectic manifold.

**Definition 1.5.** A vector field  $\eta$  is *Hamiltonian* with respect to  $\omega$  if there exists a function  $h \in C^{\infty}(M, \mathbb{R})$  such that

$$\omega(\eta, \cdot) = -\mathrm{d}h.$$

We say that *h* is the *Hamiltonian function* of  $\eta$ .

On a Kähler manifold  $\eta = J\nabla_g(h)$ , where  $\nabla_g h$  is the Riemannian gradient of h. A Hamiltonian vector field with Hamiltonian h is also called the *symplectic gradient* of h, and denoted by grad<sup> $\omega$ </sup>h.

Let *G* be a Lie group that acts on *M*, and assume that the action preserves the symplectic form, i.e. for any  $g \in G$ ,  $g^*\omega = \omega$ . Let g be the Lie algebra of *G*. For any element  $f \in g$ , the *infinitesimal action* of *f* is the vector field

$$\sigma_x(f) = \left. \frac{\mathrm{d}}{\mathrm{d}t} \right|_{t=0} \left( \exp(-tf) \cdot x \right)$$

**Definition 1.6.** Let *G* act on  $(M, \omega)$  by means of symplectomorphisms. We say that the action is Hamiltonian if there exists a *moment map* 

$$\mu: M \to \mathfrak{g}^*$$

that is equivariant with respect to the *G*-action on *M* and the co-adjoint *G*-action on the dual Lie algebra  $g^*$  and such that for each  $x \in M$ 

$$d_x\langle \mu, f \rangle = \omega(-, \sigma_x(f)),$$

i.e.  $\langle \mu, f \rangle$  is a Hamiltonian function for the vector field  $\sigma_x(f)$  on *M*.

It is clear from the definition of a Hamiltonian vector field that the Hamiltonian function is only unique up to a constant. The moment map then chooses a Hamiltonian function for the infinitesimal vector field.

An important feature of Hamiltonian actions is that they allow us to take a symplectic quotient of a projective manifold. Assume that *K* is a compact Lie group that acts on *M* and that the action is Hamiltonian. Since the moment map is equivariant, its level sets that are preserved by the co-adjoint action are also preserved by the group action. Moreover, the origin of  $t^*$  is always fixed by the coadjoint action.

The symplectic quotient is defined as

$$\mu^{-1}(0)/K.$$

This orbit space was first considered by Marsden and Weinstein [53] and Meyer [57]. In fact, it is a theorem of Marsden and Weinstein that if the action on  $\mu^{-1}(0)$  is free and proper the symplectic quotient carries a symplectic form. More generally, there is a stratified symplectic structure on  $\mu^{-1}(0)/K$  [73] such that each leaf is a smooth symplectic manifold. The symplectic quotient is also called the *symplectic reduction* of *M* by the action of *K*, and denoted by  $M // ^{\text{red}} K$ .

## 1.2.2 Geometric Invariant Theory

In this section, we briefly review some basic notions in Geometric Invariant Theory (GIT). We refer to the books of Mumford-Fogarty-Kirwan [59] and of Newstead [63] for extensive discussions and details. The main goal of Geometric Invariant Theory that we describe is to take a geometric quotient of a projective variety M with respect the action of a complex Lie group G, such that the quotient is again a projective variety. To do so, GIT introduces a notion of *stability* for the points of M and defines a quotient of stable orbits.

Let  $M \subseteq \mathbb{CP}^d$  be a smooth projective variety. Let *G* be a complex reductive Lie group acting on *M* as a subgroup of  $SL(d + 1, \mathbb{C})$ , so that the projective embedding of *M* is *G*-equivariant. In particular, the action of *G* on  $\mathbb{CP}^d$  lifts to an action on the line bundle O(-1) and it restricts to an action on the affine cone  $\widehat{M}$  of *M*. This lift of the action is called a *linearisation* of the action, and it determines the definition of stability and of the GIT quotient, as we now explain.

Let  $I_M$  be the homogeneous ideal of  $\mathbb{C}[z_0, \ldots, z_d]$  defining M. Then the homogeneous coordinate ring of M is the graded ring

$$R(M) = \mathbb{C}[z_0, \ldots, z_d]/I_M = \bigoplus_r H^0(M, O_M(r)).$$

Let  $R(M)^G$  be the graded ring of *G*-invariant sections

$$R(M)^G = \bigoplus_r H^0(M, O_M(r))^G$$

Nagata's theorem [60] (see also [63, Theorem 3.4]) guarantees that, since *G* is reductive,  $R(M)^G$  is finitely generated. The GIT quotient of *M* by *G* is defined as

$$M \not / G := \operatorname{Proj} R(M)^G$$

To understand the definition more geometrically, we introduce the notion of GIT stability.

**Definition 1.7.** Let  $x \in M$  and  $\hat{x} \in O_M(1)$  be a lift of x. We say that is

- 1. *semistable* if there exists a non-constant *G*-invariant homogeneous section *s* in  $H^0(M, O_M(r))^G$  such that  $s(x) \neq 0$ ;
- 2. *polystable* if it is semistable and the orbit  $G \cdot \hat{x}$  is closed;
- 3. *stable* if it is semistable, its stabiliser is finite and the orbit  $G \cdot \hat{x}$  is closed.

A point that is not semistable is called *unstable*. The set of semistable points is denoted by  $M^{ss}$ .

In particular, the GIT quotient is the image of *M* under the birational map

$$M \dashrightarrow \mathbb{P}\left(H^0(M, O_M(r))^G\right)^*$$

for  $r \gg 0$ , and semistable points are the ones where the map is actually defined. We can then view the GIT quotient as parametrising the semistable orbits of the action of *G* on *M*. More precisely, we can think of  $M /\!\!/ G$  as the quotient of the  $M^{ss}$  by the equivalence relation that identifies two points if and only if the closures of their orbits have non-empty intersection. The GIT quotient then may identify strictly semistable orbits, while it is a geometric quotient on stable orbits. We have the following uniqueness result for polystable points.

**Lemma 1.8** ([43, Corollary 5.13]). Let G be a reductive group acting linearly on  $M \subseteq \mathbb{CP}^d$ and let  $x \in M$  be a semistable point. Then the closure of the orbit  $G \cdot x$  contains a unique polystable orbit.

Although we have given all the definitions for a projective variety, we can take a GIT quotient also of an affine variety [63, §3] [43, §4.5, §4.6]. In this case, we give the following definition of stability that incorporates the choice of a standard linearisation.

**Definition 1.9.** Let  $\mathbb{A}^d$  be an affine space of dimension d and G a reductive affine group acting on it. Let  $(z_1, \ldots, z_d)$  be a system of coordinates on  $\mathbb{A}^d$ . The space  $\mathbb{A}^d$  can be embedded in the projective space  $\mathbb{P}^d$  as a coordinate chart with the map

$$(z_1,\ldots,z_d)\mapsto [1:z_1:\ldots:z_d].$$

We extend the action of *G* to an action on  $\mathbb{P}^d$  by acting trivially on the first coordinate. Let  $x \in \mathbb{A}^d$  and let  $\hat{x}$  be its image in  $\mathbb{P}^d$ . We say that *x* is *semistable*, *polystable* or *stable* if it is with respect to the trivial linearisation.

In particular, the polynomial  $P(\mathbf{z}) = z_0$  is a *G*-invariant homogeneous polynomial that does not vanish at any point of  $\mathbb{A}^d$ . So every point of  $\mathbb{A}^d$  is semistable.

*Remark* 1.10. The fact that every point is semistable holds because we defined the action on  $\mathbb{P}^d$  to be trivial on the first homogenous coordinate. In principle, one can choose to extend the action in a non-trivial way and can still define stability as above, but unstable points might appear. We do not treat this case here, so we have included the trivial extension of the action in the definition of stability.

#### 1.3 Scalar curvature as a moment map

The Hilbert-Mumford criterion [59, Theorem 2.1] is a useful way of establishing the stability of a point by looking at the 1-parameter subgroups of *G*. If 1-parameter subgroup  $\rho(t) : \mathbb{C}^* \subset G$  acts on a point *x*, then  $\lim_{t\to 0} \rho(t) \cdot x$  is a fixed point for the action of  $\rho(t)$ , so  $\rho(t)$  acts on the line  $O_x(-1)$  that *x* represents. Moreover,  $\rho(t)$  acts on the line by multiplication for  $t^a$ , where  $a = a(\rho, x)$  is the weight of the action.

**Theorem 1.11** (Hilbert-Mumford criterion). *A point*  $x \in M$  *is* 

- 1. semistable if and only if  $a(\rho, x) \leq 0$  for all 1-parameter subgroups  $\rho(t)$ ;
- 2. polystable if and only if  $a(\rho, x) \le 0$  for all 1-parameter subgroups  $\rho(t)$  and  $a(\rho, x) = 0$  holds if and only if  $\lim_{t\to 0} \rho(t) \cdot x$  does not lie in the orbit  $G \cdot x$ ;
- *3. stable if and only if*  $a(\rho, x) < 0$  *for all 1-parameter subgroups*  $\rho(t)$ *.*

The GIT quotient is intimately related to the symplectic quotient and the theory of moment maps by the Kempf-Ness theorem [47], [59, Theorem 8.3]. Let  $(M, O_M(1))$  be a polarised variety with a Kähler form  $\omega \in c_1(O_M(1))$ . Let *G* be a reductive Lie group acting on *M* such that the action of *G* lifts on  $O_M(1)$ . Let *K* be a maximal compact subgroup such that the action on *M* is Hamiltonian.

**Theorem 1.12** (Kempf-Ness). *There exists a moment map*  $\mu$  *for the K-action on M such that:* 

- 1. a G-orbit is semistable if and only if its closure contains a zero of the moment map; this zero is in the unique polystable orbit in the closure of the semistable orbit;
- 2. the inclusion of  $\mu^{-1}(0)$  in  $M^{ss}$  induces a homeomorphism

$$M /\!\!/ G \simeq \mu^{-1}(0)/K.$$

The Kempf-Ness theorem is one of the first instances of what has become a guiding principle in the study of many geometric problems, including the one treated in this thesis, and it serves as a motivation for studying similar problems in the infinite-dimensional setting. When the group and the variety are infinite-dimensional, a moment map is often a differential operator, so finding a zero of the moment map means finding a solution of a geometric PDE. Conversely, proving that an equation is a moment map for the action of a group is then the first step towards establishing a relation with an algebro-geometric stability condition.

# **1.3** Scalar curvature as a moment map

In this section, we focus on the moment map interpretation of the scalar curvature, due to Fujiki [28] and Donaldson [19]. For details and proofs see also [68, Chapter 1]. Indeed, the scalar curvature function can be viewed as a moment map for the

action of an infinite-dimensional Lie group on an infinite-dimensional space, as we now describe.

Let  $(M, \omega)$  be a symplectic manifold, and consider the infinite-dimensional space of compatible almost complex structures:

$$\mathscr{J} = \{J : TM \to TM \text{ almost complex structure compatible with } \omega\}.$$

The tangent space at a point *J* is given by

$$T_J \mathcal{J} = \{A: TM \to TM \mid JA + AJ = 0 \text{ and } \omega(u, Av) + \omega(Au, v) = 0\}.$$

Fix now  $J \in \mathcal{J}$  integrable complex structure and  $A \in T_J \mathcal{J}$ . Consider the Riemannian metric  $g_J$  on M induced by J and  $\omega$  by  $g_J(\cdot, \cdot) = \omega(\cdot, J \cdot)$ . Using the symmetry of  $g_J$  we have

$$g_I(Au, v) = \omega(Au, Jv) = -\omega(u, AJv) = \omega(u, JAv) = g_I(u, Av),$$

so the bilinear form  $(u, v) \mapsto g_J(Au, v)$  is symmetric. Moreover, since AJ + JA = 0, *A* maps  $T^{1,0}M$  to  $T^{0,1}M$  and  $T^{0,1}M$  to  $T^{1,0}M$ , where the splitting is considered with respect to *J*. Since *A* is real, it is uniquely determined by one of the two restrictions, and we take  $A : T^{0,1} \to T^{1,0}$ . This means that we may identify

$$T_J \mathscr{J} \longleftrightarrow T_J^{0,1} \mathscr{J} = \left\{ \alpha \in \Omega^{0,1}(T^{1,0}M) \mid \omega(\alpha(u), v) + \omega(u, \alpha(x)) = 0 \right\}.$$

Now, if  $A \in T_J \mathscr{J}$ , also  $JA \in T_J \mathscr{J}$ , so  $\mathscr{J}$  has a complex structure, which we denote by  $\mathbb{J}$ . Moreover, it has a Hermitian inner product

$$\langle A,B\rangle_J:=\int_M\langle A,B\rangle_{g_J}\frac{\omega^n}{n!}$$

and the two combine to give a Kähler form, given at the point *J* by

$$\mathbf{\Omega}_{I}(A,B) = \langle JA,B\rangle_{I}.$$

So  $\mathscr{J}$  is an infinite-dimensional Kähler manifold. Inside  $\mathscr{J}$ , we consider the complex subspace  $\mathscr{J}^{\text{int}}$  of *integrable* almost complex structures of  $\mathscr{J}$ . Its tangent space is given by those  $\alpha \in T_J^{0,1} \mathscr{J}$  such that  $\overline{\partial} \alpha = 0$ .

Consider the group of Hamiltonian symplectomorphisms of  $(M, \omega)$ , denoted by  $\mathscr{G}$ . This is the infinite-dimensional Lie group of time-one flows of Hamiltonian vector fields on M, and it acts on  $\mathscr{J}$  by pull-back:

$$J \in \mathcal{J}, \phi \in \mathcal{G} \qquad \phi^* J := \mathrm{d}\phi^{-1} \circ J \circ \mathrm{d}\phi.$$

**Lemma 1.13.** The Lie algebra of  $\mathscr{G}$  can be identified with the space  $C_0^{\infty}(M)$  of the smooth functions on M with  $\omega$ -average zero.

*Proof.* Consider a 1-parameter subgroup  $\{\phi_t\}$  of  $\mathscr{G}$ . Then  $\phi_t^* \omega = \omega$ . So, denoting by  $\eta_t$  the vector field which time-one flow is the symplectomorphism  $\phi_t$ , we have

$$\mathcal{L}_{n_t}\omega=0.$$

By the Cartan magic formula,  $d\omega(\eta_t, \cdot) = 0$ . Therefore,  $\eta_t$  is Hamiltonian vector field with some Hamiltonian function *h*. We can assume that *h* has mean-value zero in *M*, as the Hamiltonian function is unique up to a constant. So Lie( $\mathscr{G}$ ) =  $C_0^{\infty}(M)$ .

The following theorem is due to Fujiki [28] and Donaldson [19].

**Theorem 1.14.** The action of  $\mathcal{G}$  on  $\mathcal{J}$  is Hamiltonian with moment map

$$\mu: \mathscr{J} \longrightarrow \operatorname{Lie}(\mathscr{G})^*$$

$$J \longmapsto \operatorname{Scal}(\omega, J) - \widehat{S}.$$
(1.3)

If *J* is integrable,  $\text{Scal}(\omega, J)$  is the scalar curvature of the metric  $g_J$ . Otherwise, it is the Hermitian scalar curvature of the Chern connection on *TM*, which is not the same as the Levi-Civita connection in general. In particular cscK metrics on *M* correspond to  $J \in \mathscr{J}^{\text{int}}$  such that  $\mu(J) = 0$ . The function  $\text{Scal}(J) - \widehat{S}$  is viewed as an element of  $C_0^{\infty}(M)^*$  by identifying  $\text{Lie}(\mathscr{G})^*$  with its dual via the  $L^2(\omega)$ -product on *M*, i.e.

$$\phi \mapsto \langle \operatorname{Scal}(\omega, J) - \widehat{S}, \phi \rangle_{L^2}.$$

We next introduce two operators: the infinitesimal action of  $\mathcal{G}$ , denoted P, and the differential of the scalar curvature, denoted Q. Consider the scalar curvature map with respect to the complex structure:

$$S: \mathscr{J} \longrightarrow C_0^{\infty}(X)$$
  
$$J \mapsto \operatorname{Scal}(\omega, J) - \widehat{S}.$$
 (1.4)

For fixed  $J \in \mathcal{J}$ , the infinitesimal action of  $\mathcal{G}$  on  $\mathcal{J}$  is given by the operator

$$P: C_0^{\infty}(M) \longrightarrow T_J \mathscr{J}$$
$$h \longmapsto \mathcal{L}_{\eta_h} J,$$

where  $\eta_h$  is the Hamiltonian vector field with Hamiltonian function *h*. Let *Q* be the derivative at *J* of the map (1.4), i.e.

$$Q: T_{J} \mathscr{J} \longrightarrow C_{0}^{\infty}(M)$$

$$A \longmapsto d_{J} \mathcal{S}(A).$$
(1.5)

We next show that the operators P and Q are adjoint and how they are used to prove that the map (1.4) is a moment map. These operators and their properties

play a central role throughout this work. In particular, the operator P is used also in §1.3.1 to define a notion of complexified orbits of the action of the group  $\mathscr{G}$  on  $\mathscr{J}$ , even if  $\mathscr{G}$  does not admit a genuine complexification. In §1.4 we use again the operator P to describe deformations of cscK manifolds and a version of Luna's slice theorem. We will introduce a relative version of P in §2.3.2 and we will use it to describe deformations of holomorphic submersions with a fixed base.

We begin by giving an alternative definition of *P*.

Lemma 1.15. The operator P can be written as

$$P(h) = 2J\bar{\partial}\eta_h^{1,0} + 2J\overline{(\bar{\partial}\eta_h^{1,0})},$$
(1.6)

where  $\eta_h$  is the Hamiltonian vector field with Hamiltonian function h.

*Proof.* Recall that, by definition of the Lie derivative, for any vector field  $\xi$ 

$$(\mathcal{L}_{\eta_h}J)(\xi) = [\eta_h, J\xi] - J[\eta_h, \xi].$$

In coordinates, we write  $\xi$  as  $\xi^c \partial_c + \xi^{\bar{d}} \partial_{\bar{d}}$ , and we have

$$\mathcal{L}_{\xi}J(\partial_{a}) = \underbrace{\xi(i\partial_{a})}_{=0} - i\partial_{a}(\xi) - J\underbrace{(\xi(\partial_{a}))}_{=0} + J(\partial_{a}(\xi)) = -2i\left(\partial_{a}\xi^{d}\right)\partial_{\bar{d}},$$

and analogously  $\mathcal{L}_{\xi} J(\partial_{\bar{b}}) = 2i \left( \partial_{\bar{b}} \xi^c \right) \partial_c$ . Plugging in the expression for the Hamiltonian vector field  $\eta_h = \omega^{c\bar{d}} \partial_{\bar{d}} h \partial_c + \omega^{c\bar{d}} \partial_c h \partial_{\bar{d}}$  we obtain

$$\mathcal{L}_{\eta_h} J(\partial_a) = -2i\partial_a (\omega^{cd}\partial_c h)\partial_{\bar{d}},$$
$$\mathcal{L}_{\eta_h} J(\partial_{\bar{b}}) = 2i\partial_{\bar{b}} (\omega^{c\bar{d}}\partial_{\bar{d}} h)\partial_c.$$

Thus

$$\mathcal{L}_{\eta_h}J = 2J\partial_{\bar{b}}(\omega^{cd}\partial_{\bar{d}}h)\partial_c \otimes \mathrm{d}\bar{z}^b + 2J\partial_a(\omega^{cd}\partial_c h)\partial_{\bar{d}} \otimes \mathrm{d}z^a.$$

On the other hand  $\bar{\partial}\eta_h^{1,0} = \partial_{\bar{b}}(\omega^{c\bar{d}}\partial_{\bar{d}}h)\partial_c \otimes d\bar{z}^b$ , so

$$\mathcal{L}_{\eta_h}J=2J\bar{\partial}\eta_h^{1,0}+\overline{2J\bar{\partial}\eta_h^{1,0}},$$

which proves the statement.

In light of Lemma 1.15, we can and will modify our operator *P* and consider as its definition the following:

$$P: C_0^{\infty}(M) \longrightarrow T_J^{1,0} \mathscr{J}$$
$$h \longmapsto \bar{\partial} \eta_L^{1,0}.$$
(1.7)

Moreover, if *J* is integrable,  $J \mathcal{L}_{\eta_h} J = \mathcal{L}_{J\eta_h} J$ , so

$$\bar{\partial}\eta_h^{1,0} + \overline{(\bar{\partial}\eta_h^{1,0})} = -2J\mathcal{L}_{\eta_h}J = -2\mathcal{L}_{J\eta_h}J.$$

*Remark* 1.16. Fix  $J \in \mathscr{J}$  integrable complex structure. Let  $h \in C_0^{\infty}(M)$  and  $\eta_h$  be the associated Hamiltonian vector field. Using the definition of a Hamiltonian vector field, the Cartan magic formula and the relation  $\omega(Ju, v) + \omega(u, Jv) = 0$ , we have

$$\mathcal{L}_{J\eta_h}\omega = d(\omega(J\eta_h, \cdot)) = -d(\omega(\eta_h, J\cdot)) = -d(-dh(J\cdot)) = d(Jdh).$$

If *J* is integrable, by writing the expression in local coordinates, we obtain that

$$\mathcal{L}_{J\eta_h}\omega = \mathrm{d}(J\mathrm{d}h) = -2i\partial\bar{\partial}h. \tag{1.8}$$

The fact that (1.3) is a moment map means by definition that the  $L^2$ -inner product  $\langle \mu(J), \phi \rangle$  is a Hamiltonian function for the infinitesimal vector field  $\mathcal{L}_{\eta\phi}J$ , i.e.

$$\frac{\mathrm{d}}{\mathrm{d}t}\Big|_{t=0} \langle \mu, \phi \rangle (J_t) = \mathbf{\Omega}_{J_0}(\dot{J_0}, \mathcal{L}_{\eta\phi}J_0).$$

By writing the moment map condition in terms of *P*, *Q*, for  $A \in T_J \mathscr{J}$ , we obtain the following relation:

$$\langle Q(A),\phi\rangle_{L^2} = -\mathbf{\Omega}_J\left(\frac{1}{2}JP(\phi),A\right) = \mathbf{\Omega}_J\left(A,\frac{1}{2}JP(\phi)\right) = \langle JA,\frac{1}{2}JP(\phi)\rangle_J = \frac{1}{2}\langle A,P(\phi)\rangle_J.$$

Therefore  $Q^* = \frac{1}{2}P$ , and conversely  $P^* = 2Q$ . The operators Q and P are thus adjoint.

*Remark* 1.17. Let  $J_1$ ,  $J_2$  be two almost complex structures compatible with  $\omega$ , and assume that  $J_2 = f^*J_1$  for some  $f \in \text{Diff}(M)$ . Denoting by  $g(J_i, \omega)$ , i = 1, 2 the corresponding Riemannian metrics, we have

$$g(J_2, \omega) = f^* g(J_1, (f^{-1})^* \omega).$$

If moreover  $f \in \mathcal{G}$ , we have that  $g(J_1, \omega)$  and  $g(J_2, \omega)$  are isometric.

Let  $\mathcal{D}^*\mathcal{D}$  be the Lichnerowicz operator with respect to the metric  $g_J$ . Its coordinates expression is [78, Theorem 4.2]

$$\mathcal{D}^*\mathcal{D}(\phi) = g_J^{d\bar{b}} g_J^{a\bar{c}} \nabla_d \nabla_a \nabla_{\bar{c}} \nabla_{\bar{b}} \phi = g_J^{d\bar{b}} g_J^{a\bar{c}} \nabla_a \nabla_d \nabla_{\bar{c}} \nabla_{\bar{b}} \phi.$$

We use this expression to relate the operators *P* and *Q*. With the following result we prove that along the direction given by the infinitesimal action of  $\mathscr{G}$  on  $\mathscr{J}$ , the derivative of the scalar curvature S is the real part of the Lichenrowicz operator.

**Lemma 1.18** ([20]). *The maps P and Q satisfy the following property:* 

$$Q(P(\phi)) = \operatorname{Re}(\mathcal{D}^*\mathcal{D}\phi). \tag{1.9}$$

*Proof.* Start by writing *Q* as the composition of two operators,

$$Q_1: T_J \mathscr{J} \subset \Omega^{0,1}(T^{1,0}M) \to \Omega^{1,0}(M)$$

and

$$Q_2: \Omega^{1,0}(M) \to C^{\infty}(M).$$

The operator  $Q_1$  is defined as

$$Q_1(\alpha) = \operatorname{Re}(\nabla_a \alpha^a_{\ \bar{b}} \mathrm{d}\bar{z}^b),$$

for  $\alpha \in T_I^{0,1} \mathscr{J}$ . As for  $Q_2$ , let  $\theta = \theta_a dz^a \in \Omega^{1,0}(M)$ . Then

$$Q_2(\theta) = -\operatorname{div}(J\theta^{\sharp}) = \operatorname{div}(ig^{ab}\theta_a\partial_{\bar{b}})$$

where  $\theta^{\sharp}$  is the vector field obtained by  $\theta$  by raising the index with the metric *g*. Moreover,  $Q_2$  satisfies the equation

$$Q_2(\theta)\omega^n = n\mathrm{d}\theta \wedge \omega^{n-1}.$$

We have that

$$Q(A) = Q_2(Q_1(A)) = Q_2\left(\operatorname{Re}(\nabla_a A^a_{\ \bar{b}} \mathrm{d}\bar{z}^b)\right) = -\operatorname{div}\left(\operatorname{JRe}(g^{c\bar{b}}\nabla_a A^a_{\ \bar{b}}\partial_c)\right)$$
$$= -\operatorname{div}\left(\operatorname{Re}(ig^{c\bar{b}}\nabla_a A^a_{\ \bar{b}}\partial_c)\right) = \operatorname{div}\left(\operatorname{Im}(g^{c\bar{b}}\nabla_a A^a_{\ \bar{b}}\partial_c)\right) \qquad (1.10)$$
$$= \operatorname{Im}\left(\nabla_c(g^{c\bar{b}}\nabla_a A^a_{\ \bar{b}})\right).$$

Let us compose this with the operator

$$P(\phi)^{1,0} = \bar{\partial} \left(\xi_{\phi}\right)^{1,0} = \partial_{\bar{c}} \left(\omega^{a\bar{b}}\partial_{\bar{b}}\phi\right) \mathrm{d}\bar{z}^{c} \otimes \partial_{a}.$$

Hence we obtain

$$Q(P(\phi)) = Q\left(\partial_{\bar{c}}\left(\omega^{a\bar{b}}\partial_{\bar{b}}\phi\right)d\bar{z}^{c}\otimes\partial_{a}\right)$$
$$= Q\left(ig^{a\bar{b}}\nabla_{\bar{c}}\nabla_{\bar{b}}\phi d\bar{z}^{c}\otimes\partial_{a}\right)$$
$$= \operatorname{Im}\left(i\nabla_{d}\left(g^{d\bar{b}}\nabla_{a}g^{a\bar{c}}\nabla_{\bar{b}}\nabla_{\bar{c}}\phi\right)\right)$$
$$= \operatorname{Re}(\mathcal{D}^{*}\mathcal{D}\phi),$$

``

as claimed.

We can compare the linearisation (1.9) of the scalar curvature map S, where  $\omega$ is fixed, to the linearisation of the scalar curvature map described in §1.1, where the complex structure is fixed and the Kähler potential varies. When  $(\omega, J)$  has constant scalar curvature, the linearisation of the scalar curvature coincides with the Lichenrowich operator, which is real. Thus it coincides with the expression (1.9). However, in the proof of (1.9) we do not use any hypothesis on the scalar curvature being constant, so when the reference metric  $g(\omega, J)$  has non-constant scalar curvature the two linearisations may be different. In the next section, we will expand on the interplay between fixing the complex structure and varying the Kähler potential and fixing the symplectic form and varying the complex structure.

### 1.3.1 Complexified orbits and Kähler potentials

The group  $\mathscr{G}$  does not admit a formal complexification. Nonetheless, there is a notion of complexified orbits for the action of  $\mathscr{G}$  on  $\mathscr{J}$ , at the infinitesimal level. These orbits play a role in the interaction between the complex structure and its deformations and the changes in the Kähler metric within its Kähler class.

**Lemma 1.19.** Let  $J \in \mathcal{J}$  and let  $O_I$  be the orbit of J for the action of  $\mathcal{G}$ . Then

$$T_I O_I = \{ JP(h) \mid h \in C_0^\infty(M) \}.$$

Lemma 1.19 is a consequence of Remark 1.15, where *P* is related to the infinitesimal action of  $\mathscr{G}$ . In order to define a formal complexification of  $\mathscr{G}$ , we proceed by complexifying the tangent space to the orbits. Indeed, we can consider the complexification of the Lie algebra of  $\mathscr{G}$ , i.e.  $C_0^{\infty}(M, \mathbb{C})$ . Thus we can complexify the infinitesimal action *P* to the operator

$$P^{\mathbb{C}}: C_0^{\infty}(M, \mathbb{C}) \longrightarrow T_I \mathscr{J}$$
(1.11)

defined as follows: if  $h = u + iv \in C_0^{\infty}(M, \mathbb{C})$ , then

 $P^{\mathbb{C}}(h) = P(u) + JP(v).$ 

The complexified infinitesimal action defines for every  $J \in \mathscr{J}$  the set

$$\mathcal{D}_I = \{ (P(h), JP(h)) \mid h \in \operatorname{Lie}(\mathscr{G}) \}.$$

**Lemma 1.20.** 1. For every almost complex structure J,  $\mathcal{D}_J$  is a subset of  $T_J \mathscr{J}$ , hence  $\mathscr{D}$  is a distribution;

2. The distribution  $\mathcal{D}$  is integrable.

For a proof of this result in the case of *J* integrable complex structure, see [79, Chapter 4].

**Definition 1.21.** The leaves of the foliation  $\mathcal{D}$  are defined to be the tangent spaces to the *complexified orbits* of  $\mathscr{G}$  or  $\mathscr{G}^c$ -orbits.

We next explain that the complexified orbits can be interpreted in terms of Kähler metrics in a fixed Kähler class. Let *J* be an integrable compatible complex structure on  $(M, \omega)$ , and consider the space  $\mathcal{K}_J(\omega)$  of Kähler potentials in the class  $[\omega]$  (1.1). The following rather technical proposition of Donaldson [20, p.17] is the key property in the interpretation of the relation between  $\mathscr{G}^c$ -orbits and  $K_J(\omega)$ . We will extend it to the setting of fibrations in Proposition 4.5.

**Proposition 1.22.** For every  $\omega_{\phi} \in \mathcal{K}_{J}(\omega)$  there exist  $f \in \text{Diff}_{0}(M)$  such that  $f^{*}\omega_{\phi} = \omega$  and  $(M, \omega_{\phi}, J)$  is isomorphic to  $(M, \omega, f^{*}J)$ .

*Proof.* Consider *J* fixed, and pick a Kähler metric in the class  $[\omega]$ :

$$\omega_{\phi} = \omega + 2i\partial\bar{\partial}\phi.$$

Using the relation (1.8), a path  $\omega_t$  between  $\omega$  and  $\omega_{\phi}$  for  $t \in [0, 1]$  can be written as

$$\omega_t = \omega + 2it\partial\bar{\partial}\phi = \omega - td(Jd\phi).$$

For each  $t \in [0, 1]$  let  $\eta_t$  be the Hamiltonian vector field

$$\eta_t = \operatorname{grad}^{\omega_t} \dot{\phi}_t.$$

Consider the vector field  $\xi_t = J\eta_t$ . Then

$$\frac{\mathrm{d}}{\mathrm{d}t}\omega_t = -\mathrm{d}(J\mathrm{d}\phi) = -\mathcal{L}_{\xi_t}\omega_t, \qquad (1.12)$$

where the second equality is given again by (1.8). Define  $\{f_t, t \in [0, 1]\}$  to be the *isotopy* of the time-dependent vector field  $\xi_t$ , i.e.

$$\frac{\mathrm{d}}{\mathrm{d}t}f_t = \xi_t(f_t), \quad f_0 = \mathrm{id}.$$

The following result is a standard application of Moser's trick in symplectic geometry [8, 6.4]: for a smooth family  $\eta_t$  of *p*-forms

$$\frac{\mathrm{d}}{\mathrm{d}t}f_t^*\eta_t = f_t^*\left(\mathcal{L}_{\xi_t}\eta_t + \frac{\mathrm{d}\eta_t}{\mathrm{d}t}\right). \tag{1.13}$$

Applying the relations (1.13) and (1.12) to  $\omega_t$  lead

$$\frac{\mathrm{d}}{\mathrm{d}t}f_t^*\omega_t = 0$$

which implies that  $f_t^* \omega_t = f_0^* \omega = \omega$ . Let  $J_t$  be the pull-back  $f_t^* J$ . Then, for t = 1, the two Riemannian metrics  $g(\omega, f_1^* J)$  and  $g(\omega_{\phi}, J)$  are isometric, hence the Kähler manifolds  $(M, \omega, f_1^* J)$  and  $(M, \omega_{\phi}, J)$  are isomorphic.

In particular, if we fix  $J \in \mathcal{J}$  integrable, from Proposition 1.22 we have a map

$$F: \{\phi \in C^{\infty}(M, \mathbb{R}) \mid \omega_{\phi} \in \mathcal{K}_{J}(\omega)\} \longrightarrow \mathscr{J}$$
  
$$\phi \longmapsto F_{\phi}J := f_{1}^{*}J.$$
(1.14)

Using again equation (1.13), which can be generalised to all tensors, we see that the differential at 0 of F is given by

$$d_0 F(\dot{\phi}) = \left. \frac{\mathrm{d}}{\mathrm{d}t} \right|_{t=0} J_t = \left. \frac{\mathrm{d}}{\mathrm{d}t} \right|_{t=0} f_t^* J = \mathcal{L}_{\xi_t} J = \mathcal{L}_{JX_{\dot{\phi}}(\omega)} J = J \mathcal{L}_{\xi_{\dot{\phi}}(\omega)} J = -\frac{1}{2} P(\dot{\phi}).$$

In particular, we have obtained that the differential  $d_0 F(\dot{\phi})$  lies in the leaf  $\mathcal{D}_J$ . This means that a variation of the Kähler form in a given Kähler class for J fixed corresponds to a variation of the complex structure J in the same  $\mathscr{G}^c$ -orbit, for  $\omega$  fixed. In other terms, the  $\mathscr{G}^c$ -orbits of J integrable are in bijection with the space  $\mathcal{K}_I(\omega)$  of Kähler metrics in the class  $[\omega]$ .

# 1.4 Deformation theory of Kähler metrics with constant scalar curvature

In this section we follow Székelyhidi [77], although similar results were obtained also by Brönnle in his PhD thesis [6]. Let (M, L) be a polarised Kähler manifold and let  $\omega$  be a fixed Kähler form in  $c_1(L)$ . We fix  $J \in \mathscr{J}$  an integrable complex structure on  $(M, \omega)$  such that  $Scal(\omega, J)$  is constant. In this section, we describe the deformations of the complex structure J and we explain which deformations still define a cscK metric. We will use these results in §2.3, where we describe the deformations of *families* of cscK complex structures.

The deformations of the complex structure are encoded in a complex

$$C_0^{\infty}(M,\mathbb{C}) \xrightarrow{P^{\mathbb{C}}} T_J \mathscr{J} \xrightarrow{\bar{\partial}} \Omega^{0,1}(T^{1,0}M),$$

where  $P^{\mathbb{C}}$  is the complexification of the operator P (1.11). Let  $\tilde{H}^1$  be the cohomology of the complex. Then  $\tilde{H}^1$  can be described as

$$\widetilde{H}^{1} = \left\{ \alpha \in T_{J} \mathscr{J} \mid P^{\mathbb{C}^{*}} \alpha = \overline{\partial} \alpha = 0 \right\}.$$
(1.15)

This is a finite-dimensional vector space since it is the kernel of the elliptic operator

$$\Box = P^{\mathbb{C}}P^{\mathbb{C}^*} + (\bar{\partial}^*\bar{\partial})^2 \tag{1.16}$$

on  $T_J \mathscr{J}$ . Consider the group of Hamiltonian isometries of  $(M, \omega, J)$ , denoted by *K*: it is the group of functions  $\varphi \in \mathscr{G}$  such that

$$\mathrm{d}\varphi^{-1}J\mathrm{d}\varphi=J.$$

In particular, the group *K* is the stabiliser of the complex structure *J* for the action of  $\mathscr{G}$  which means that, by definition, it is the intersection of  $\mathscr{G}$  with Aut(*M*, *J*). The Lie algebra of *K*, denoted by  $\mathfrak{k}$ , consists of smooth functions over *M* such that their Hamiltonian vector field is also holomorphic, thus it can be identified with the kernel of *P*. The group *K* can be complexified and from Theorem 1.4 its complexification is Aut(*M*, *L*).

The map (1.14) can be generalised to a map between Sobolev spaces

$$F: L^2_k \to \mathscr{J}^2_{k-2}$$
$$\phi \mapsto F_{\phi}(J)$$

where  $\mathscr{J}_{k-2}^2$  is the set of almost complex structures on  $(M, \omega)$  with coefficients in  $L_{k-2}^2$ . This map is *K*-equivariant, with respect to the natural pull-back action on  $L_k^{21}$ .

From Kuranishi [49], we may construct a holomorphic embedding

$$\Phi_1: V_1 \to \mathscr{J} \tag{1.17}$$

where  $V_1 \,\subset \, \overline{H}^1$  is a ball around the origin. The map  $\Phi_1$  maps the origin 0 to the reference complex structure J. The group K acts naturally on  $\widetilde{H}^1$  by pull-back, and hence on  $V_1$ , and the map  $\Phi_1$  is K-equivariant. Moreover,  $V_1$  parametrises  $\mathscr{G}^c$ -orbits of integrable complex structures near  $J_0$ , in the sense that the  $\mathscr{G}^c$ -orbit of every integrable complex structure near  $J_0$  intersects the image of  $\Phi_1$ . The following theorem is mainly due to Kuranishi [49]. A proof of items 1. and 2. adapted to take into account the compatibility with the symplectic form can be found in [12, §6]. The third claim is due to Székelyhidi [77].

**Theorem 1.23.** There exists a ball around the origin  $V \subset \widetilde{H}^1$  and a K-equivariant map

$$\Phi: V \to \mathscr{J} \tag{1.18}$$

such that  $\Phi(0) = J$  and

- 1. the  $\mathscr{G}^{c}$ -orbit of every integrable complex structure near  $J_{0}$  intersects the image of  $\Phi$ ;
- 2. *if two points x and x' of V are in the same orbit for the complexified action of K, and*  $\Phi(x)$  *is integrable, then their images*  $\Phi(x)$  *and*  $\Phi(x')$  *are in the same*  $\mathscr{G}^c$ *-orbit;*
- 3. Scal( $\omega$ ,  $\Phi(x)$ )  $\widehat{S}$  is an element of the Lie algebra of K.

The open ball *V* is a local slice of the  $\mathscr{G}^c$ -action near the reference complex structure *J*. We will refer to it as the *Kuranishi space* and to  $\Phi$  as the *Kuranishi map*. Since we allow also non integrable almost complex structure, the slice is an actual ball. Instead, in the original work by Kuranishi, the set *V* parametrises only *integrable* complex structures, hence it is a complex analytic subspace of our *V*. Moreover, while the map  $\Phi_1$  (1.17) is holomorphic, our Kuranishi map  $\Phi$  (1.18) is not holomorphic in general: this is due to the fact that  $\Phi$  is perturbed from  $\Phi_1$  using the implicit function theorem in order to meet the third requirement of Theorem 1.23.

Consider the symplectic form on *V* pulled-back via  $\Phi$  from the Kähler form  $\Omega$  on  $\mathscr{J}$ . We will denote this symplectic form again by  $\Omega$ , since it is essentially the same. Then we obtain a moment map for the *K*-action on *V*:

$$\mu: V \to \mathfrak{k}$$

$$x \mapsto S(\omega, \Phi(x)) - \widehat{S},$$
(1.19)

<sup>&</sup>lt;sup>1</sup>One can see this by using the following relation on the pull-back of a Hamiltonian vector field:  $F^*(\xi_h(\omega)) = \xi_h(F^*\omega)$ .

where t is identified with its dual via the  $L^2$ -product of functions. So the question about which deformations of the complex structure *J* admit a cscK metric can be restated in terms of finding a  $K^{\mathbb{C}}$ -orbit which contains a zero of the moment map  $\mu$ . We will expand on this in the next section.

Until this point, we have considered deformations of the complex structure of a fixed symplectic manifold and we have treated them as (0, 1)-forms with values in the (1, 0)-tangent bundle. Another point of view consists in considering deformations of the manifold M as a family  $\pi : \mathcal{U} \to T$ , where  $\pi$  is a smooth proper morphism and T is a complex space. The simplest example is when T is the double point Spec  $\mathbb{C}[\varepsilon]/\varepsilon^2$ . However, if we consider also non integrable deformations, we can consider T to be an open disk.

**Definition 1.24.** We say that the deformation family is *complete* if, for any other deformation family  $\mathcal{U}' \to T'$ , there exists a map

$$\tau: T' \to T$$

such that  $\mathcal{U}' = \tau^* \mathcal{U}$ . Moreover, if the differential of  $\tau$  at 0 is unique we say that the deformation family is *versal*, and *universal* if  $\tau$  itself is unique.

Ehresmann's fibration theorem [44, Theorem 6.2.2] guarantees that the two points of view are interchangeable.

**Theorem 1.25** (Ehresmann). Let  $\pi : \mathcal{U} \to T$  be a proper family of differentiable manifolds. If *T* is connected, then all the fibres are diffeomorphic.

Throughout this work, we will make extensive use of Ehresmann's theorem to view any family of deformations of a given complex manifold as a family of deformations of its complex structure. In particular Kuranishi's Theorem [49] gives the existence of a versal deformation family centred at M with the complex structure J and base the Kuranishi space V. Moreover, the Kuranishi deformation family is complete for nearby complex structures.

#### 1.4.1 Reduction to the finite dimensional problem

In this section we follow Inoue [45, §3] to compute a moment map for the linearised action of *K* on the tangent space to the Kuranishi space (see also [64, §3.5]). Such a computation will be used in the definition of an optimal symplectic connection and its linearisation. We then give a proof of a theorem of Székelyhidi [77] on finding a  $K^{\mathbb{C}}$ -orbit in *V* which contains a zero of the moment map (1.19).

On *V* we have the symplectic form  $\Omega$  pulled back from the one of  $\mathcal{J}$  and by definition the moment map (1.19) satisfies

$$\mathrm{d}_{x}\langle \mu, f\rangle(v) = \Omega_{x}(v, \mathcal{L}_{\eta_{f}}x),$$

where  $x \in V$  is a (0, 1)-form with values in the holomorphic tangent bundle and  $\eta_f$  is the Hamiltonian vector field of f, also denoted by grad<sup> $\omega$ </sup> f. We will also use

the notation  $\sigma_x(f)$  for the infinitesimal action given by the Lie derivative of *x* along the Hamiltonian vector field associated with *f*.

The origin of *V* is a fixed point of the action. By identifying  $T_0V$  with  $\widetilde{H}^1$ , we consider on  $\widetilde{H}^1$  the linear symplectic form

$$\Omega_0(\cdot, \cdot) = \mathbf{\Omega}_{I_0}(\mathbf{d}_0 \Phi \cdot, \mathbf{d}_0 \Phi \cdot), \tag{1.20}$$

and the linear action of *K* induced by the one on *V*. For any  $f \in \mathfrak{k}$ , consider the endomorphism of  $\widetilde{H}^1$ 

$$A_f(t) = d_0 \left( y \mapsto \exp(tf) \cdot y \right)$$

where by  $\exp(tf)$  we denote the 1-parameter subgroup of *K* defined by the element  $f \in \mathfrak{k}$ . It corresponds via  $\Phi$  to the flow of the Hamiltonian vector field  $\eta_f$  on *M*, which we denote by  $\rho_t^f$ . The operator  $A_f(t)$  is a unitary operator, since it is linear and symplectic, because the group *K* acts by symplectomorphisms on *V*. Let

$$A_f := \left. \frac{\mathrm{d}}{\mathrm{d}t} \right|_{t=0} A_f(t). \tag{1.21}$$

We have the following properties:

- 1.  $A_f$  is a skew-hermitian endomorphism of  $(\widetilde{H}^1, J_0)$  and  $A_f(t) = \exp(tA_f)$ ;
- 2. For  $v \in \widetilde{H}^1$ , denote by **v** a vector field on *V* such that **v**  $|_0 = v$ . Then

$$A_{f}(v) = \partial_{t} |_{t=0} A_{f}(t)v = \partial_{t} |_{t=0} \left( (\rho_{t}^{f})_{*}(\mathbf{v}) \right)_{0} = -(\mathcal{L}_{\eta_{f}}\mathbf{v})_{0} = [\mathbf{v}, \eta_{f}]_{0}.$$
(1.22)

**Definition 1.26.** We define a map  $\nu : \widetilde{H}^1 \to \mathfrak{k}$  by

$$\langle v(v), f \rangle = \frac{1}{2} \Omega_0(A_f v, v).$$

The map  $\nu$  can be characterised as a moment map by relating it to the scalar curvature (1.19) as follows [45, §3]. We begin by computing the second derivative of the moment map  $\mu$ , as follows:

$$\begin{aligned} \frac{\mathrm{d}^2}{\mathrm{d}t^2} \Big|_{t=0} \left\langle \mu(tv), f \right\rangle &= \left. \frac{\mathrm{d}}{\mathrm{d}t} \right|_{t=0} \left\langle \frac{\mathrm{d}}{\mathrm{d}t} \mu(tv), f \right\rangle \\ &= \left. \frac{\mathrm{d}}{\mathrm{d}t} \right|_{t=0} \left\langle \mathrm{d}_{tv} \mu(v), f \right\rangle \\ &= \left. \frac{\mathrm{d}}{\mathrm{d}t} \right|_{t=0} \Omega_{tv}(v, \mathcal{L}_{\eta_f}(tv)) \\ &= \Omega_0(v, -A_f v) \\ &= \left\langle \nu(v), f \right\rangle. \end{aligned}$$
In particular, written as a matrix product, the above relation becomes

$$v^{\mathsf{T}} \cdot \operatorname{Hess}_0(\langle \mu, f \rangle) \cdot v = -v^{\mathsf{T}} \Omega_0 A_f v,$$

hence the matrix  $\Omega_0 A_f$  is symmetric. Therefore, given a path  $v_t$  in  $\widetilde{H}^1$  such that  $v_{t=0} = v_0$  and  $\partial_t|_{t=0}v_t = v$ ,

$$\begin{aligned} \frac{\mathrm{d}}{\mathrm{d}t} \bigg|_{t=0} \left\langle v(v_t), f \right\rangle &\stackrel{\text{def}}{=} \left. \frac{\mathrm{d}}{\mathrm{d}t} \bigg|_{t=0} \frac{1}{2} \Omega_0(A_f v_t, v_t) \\ &= \frac{1}{2} \Omega_0(A_f v_0, v) + \frac{1}{2} \Omega_0(A_f v, v_0) \\ &= \Omega_0(A_f v_0, v) \\ &= \Omega_0(v, (\mathcal{L}_{\eta_f} \mathbf{v})_0), \end{aligned}$$

i.e. *v* satisfies the moment map condition. The following theorem was first considered by Székelyhidi [77, Proposition 8].

**Theorem 1.27.** Let  $v \in V \subset \widetilde{H}^1$  be a GIT-polystable point for the  $K^{\mathbb{C}}$ -action on  $\widetilde{H}^1$ . Then there is a point  $x_0 \in V$  in the same  $K^{\mathbb{C}}$ -orbit as v such that  $\mu(x_0) = 0$ .

*Remark* 1.28 (Postscriptum). In the approved version of this thesis, we included a proof of the above statement, which differed slightly from the original and used a technique involving the implicit function theorem. Unfortunaltely, both Székely-hidi's original proof and the one included in the earlier version of this manuscript have a gap. In fact, the argument relies on a lower bound on the norm of the differential of  $\mu$  along the infinitesimal vector field that is uniform in a ball around the origin; this uniform bound cannot exist, because the differential of  $\mu$  vanishes at the fixed points, which form a linear subspace.

A proof that uses a different technique involving the moment map flow is given in the author's subsequent work [67, Theorem A.3].

## 1.5 The moduli space of cscK manifolds

Fujiki and Schumacher [31] have constructed a moduli space of polarised cscK manifolds with a discrete group of automorphisms, using the Kuranishi theory we described in §1.4. Dervan, Neumann [13] and Inoue [45] extended their result to the case when the group of automorphisms is not discrete. We conclude this chapter with a brief discussion of their construction and results, which we will use in various instances in the following chapters.

**Theorem 1.29** ([31, 13, 45]). There is a Hausdorff complex space  $\mathcal{M}^{cscK}$  that parametrises Kähler manifolds with constant scalar curvature. The space  $\mathcal{M}^{cscK}$  admits a Weil-Petersson type Kähler metric  $\alpha_{WP}$  that represents the first Chern class of a line bundle over  $\mathcal{M}^{cscK}$ .

#### Background

In the case of discrete automorphism group, the moduli space is defined locally around a cscK manifold (M, L) as the Kuranishi space V defined in Theorem 1.23 quotient by the action of the group of automorphisms. More precisely, we consider the complex analytic subspace  $V^{int}$  of V of integrable deformations of the complex structure of M. Fujiki and Schumacher prove that there exists an open neighbourhood of the origin of  $V^{int}$  where the cscK equation has a solution, using the implicit function theorem and the fact the automorphism group is discrete. The local structure of the moduli space is then given as the quotient of  $V^{int}$  by the action of Aut(M, L), which is a *finite* discrete group. In particular, the moduli space has locally the structure of a complex orbifold space. When (M, L) has continuous automorphisms, the local structure of the moduli space is instead that of a GIT quotient  $V^{int}$  // Aut(M, L).

We next describe the definition of the Weil-Petersson type Kähler metric on the moduli space. We start by recalling that the notions of a smooth Hermitian metric and of a Kähler metric are well posed on a complex space [31, Definitions 1.1, 1.2]. Let (M, L) be a cscK manifold of complex dimension m and let  $\mathcal{U} \to V^{int}$  be the Kuranishi family with central fibre (M, L) described with the discussion following Definition 1.24. In particular,  $\mathcal{U} \to V^{int}$  is a family of cscK manifolds. From Theorem 1.23 we get an injective map

$$d_0\Phi: T_0V^{int} \to H^0(M, T^{1,0}M).$$
(1.23)

We use this map to pull-back the standard  $L^2$ -product on  $H^0(M, T^{1,0}M)$ . More precisely, let  $\alpha, \beta \in T_0V$ . Then we can consider  $\alpha$  and  $\beta$  as *harmonic* (0, 1)-forms with values in  $T^{1,0}M$ , where harmonicity is defined with respect to the Laplacian type operator (1.16). Then for each  $t \in V^{int}$ ,

$$\alpha_t := \langle \alpha, \beta \rangle_t = \int_{M_t} \Lambda_{\omega_t} \operatorname{Tr}_{\omega_t}(\alpha \overline{\beta}) \omega_t^m.$$

**Definition 1.30.** The *Weil-Petersson metric* on  $V^{int}$  is the two-form given by the collection  $\{\alpha_t\}$ .

The Weil-Petersson metric satisfies the following fibre-integral formula.

**Theorem 1.31** ([13, 31]). The Weil-Petersson metric on  $V^{int}$  coincides with the (1, 1)-form:

$$\alpha_{WP} = \frac{S}{m+1} \int_{\mathcal{U}/V} \omega^{m+1} - \int_{\mathcal{U}/V} \rho \wedge \omega^m, \qquad (1.24)$$

where  $\widehat{S}$  is the average scalar curvature of M and  $\rho$  is the relative Ricci form induced by  $\omega$  on the fibres of  $\mathcal{U} \to V$ :

$$\rho = i\partial\bar{\partial}\log(\omega^m).$$

To produce an actual Kähler metric on the moduli space, the two-form (1.24) is then glued on the local charts  $V^{int} / \operatorname{Aut}(M, L)$ . Moreover, for any family  $\pi_X : X \to$ 

#### 1.5 The moduli space of cscK manifolds

*B* of cscK manifolds, a Weil-Petersson type form can be defined by the expression (1.24) as a fibre integral for the fibration  $\pi_X$ . More precisely,  $\pi_X$  induces a map

$$q: B \to \mathcal{M}^{cscK}$$
,

such that the pull-back of the Weil-Petersson metric on  $\mathcal{M}^{cscK}$  via q is a semipositive closed two-form on B and it can be written as the fibre integral (1.24) over X/B. Moreover,  $\alpha_{WP}$  is positive definite if and only if the map (1.23) for the family  $X \to B$  is an immersion.

Background

## Chapter 2

## **Holomorphic submersions**

Let  $\pi_X : X \to B$  be a holomorphic submersion between compact Kähler manifolds whose fibres are all smooth and connected. By Ehresmann's fibration theorem 1.25 the fibres are diffeomorphic. Let *n* be the dimension of *B* and *m* be the dimension of the fibres, so that dim(X) = n + m.

**Definition 2.1.** A line bundle  $H_X$  on the total space X of the submersion  $\pi_X$  is said to be *relatively ample* if its restrictions to the fibres of  $\pi_X$  are ample. Similarly, a representative  $\omega \in c_1(H_X)$  is called a *relatively Kähler metric* if its restrictions to the fibres of  $\pi_X$  are Kähler metrics.

In the following, we always assume that *X* admits a relatively ample line bundle  $H_X$  and *B* admits an ample line bundle *L*, and we consider the line bundles as fixed. Thus when we write  $\pi_X : X \to B$  we always mean  $\pi_X : (X, H_X) \to (B, L)$ .

Let  $\omega$  be a relatively Kähler metric on X in  $c_1(H_X)$  and  $\omega_B$  a Kähler metric on B in  $c_1(L)$ . We can define a Kähler metric on X by taking the relative Kähler metric  $\omega$  and adding a large multiple of the base metric, pulled-back on X

$$\omega_k = \omega + k \pi_X^* \omega_B \qquad k \gg 0.$$

We often omit the pull-back and write  $\omega + k\omega_B$ . The relatively Kähler form  $\omega$  determines a splitting of the tangent space

$$TX = \mathcal{V} \oplus \mathcal{H}^{\omega} \tag{2.1}$$

where  $\mathcal{V}_x = T_x X_{\pi_X(x)}$  is the tangent space to the fibre, and

$$\mathcal{H}_x^{\omega} = \{ u \in T_x X \mid \omega(u, v) = 0 \; \forall v \in \mathcal{V}_x \}$$

In the context of symplectic fibrations  $\omega$  is called a *symplectic connection* [56, Chapter 6] and the splitting (2.1) induces a splitting on all tensor bundles. We denote by  $\omega_F$  the purely vertical part of  $\omega$  and by  $\omega_H$  the purely horizontal part of  $\omega$ .

In what follows will also need the groups of automorphisms of the projections  $\pi$ .

#### Holomorphic submersions

**Definition 2.2.** For  $\pi_X : X \to B$  the group of *relative automorphisms* of the fibration, or the group of *automorphisms of*  $\pi$ , is

$$\operatorname{Aut}(\pi_X) := \{ f \in \operatorname{Aut}(X, H_X) \mid \pi_X \circ f = \pi_X \}.$$

## 2.1 Splitting of the function space

In this section, we assume that the fibres of *X* each admits a constant scalar curvature Kähler metric. We also assume that the spaces  $H^0(X_b, T^{1,0}X_b)$  of holomorphic vector fields on the fibre  $X_b$  are isomorphic as Lie algebras for all *b*. The following lemma explains that one can define a relatively Kähler metric on the total space which is relatively cscK.

**Lemma 2.3** ([14, Lemma 3.8]). For any  $b \in B$ , let  $\omega_b$  be a cscK metric on the fibre  $X_b$  in the class  $c_1(H_X|_b)$ . Then there exists a  $\omega \in c_1(H_X)$  which is relatively cscK.

Denoting by *I* the complex structure of *X*, we next explain how the relative Kähler metric ( $\omega$ , *I*) induces a splitting of the space of smooth functions on *X*. Let

$$\mathcal{D}^*_{\mathcal{W}}\mathcal{D}_{\mathcal{V}}: C^{\infty}(X,\mathbb{R}) \to C^{\infty}(X,\mathbb{R})$$

be the vertical Lichnerowicz operator, defined fibrewise as

$$\left(\mathcal{D}_{\mathcal{V}}^{*}\mathcal{D}_{\mathcal{V}}\varphi\right)|_{X_{b}} = \mathcal{D}_{b}^{*}\mathcal{D}_{b}\left.\varphi\right|_{X_{b}}$$

It is a real operator since the fibrewise metric is cscK. By integrating a function  $\varphi \in C^{\infty}(X, \mathbb{R})$  over the fibres of  $\pi_X$ , we define a projection

$$C^{\infty}(X,\mathbb{R}) \longrightarrow C^{\infty}(B,\mathbb{R})$$
$$\varphi \longmapsto \int_{X/B} \varphi \omega^{m}$$

Its kernel is given by the space  $C_0^{\infty}(X, \mathbb{R})$  of functions that have fibrewise mean value zero. A key step in the study of optimal symplectic connections is that we can further split this space as follows.

Consider a real vector bundle  $E \to B$  [14, §3.1], whose fibre over  $b \in B$  is the real finite-dimensional vector space ker<sub>0</sub>( $\mathcal{D}_b^* \mathcal{D}_b$ ) of holomorphy potentials on the fibre  $X_b$  with mean-value zero with respect to  $\omega_b$ . E is well defined as a vector bundle since we are assuming that the complex dimension of the Lie algebra  $\mathfrak{h}(X_b)$  of holomorphic vector fields on  $X_b$  is independent of b. Its smooth global sections are

$$C^{\infty}(E) = \ker_0 \mathcal{D}^*_{\mathcal{V}} \mathcal{D}_{\mathcal{V}}.$$

In [34, Lemma 2.7], Hallam showed - using the Cartan decomposition for the space  $\mathfrak{h}(X_b)$  of holomorphic vector fields of the fibre - that  $E_b$  can be also viewed as

the vector space of all Kähler potentials  $\varphi_b$  on  $X_b$  of mean-value zero for which  $\omega_b + i\partial \bar{\partial} \varphi_b$  is still cscK. We can split  $C_0^{\infty}(X)$  as

$$C_0^{\infty}(X,\mathbb{R}) = C^{\infty}(E) \oplus C^{\infty}(R),$$

where  $C^{\infty}(R)$  is the fibrewise  $L^2$ -orthogonal complement with respect to the fibre metric  $\omega_b$ , i.e. for all  $\varphi \in \ker_0 \mathcal{D}_b^* \mathcal{D}_b$ ,  $\psi \in C^{\infty}(R)$ 

$$\langle \varphi, \psi \rangle_b := \int_{X_b} \varphi \psi \omega_b^m = 0.$$

In the end, we obtain

$$C^{\infty}(X, \mathbb{R}) = C^{\infty}(B) \oplus C^{\infty}(E) \oplus C^{\infty}(R).$$
(2.2)

We denote by  $p_E : C^{\infty}(X) \to C^{\infty}(E)$  the projection.

Since we are interested in deformations of the complex structure of *X*, sometimes we will denote the vector bundle *E* as  $E(\omega, I)$  to underline its dependence on the Kähler metric. Notice that if we change just the relatively Kähler metric  $\omega$  to  $\omega + i\partial \bar{\partial} \varphi$ , for  $\varphi \in C^{\infty}(X)$ , the vector bundles  $E(\omega, I)$  and  $E(\omega + i\partial \bar{\partial} \varphi, I)$  are isomorphic. We give the following definition.

**Definition 2.4.** We denote by  $\mathcal{K}_E$  the space of all smooth functions  $\varphi \in C^{\infty}(X)$  such that  $\omega + i\partial \bar{\partial} \varphi$  is still a *fibrewise cscK metric*.

The following proposition [17, Lemma 4.20] relates the space  $\mathcal{K}_E$  to the vector bundle  $E \rightarrow B$ , justifying the notation.

**Proposition 2.5.** Let  $\varphi_t : [0, 1] \to \mathcal{K}_E$  be a smooth path. Then for all t

$$\dot{\varphi}_t \in C^{\infty}(B) \oplus C^{\infty}(E(\omega + i\partial \partial \varphi_t, I))$$

that is to say that, up to a function on the base,  $\dot{\varphi}_t$  is a fibrewise holomorphy potential with respect to  $\omega + i\partial \bar{\partial} \varphi_t$ .

Again, if we want to underline the dependence on the complex structure, we write  $\mathcal{K}_E(I)$ .

## 2.2 Optimal symplectic connections

In this section, we give the definition of optimal symplectic connection. We begin in §2.2.1 by describing optimal symplectic connections on fibrations with cscK fibres, following Dervan-Sektnan [14]. In §2.2.2 we describe a generalisation of optimal symplectic connection to fibrations with analytically K-semistable fibres.

#### 2.2.1 The relatively cscK case

Let  $\omega \in c_1(X)$  be a relatively Kähler metric. Then  $\omega$  defines two curvature quantities on *X*, the symplectic curvature and the relative Ricci curvature.

**Definition 2.6.** The *symplectic curvature* of  $\omega$  is a two-form on *B* with values in the fibrewise Hamiltonian vector fields given as follows: if  $v_1, v_2 \in \mathfrak{X}(B)$ ,

$$F_{\mathcal{H}}(v_1, v_2) = [v_1^{\sharp}, v_2^{\sharp}]^{\text{vert}},$$

where  $v_i^{\sharp}$  denotes the horizontal lift.

Let  $\gamma^*$  be the map which associates to a fibrewise Hamiltonian vector field its fibrewise Hamiltonian function with fibrewise mean value zero. Thus we consider  $\gamma^*(F_{\mathcal{H}})$ , which is a two-form on *B* with values in  $C_0^{\infty}(X, \mathbb{R})$ , and we pull it back on *X*. Notice that the two-form  $\gamma^*(F_{\mathcal{H}})$  depends only on the relatively symplectic form and not on the complex structure. Moreover, the symplectic curvature is related to the symplectic connection as follows [14, §3.2]:

$$\omega_{\mathcal{H}} = \gamma^*(F_{\mathcal{H}}) + \pi^*_X\beta,$$

where  $\beta$  is a two-form on *B*.

**Definition 2.7.** The relatively Kähler form  $\omega$  induces a Hermitian metric  $\wedge^m g_\omega$  on the top-wedge power of the vertical tangent bundle

$$\bigwedge^m \mathcal{V}^{1,0} = -K_{X/B}.$$

Explicitly, if  $A, B \in \bigwedge^m \mathcal{V}^{1,0}$ , we can write locally  $A = f_A \partial_{w^1} \wedge \cdots \wedge \partial_{w^m}$  and  $B = f_B \partial_{w^1} \wedge \cdots \wedge \partial_{w^m}$ . Then

$$\langle A, B \rangle_{\wedge^m g_\omega} = \det(g_\omega) f_A f_B.$$

Its curvature, denoted by  $\rho$ , represents the first Chern class of the relative anticanonical bundle  $-K_{X/B}$ . We call  $\rho$  the *relative Ricci curvature* of  $\omega$  and we denote by  $\rho_{\mathcal{H}}$  its purely horizontal part.

We can now give the definition of an optimal symplectic connection according to [14].

**Definition 2.8.** Let  $\omega$  be a relatively cscK metric. Then  $\omega$  is an *optimal symplectic connection* if

$$p_E \left( \Delta_{\mathcal{V}}(\Lambda_{\omega_B} \gamma^*(F_{\mathcal{H}})) + \Lambda_{\omega_B} \rho_{\mathcal{H}} \right) = 0.$$
(2.3)

This is a second-order elliptic equation on the vector bundle  $E \to B$  [14, Theorem 4.9]. In the following, we will use the notation  $\Theta(\omega, I) = \Delta_{\mathcal{V}}(\Lambda_{\omega_B}\gamma^*(F_{\mathcal{H}})) + \Lambda_{\omega_B}\rho_{\mathcal{H}}$ .

#### 2.2 Optimal symplectic connections

The optimal symplectic connection equation arises from expanding the scalar curvature of  $\omega_k$  in negative powers of k. Indeed, to leading order the expansion reads

$$\operatorname{Scal}(\omega_k) = \widehat{S}_b + O(k^{-1}),$$

where the leading order term is the scalar curvature of the fibres, which we assume to be constant and is independent of *b*. The left-hand side of equation (2.3) is then the projection onto  $C^{\infty}(E)$  of the  $k^{-1}$ -term of the expansion. x The linearisation of the equation at a solution is given by the operator  $\mathcal{R}^*\mathcal{R}$  [14, §4.3], where

$$\mathcal{R}(\varphi_E) = \bar{\partial}_B \nabla_{\mathcal{V}}^{1,0} \varphi_E \tag{2.4}$$

and the adjoint is computed with respect to the Hermitian metric  $\omega_F + \omega_B$ . Here  $\nabla_V^{1,0} \varphi_E$  is a section of the holomorphic tangent bundle; the vertical part of  $\bar{\partial} \nabla_V^{1,0} \varphi_E$  vanishes since  $\varphi \in C^{\infty}(E)$  and the horizontal part is denoted by the expression (2.4). The operator (2.4) can be described as follows [14, §4.3]: let  $\mathcal{D}_k^* \mathcal{D}_k$  be the Lichnerowicz operator with respect to the Kähler metric  $\omega_k$ . It can be written as a power series expansion in negative powers of *k*:

$$\mathcal{D}_k^*\mathcal{D}_k = \mathcal{L}_0 + k^{-1}\mathcal{L}_1 + O\left(k^{-2}\right)$$

where  $\mathcal{L}_0$  is the *vertical* Lichnerowicz operator  $\mathcal{D}^*_{\mathcal{V}}\mathcal{D}_{\mathcal{V}}$ . Then for  $\varphi, \psi$  fibrewise holomorphy potentials

$$\int_{X} \varphi \mathcal{L}_{1}(\psi) \omega^{m} \wedge \omega_{B}^{n} = \int_{X} \langle \mathcal{R} \varphi, \mathcal{R} \psi \rangle_{\omega_{F} + \omega_{B}} \omega^{m} \wedge \omega_{B}^{n}.$$

This means that the operator  $\mathcal{R}^*\mathcal{R}$  can actually be seen as  $p_E \circ \mathcal{L}_1$  restricted to  $C_E^{\infty}(X)$ . The kernel of  $\mathcal{R}$ , thus of  $\mathcal{R}^*\mathcal{R}$ , consists of fibrewise holomorphy potentials which are global holomorphy potentials on X with respect to  $\omega_k$ .

#### 2.2.2 Optimal symplectic connections in general

Let now  $(Y, H_Y) \rightarrow (B, L)$  be a polarised holomorphic submersion with  $\omega \in c_1(H_Y)$  a relatively Kähler metric. The following assumptions restrict the class of admissible fibrations to those whose fibres satisfy a stability property defined in terms of K-stability. More precisely we assume that:

- 1. the fibres  $Y_b$  are analytically K-semistable, which means by definition that they each admit a degeneration to a cscK manifold  $X_b$ . We also assume that the deformation is compatible with the fibration structure in the following sense: there exists a holomorphic map  $\hat{\pi} : (X, \mathcal{H}) \to (B, L) \times S$ , parametrised by a disk *S*, such that for  $s \neq 0$ , the family  $(X_s, \mathcal{H}_s) \to B$  is isomorphic to the original fibration  $\pi_Y : (Y, H_Y) \to B$  and the central fibration at s = 0 is a family  $\pi_X : (X, H_X) \to B$  whose fibres are cscK;
- 2. the automorphism groups  $Aut_0(X_b, H_b)$  of the fibres are all isomorphic.

The first hypothesis is a stability assumption. We will refer to the submersion  $X \rightarrow B$  as the *relatively cscK degeneration* of  $Y \rightarrow B$ . The second hypothesis holds if and only if the spaces  $H^0(X_b, T^{1,0}X_b)$  are isomorphic as Lie algebras, which we assumed in §2.1 to define the vector bundle  $E \rightarrow B$ . It is needed to define optimal symplectic connections and it is a key hypothesis for the construction of the moduli space of fibrations in §4.1.

A relative version of Ehresmann's theorem, which will be proved as Proposition 2.19, implies that we can locally trivialise the family in such a way that all  $X_s$  are diffeomorphic. So we can take the perspective of fixing  $\omega$  and seeing  $X \rightarrow B \times S$  as a family of compatible complex structures  $\{J_s\}$  which keep  $\pi_X$  a holomorphic submersion and are all biholomorphic except for  $J_0$ .

#### Let

 $\mathscr{J}_{\pi} = \{ J \text{ almost complex structure compatible with } \omega \text{ and s.t. } d\pi \circ J = J_B \circ d\pi \}.$ (2.5)

Compatibility with  $\omega$  means that  $\omega(J, J) = \omega(\cdot, \cdot)$  and that  $\omega_b \circ J |_{X_b}$  is non degenerate and positive definite. The tangent space at *I* to  $\mathscr{J}_{\pi}$  can be identified with

$$T_{I}^{0,1}\mathscr{J}_{\pi} = \left\{ A \in \Omega^{0,1}(\mathcal{V}^{1,0}) \mid \omega_{F}(A,\cdot,\cdot) + \omega_{F}(\cdot,A,\cdot) = 0 \right\},$$
(2.6)

where  $\omega_F$  is the purely vertical part of  $\omega$ .

As in §1.4, for any fibre  $X_b$  let  $V_b$  be the Kuranishi space,  $K_b$  the group of Hamiltonian isometries and  $\Psi_b$  the Kuranishi map (1.18) of the fibre. Let  $x_{s,b} \in V_b$  be such that  $\Psi_b(x_{s,b}) = J_s |_{X_b}$ . Let  $\mu_b$  be the moment map (1.19). Then we can define a section of  $C^{\infty}(E)$  as

$$\mu_s(b) = p_E\left(\operatorname{Scal}_{X_b}\left(\omega_b, \Psi_b(x_{s,b})\right)\right). \tag{2.7}$$

Note that  $\Psi_b$  may not vary smoothly with b, but when applied to  $x_{s,b}$  it gives the complex structure  $J_s |_{X_b}$ . Since  $J_s$  is a complex structure defined on the whole X, it varies smoothly with the base, so  $\mu_s$  is a smooth section. For each fibre  $X_b$  we can linearise the action to the tangent space to  $V_b$  at 0 as in §1.4, so we can define another section  $\nu$  of E by

$$\nu(b) = \nu_b(v_b). \tag{2.8}$$

Here  $v_b$  is the moment map defined in Definition 1.26 for the linear action of  $K_b$  on  $\tilde{H}^1(X_b)$ , and  $v_b \in \tilde{H}^1(X_b)$  is tangent at  $0 \in V_b$ . Even if  $v_b$  does not necessarily vary smoothly with b, v is a smooth section because there is an expansion

$$\mu_s(b) = \frac{s^2}{2}v + O(s^3) \tag{2.9}$$

as explained in Proposition **??**, and  $\mu_s$  is smooth.

**Definition 2.9.** We say that the relative Kähler form  $\omega$  is an *optimal symplectic connection* on  $(Y, H_Y) \rightarrow B$  if it satisfies the equation

$$p_E \left( \Delta_{\mathcal{V}} (\Lambda_{\omega_B} \gamma^*(F_{\mathcal{H}})) + \Lambda_{\omega_B} \rho_{\mathcal{H}} \right) + \frac{\lambda}{2} \nu = 0$$
(2.10)

for some constant  $\lambda > 0$ .

The first term is the left-hand side of the optimal symplectic connection equation (2.3) for fibrewise cscK metrics, and it involves only the complex structure I of the relatively cscK degeneration. The second term represents the deformation of the complex structure, in terms of the first-order deformation of the fibres.

*Remark* 2.10. Similarly to the relatively cscK case, the equation arises by expanding the scalar curvature of the Kähler metric ( $\omega_k$ ,  $J_s$ ) in inverse powers of k and powers of s and then relating the two parameters to obtain a single expansion. The left-hand side of (2.10) is then the projection onto  $C^{\infty}(E(\omega, I))$  of the sub-leading order term of such an expansion. We explain the expansion of the scalar curvature in detail in §3.1.

In §3.3, we will prove that the linearisation of the equation at a solution is given by an operator

$$\widehat{\mathcal{L}} = \mathcal{R}^* \mathcal{R} + \mathcal{A}^* \mathcal{A},$$

where  $\mathcal{R}$  is the operator (2.4) and  $\mathcal{A}$  is obtained by extending the map (1.21) to a fibrewise map. Its kernel is given by the fibrewise *I*-holomorphy potentials which are global holomorphy potentials with respect to  $J_s$ .

The definition of an optimal symplectic connection can be generalised to that of an extremal symplectic connection as follows.

**Definition 2.11.** The relative Kähler form  $\omega$  is an *extremal symplectic connection* on *Y* if

$$\widehat{\mathcal{L}}\left(p_E(\Theta(\omega,I)) + \frac{\lambda}{2}\nu\right) = 0.$$

In particular, if  $\omega$  is an extremal symplectic connection, the fibrewise holomorphy potential for the complex structure *I* 

$$h_1 := p_E(\Theta(\omega, I)) + \frac{\lambda}{2}\nu \tag{2.11}$$

is also a holomorphy potential for the complex structure of *Y*. The definition of an extremal symplectic connection extends the one given by Dervan-Sektnan on relatively cscK fibrations [14, §3.4].

### 2.3 Deformations of fibrations

In this section, we study in more detail the deformations of a holomorphic fibrewise cscK fibration. In particular, we prove a relative version of Ehresmann's fibration theorem in Proposition 2.19, which allows us to view families of fibrations as families of complex structures in  $\mathcal{J}_{\pi}$  on the same underlying smooth fibration. The main result of this section is a relative version of Kuranishi's Theorem 1.23, which

will be needed to describe the linearisation of the optimal symplectic connection equation (2.10) and to study the moduli space of its solutions.

We start by giving a description in local coordinates of a first-order deformation  $A \in T_I \mathscr{J}_{\pi}$ . In local computations, we make the following notational conventions:

 $\{w^1, \ldots, w^m\}$  denote vertical holomorphic coordinates; indices are denoted with the letters *a*, *b*, *c*, ...;

 $\{z^1, \ldots, z^n\}$  denote holomorphic coordinates on the base; indices are denoted with the letters *i*, *j*, *k*, ...;

 $\{\zeta^1, \ldots, \zeta^{n+m}\}$  denote holomorphic coordinates on the total space; indices are denoted with the letters  $p, q, r, \ldots$ 

We write  $A \in T_J \mathscr{J}_{\pi}$  locally as

$$A = A^{a}_{\ \bar{b}} \mathrm{d}\bar{w}^{b} \otimes \partial_{w^{a}} + A^{a}_{\ \bar{i}} \mathrm{d}\bar{z}^{j} \otimes \partial_{w^{a}},$$

since *A* takes values in the vertical vector fields. The following lemma explains the relation between  $A^a_{\ \bar{b}}$  and  $A^a_{\ \bar{i}}$ .

**Lemma 2.12.** *For*  $A \in T_I \mathscr{J}_{\pi}$  *we have that:* 

- 1. A vanishes on horizontal vector fields;
- 2.  $A^a_{\overline{i}} = A^a_{\overline{c}}(\omega_F)^{d\overline{c}}(\omega)_{d\overline{j}}$ .

*Proof.* As for the first claim, if  $u \in \mathcal{V}$ ,  $v \in \mathcal{H}^{\omega}$ , then

$$\omega(u, Av) = \omega_F(u, Av) = -\omega_F(Au, v) = 0.$$
(2.12)

Indeed, the first equality comes from the fact that Av is vertical and  $\omega$  coincides with  $\omega_F$  on a pair of vertical vector fields. The middle equality follows from the compatibility of the deformation with the fibrewise symplectic structure (2.6). The last equality follows from the fact that v is horizontal. So Av is horizontal, since the relation (2.12) holds for any  $u \in V$ ; but Av is also vertical. Thus Av = 0. This proves the first claim.

We prove the second claim. While  $\partial_{w^a}$ ,  $\partial_{\bar{w}^a}$  are vertical vector fields on X,  $\partial_{z^j}$ ,  $\partial_{\bar{z}^j}$  are not horizontal in general. So we have a splitting

 $\partial_{\bar{z}^j} = \varepsilon_{\bar{i}} + \eta_{\bar{i}}$  with  $\varepsilon_{\bar{i}} \in \mathcal{H}^{\omega}, \eta_{\bar{i}} \in \mathcal{V}.$ 

Then from  $\omega(\partial_{w^a}, \varepsilon_{\overline{i}}) = 0$  it follows that

$$(\omega)_{a\bar{b}} \eta^b_{\bar{j}} = (\omega)_{a\bar{j}}.$$

So  $\eta_{\bar{i}} = (\omega_F)^{a\bar{c}} (\omega_X)_{a\bar{i}} \partial_{\bar{w}^c}$ . Thus we can write the horizontal part of  $\partial_{\bar{z}^j}$  as

$$\varepsilon_{\bar{j}} = \partial_{\bar{z}^j} - (\omega_F)^{ac}(\omega)_{a\bar{j}}\partial_{\bar{w}^c}.$$

#### 2.3 Deformations of fibrations

Since *A* takes value in the vertical vector fields,  $A(\varepsilon_{i}) = 0$ , so

$$0 = -A^a_{\ \bar{c}}(\omega_F)^{d\bar{c}}(\omega)_{d\bar{j}}\partial_{w^a} + A^a_{\ \bar{j}}\partial_{w^a},$$

hence the claim.

The following lemma explains the relation between a  $J \in \mathscr{J}_{\pi}$  and the splitting (2.1) of the tangent bundle of *X* induced by  $\omega$ , by showing that the elements of  $\mathscr{J}_{\pi}$  in a neighborhood of *I* differ from *I* only on the vertical vector bundle.

**Lemma 2.13.** Any  $J \in \mathscr{J}_{\pi}$  preserves the splitting of the tangent space  $TX = \mathcal{V} \oplus \mathcal{H}^{\omega}$ . Moreover, J(u) = I(u) for all  $u \in \mathcal{H}^{\omega}$ .

*Proof.* For the first fact to hold, we have to prove that  $J(\mathcal{V}) \subseteq \mathcal{V}$  and  $J(\mathcal{H}^{\omega}) \subseteq \mathcal{H}^{\omega}$ .

1. Let  $v \in \mathcal{V}$ . Then

$$d\pi(Jv) = J_B(\underbrace{d\pi(v)}_{=0}) = 0,$$

so  $Jv \in \mathcal{V}$ .

2. Let  $u \in \mathcal{H}^{\omega}$ . Then  $\omega(u, v) = 0$  for every  $v \in \mathcal{V}$ . So

$$\omega(Ju, v) = -\omega(u, Jv) = 0,$$

since *Jv* is vertical by the previous step.

We now prove that indeed J(u) = I(u) for all  $J \in \mathscr{J}_{\pi}$  and all  $u \in \mathcal{H}^{\omega}$ . Consider for instance a first order deformation  $I + \varepsilon A$ . Since A vanishes on horizontal vector fields by Lemma 2.12, if  $u \in \mathcal{H}^{\omega}$ ,  $(I + \varepsilon A)(u) = I(u)$ . Let now  $J_s$  be a path in  $\mathscr{J}_{\pi}$ which joins I to J. Then

$$\partial_s J_s(\mathcal{H}^{\omega}) = A_s(\mathcal{H}^{\omega}) = 0,$$

so for all *s* we have  $J_s(u) = I(u)$ , for all  $u \in \mathcal{H}^{\omega}$ , from which the claim follows.  $\Box$ 

In particular, the last part of the proof shows that the horizontal parts of the operators  $\partial$ ,  $\bar{\partial}$  with respect to *I* remain the same for any *J* in  $\mathscr{J}_{\pi}$ .

*Remark* 2.14. Let  $k \gg 0$  be such that  $\omega + k\omega_B$  is a Kähler form on X. Then  $\mathcal{J}_{\pi}$  embeds into  $\mathcal{J}(\omega + k\omega_B)$ . Indeed for  $J \in \mathcal{J}_{\pi}$ 

$$\omega_k(J, J) = \omega(J, J) + k\pi^* \omega_B(J, J)$$

and  $\pi^* \omega_B(J, J) = \omega_B(d\pi J, d\pi J) = \omega_B(J_B d\pi, J_B d\pi) = \pi^* \omega_B(\cdot, \cdot)$ .

#### 2.3.1 Families of holomorphic submersions

In this section, we give a more rigorous definition of a family of fibrations and we prove a relative version of Ehresmann's fibration theorem. Families of fibrations are in particular families of holomorphic maps, for which a deformation theory has been developed by Horikawa [41, 42], under the assumption that  $H^2(X, TX)$  embeds into  $H^2(X, \pi^*TB)$ .

**Definition 2.15.** A family of holomorphic submersions onto *B* with central fibre  $X \rightarrow B$  is the data of  $(X, \hat{\pi}, p, S)$ , where:

- 1. *X*, *S* are complex manifolds;
- 2.  $p: X \to S$  and  $\hat{\pi}: X \to B \times S$  are proper holomorphic submersions;
- 3. there is a distinguished point  $0 \in S$  such that  $\widehat{\pi}_0$  induces  $\pi : X \to B$  and  $p = \operatorname{pr}_2 \circ \widehat{\pi}$ .

A family of holomorphic maps  $(X, \hat{\pi}, p, S)$  onto *B* is *complete* if for any other family  $(X', \hat{\pi}', p', S')$  with the same central fibre, there exists a map  $\tau : S' \to S$ , defined locally on neighbourhoods of the distinguished points, such that the family  $(X', \hat{\pi}', p', S')$  can be obtained by  $(X, \hat{\pi}, p, S)$  via pull-back using  $\tau$ .

In order to parametrise compatible almost complex structures, we need to take into account that all the fibres admit a relative Kähler metric in the same cohomology class as the central fibre  $(X, \omega)$ . Following Schumacher [70], we introduce the definition of a polarised family.

**Definition 2.16.** A polarised family  $(p : \mathcal{M} \to S, \gamma)$  is a family of compact complex manifolds with a section  $\gamma \in \Gamma(S, R^2p_*\mathbb{R})$  such that  $\gamma|_{M_t} \in H^2(M_t, \mathbb{R})$  is a Kähler class. Analogously, a polarised family of maps onto *B* is a family of maps  $(\mathcal{X}, \hat{\pi}, p, S)$  with a section  $\gamma \in \Gamma(S, R^2p_*\mathbb{R})$  such that  $\gamma|_{X_s} \in H^2(X_s, \mathbb{R})$  is a relative Kähler class with respect to the projection  $\hat{\pi}_s : X_s \to B$ .

The following theorem guarantees that a polarised family exists for a Kähler manifold (M,  $\gamma_0$ ). In the next section, we will prove an analogous result for holomorphic submersions.

**Theorem 2.17** (Schumacher [70]). Let  $(X, \gamma_0)$  be a Kähler manifold, with  $\lambda_0 \in H^2(X, \mathbb{R})$ a Kähler class. Let  $p : X \to S$  be a versal family with central fibre  $(X, \lambda_0)$  (which exists by Kuranishi's Theorem). Then there exists  $S' \subset S$  such that  $\lambda_0$  can be extended to a global section  $\gamma \in \Gamma(S', \mathbb{R}^2 p_*\mathbb{R})$ , thus  $(X \to S', \gamma)$  is a polarised family.

We conclude this section with a proof of a relative version of Ehresmann's fibration theorem [44, Proposition 6.2.2]. For a smooth proper morphism  $p : M \rightarrow N$  with N connected, Ehresmann's fibration theorem says that the fibres of p are all diffeomorphic, and more precisely that M is locally diffeomorphic to a product. To generalise it to fibrations, we need the following definition.

#### 2.3 Deformations of fibrations

**Definition 2.18.** Let  $F : M \to N$  be a smooth map, u a vector field on M, u' a vector field on N. We say that u and u' are *F*-related if

$$F \circ \phi_{u'}^t = \phi_u^t \circ F.$$

Then we can extend Ehresmann's theorem 1.25 to our setting as follows.

**Proposition 2.19** (Relative Ehresmann's Theorem). Let  $(X, \hat{\pi}, p, S)$  be a family of holomorphic maps onto B with S connected. Then there exists a diffeomorphism  $\tau : X \xrightarrow{\sim} X \times S$  which commutes with the projections to B, *i.e.* 

$$\begin{array}{ccc} \mathcal{X} & \xrightarrow{\tau} & X \times S \\ \widehat{\pi} & & & & \\ \widehat{\pi} & & & & \\ \mathcal{B} \times S & & & \\ \end{array}$$

*Proof.* Up to restricting to a segment, we can assume that *S* is a small open neighbourhood of the origin in  $\mathbb{R}$ . We can then consider the vector field  $u = \partial_s$ , and view it as a vector field in  $B \times S$ , denoted by u'. This means that, denoting by  $\phi_{u'}^t$  its flow and  $\operatorname{pr}_2 : B \times S \to S$  the second projection, u' is  $\operatorname{pr}_2$ -related to u. It is a consequence of the implicit function theorem that if  $F : M \to N$  is a smooth *submersion* of manifolds, then for any vector field on N there exists a vector field on M which is  $\overline{r}$ -related to it. Let then v be a vector field on X which is  $\overline{\pi}$ -related to u', i.e.

$$\widehat{\pi}\circ\phi_v^t=\phi_{u'}^t\circ\widehat{\pi}.$$

Then, using  $p = \text{pr}_2 \circ \hat{\pi}$ , we obtain

$$p \circ \phi_v^t = \phi_u^t \circ p,$$

thus v is p-related to u. Hence we can define a map

$$\begin{aligned} \tau : \mathcal{X} &\longrightarrow X \times S \\ z &\longmapsto (\phi_v^{-t}(z), p(z)), \end{aligned}$$

which is a diffeomorphism with inverse

$$(x,s) \mapsto \phi_v^s(x).$$

Since *v* is  $\hat{\pi}$ -related to *u'*, this diffeomorphism commutes with the projections to *B*, as required.

We can then formalise the degeneration family of fibrations introduced in §2.2.2. Let  $(X, H_X) \rightarrow (B, L)$  be a polarised holomorphic submersion with  $\omega \in c_1(H_X)$  a relatively Kähler metric. Then we consider the following setting:  $(X, \mathcal{H}, \hat{\pi}, p, S)$  is a smooth *polarised family* of maps onto *B* with central fibration  $(X, H_X) \rightarrow B$ , where we can assume for simplicity that *S* is a disk in  $\mathbb{C}$ . In particular, the line bundle

#### Holomorphic submersions

 $\mathcal{H}$  on X restricts to a relatively ample line bundle  $\mathcal{H}_s$  on each fibration  $X_s \to B$ . Ehresmann's theorem implies that we can locally trivialise the family in such a way that all  $X_s$  are diffeomorphic, so we can interpret the family as a family of almost complex structures  $\{J_s\}$  on X which preserve the projection onto B.

Moreover, since  $(X_s, H_s)$  is a small deformation of  $(X, H_X)$ , we have that  $c_1(H_Y) = c_1(H_X)$ . Then by Moser's theorem [8, Theorem 7.2] we can assume that  $\omega$  is relatively Kähler with respect to the complex structures  $J_s$ , up to modifying  $(X_s, H_s)$  by a small diffeomorphism. So we can view a family  $X \to B \times S$  as a family of complex structures on X which keep  $\pi_X$  a holomorphic submersion and  $\omega$  a relatively Kähler metric.

*Remark* 2.20. A possible example of a degeneration family  $(X, \mathcal{H}, \hat{\pi}, p, S)$  is obtained by assuming that *S* is one-dimensional and there is an action of  $\mathbb{C}^*$  on  $S \times B$  which lifts to  $(X, \mathcal{H})$  such that  $\hat{\pi}$  is  $\mathbb{C}^*$ -equivariant. It follows that for  $s \neq 0$  the fibrations  $(X_s, \mathcal{H}_s) \rightarrow B$  are all biholomorphic. In this context, one can think of the family  $(X, \mathcal{H}, \hat{\pi}, p, S)$  as a family of test configurations varying holomorphically over *B*.

#### 2.3.2 Relative Kuranishi's Theorem

As in the previous section, we consider a holomorphic submersion  $\pi_X : (X, H_X) \rightarrow (B, L)$  with a relative Kähler metric  $\omega$  and a complex structure I which is fibrewise cscK. We require a relative version of Székelyhidi's and Brönnle's deformation theory described in §1.4. Let  $\mathscr{J}_{\pi}$  the space of compatible almost complex structures defined in (2.5) and  $T_J \mathscr{J}_{\pi}$  be its tangent space at a point J. If  $A \in T_J \mathscr{J}_{\pi}$  also  $JA \in T_J \mathscr{J}_{\pi}$ , so  $\mathscr{J}_{\pi}$  admits an almost complex structure. Consider the map

$$P_{\mathcal{V}}: C_0^{\infty}(X, \mathbb{R}) \longrightarrow T_I^{0,1} \mathscr{J}_{\pi}$$
$$\varphi \longmapsto \bar{\partial}(\operatorname{grad}^{\omega_F} \varphi)^{1,0},$$

which is the relative version of the map defined in (1.7). Let  $\widetilde{H}_{V}^{1}$  be the kernel in  $T_{I}^{0,1} \mathscr{J}_{\pi}$  of the operator

$$\Box_{\mathcal{V}} = P_{\mathcal{V}} P_{\mathcal{V}}^* + (\bar{\partial}^* \bar{\partial})^2$$

inside  $T_I^{0,1} \mathscr{J}_{\pi}$ . This is an elliptic operator because  $P_{\mathcal{V}} P_{\mathcal{V}}^*$  is trivial in horizontal directions, where the adjoint is computed with respect to any Kähler metric on X which restricts to  $\omega_F$  vertically. So its kernel is a finite dimensional vector space and it can be described as

$$\widetilde{H}^1_{\mathcal{V}} = \left\{ \alpha \in T^{0,1}_I \mathcal{J}_\pi \mid P^*_{\mathcal{V}} \alpha = 0 = \bar{\partial} \alpha \right\}.$$

Fibrewise,  $\Box_V$  restricts to the operator (1.16) and  $\widetilde{H}_V^1$  restricts to the vector space described in (1.15). In particular,  $\widetilde{H}_V^1$  depends only on the vertical part of the metric,  $\omega_F$ .

Consider the smooth fibre bundle  $\mathcal{K} \to B$  whose fibre  $K_b$  is the stabiliser of  $I|_{X_b}$  for the  $\mathscr{G}_b$ -action. in particular,  $K_b$  coincides with the group  $\text{Isom}(X_b, \omega_b)$  of

#### 2.3 Deformations of fibrations

biholomorphic isometries of the fibre. Thanks to our hypothesis 2, the groups  $K_b$  are finite-dimensional with dimension independent of b. The group of global sections of  $\mathcal{K}$  is

$$K_{\pi} := \operatorname{Isom}(\pi_X, \omega) = \left\{ f \in \operatorname{Aut}(X) \mid f^* \omega = \omega \text{ and } \pi_X \circ f = \pi_X \right\}.$$
(2.13)

We next prove a relative version of Kuranishi Theorem, adapted from Chen-Sun [12, §6].

**Theorem 2.21** (Relative Kuranishi Theorem). *There exists a neighborhood of the origin*  $V_{\pi} \subset \widetilde{H}^{1}_{\alpha}$  and a  $K_{\pi}$ -equivariant holomorphic map

$$\Psi: V_{\pi} \to \mathscr{J}_{\pi}$$

such that:

- 1.  $\Psi(0) = I;$
- 2. If  $v_1, v_2 \in V$  and  $v_1|_b \in K_b^{\mathbb{C}} \cdot v_2|_b$  for all b, and if  $\Psi(v_1)$  is integrable, then  $\Psi(v_1)|_{X_b}$  is in the same  $\mathscr{G}_b^c$ -orbit as  $\Psi(v_2)|_{X_b}$ ;
- 3. For any  $J \in \mathcal{J}_{\pi}$  integrable close to I, there exists J' in the image of  $\Psi$  such that, for all b,  $J'_{h}$  is in the same  $\mathscr{G}_{h}^{c}$ -orbit as  $J_{b}$ .

*Proof.* We can identify any *J* close to *I* with an element  $\alpha \in T_I^{0,1} \mathscr{J}_{\pi}$ , i.e. with a (0, 1)-form with values in the vertical holomorphic tangent bundle compatible with  $\omega_F$ . So we have an embedding from an open subset in  $T_I^{0,1} \mathscr{J}_{\pi}$  into  $\mathscr{J}_{\pi}$ :

$$f: \mathcal{U}(T_I^{0,1}\mathscr{J}_\pi) \hookrightarrow \mathscr{J}_\pi.$$

Given  $b \in B$ , we denote by  $\rho_b$  the restriction  $\mathscr{J}_{\pi} \to \mathscr{J}(X_b)$ . Then  $f_b(\alpha \mid_{X_b}) = \rho_b \circ f(\alpha)$ . We define now a new embedding  $\hat{f} : \mathcal{U}(T_1^{0,1} \mathscr{J}_{\pi}) \hookrightarrow \mathscr{J}_{\pi}$  as follows:

$$\hat{f}(\alpha) = \int_{\mathcal{K}/B} g^{-1} f(g \cdot \alpha) \mathrm{d}\mu_{\mathcal{K}/B}(g),$$

where  $d\mu_{\mathcal{K}/B}$  is the fibrewise Haar measure on  $\mathcal{K} \to B$ . Then  $\hat{f}$  is such that

$$\hat{f}(k \cdot \alpha) \mid_{b} = k \mid_{b} \cdot \hat{f}(\alpha) \mid_{b}$$
.

Now,  $\alpha$  is an integrable deformation if and only if it satisfies [44, Lemma 6.1.2]

$$N(\alpha) = \bar{\partial}\alpha + [\alpha, \alpha] = 0. \tag{2.14}$$

Note that if  $\alpha$  is integrable its restriction to each fibre is also integrable, so equation (2.14) holds also fibrewise. For any  $b \in B$ , let  $H_b : T_I^{0,1} \mathscr{J}_{\pi} |_{X_b} \to \widetilde{H}_b^1$  be the

 $L_k^2$ -orthogonal projection and let  $G_b$  be the Green operator of  $\Box_b$ , defined by the condition

$$\mathbb{1} = G_b \square_b + H_b = \square_b G_b + H_b$$

Let  $\alpha$  be an integrable deformation. A simple computation starting from (2.14) leads to the identity

$$\alpha \mid_b + G_b \bar{\partial}_b^* \bar{\partial}_b \bar{\partial}_b^* [\alpha \mid_b, \alpha \mid_b] = H_b \alpha \mid_b.$$

Then we can define a map

$$F: B \times T_I^{0,1} \mathscr{J}_{\pi} \to T_I^{0,1} \mathscr{J}_{\pi}$$
$$(b, \alpha) \mapsto \alpha \mid_b + G_b \bar{\partial}_b^* \bar{\partial}_b \bar{\partial}_b^* [\alpha \mid_b, \alpha \mid_b],$$

where  $T_I^{0,1} \mathscr{J}_{\pi}$  is endowed with the Sobolev  $L_k^2$ -norm. The differential of F in the second component at the origin is the identity, since  $G_b \bar{\partial}_b^* \bar{\partial}_b \bar{\partial}_b^* [\alpha \mid_b, \alpha \mid_b]$  is quadratic in  $\alpha$ . Hence by the implicit function theorem we can locally invert F and the inverse varies smoothly with b. We consider the inverse restricted to an open ball in  $\tilde{H}_{\mathcal{V}}^1$ , which we define to be  $V_{\pi}$ , and for  $x \in V_{\pi}$  we denote it by  $\alpha(x)$ . Thus we have a family

$$U := \{ \alpha(x) \mid x \in V_{\pi} \} \subset T_{I}^{0,1} \mathscr{J}_{\pi},$$
(2.15)

and we can define

$$\Psi: V_{\pi} \to \mathscr{J}_{\pi}$$
$$x \mapsto \hat{f}(\alpha(x)).$$

We begin by proving that this map satisfies the required properties. Denoting

$$U^{\text{int}} = \{ \alpha(x) \mid N(\alpha(x)) = 0 \}$$

and

$$U_{\mathcal{U}}^{\text{int}} = \{ \alpha(x) \mid N_b(\alpha(x) \mid_b) = 0 \ \forall b \in B \},\$$

we want to prove that  $U^{\text{int}}$  is an analytic subset of U. We begin by showing that  $U_V^{\text{int}}$  is an analytic subset of U. On each fibre  $X_b$ ,  $\alpha(x) \mid_b$  is integrable if and only if  $H_b[\alpha(x) \mid_b, \alpha(x) \mid_b] = 0$ . Indeed

$$N_{b}(\alpha(x)|_{b}) = \partial_{b}\alpha(x)|_{b} + [\alpha(x)|_{b}, \alpha(x)|_{b}]$$
  
=  $2G_{b}\bar{\partial}_{b}^{*}\bar{\partial}_{b}\bar{\partial}_{b}^{*}[N(\alpha(x)|_{b}), \alpha(x)|_{b}] + H[\alpha(x)|_{b}, \alpha(x)|_{b}].$  (2.16)

The map

$$B \times V_{\pi} \to \overline{H}_{\mathcal{V}}$$
$$(b, v) \mapsto H_{b}[\alpha(x) \mid_{b}, \alpha(x) \mid_{b}]$$

is holomorphic, so  $U_V^{\text{int}}$  is an analytic subset of U. Then, denoting  $\overline{U}^{\text{int}}$  the analytic family given by the Kuranishi Theorem [12, Lemma 6.1] applied to X, we see that

#### 2.3 Deformations of fibrations

 $U^{\text{int}}$  is the intersection of  $\overline{U}^{\text{int}}$  and  $U_V^{\text{int}}$ , so it is itself an analytic family. Moreover, when restricted to the fibre  $X_b$ , both maps f and F are  $K_b$ -equivariant and holomorphic, so (2) is also proved.

We prove (3). Let  $J_x = \Psi(x) \in \mathcal{J}_{\pi}$ , and fix  $b \in B$ . Given  $\xi \in \Gamma(X, \mathcal{V})$  a vertical vector field, define

$$F_{\xi}: X \to X$$
$$p \mapsto \exp_p(\xi_p, g_{\pi(p)}),$$

where  $g_{\pi(p)}$  is the Riemannian metric on the fibre  $X_{\pi(p)}$  with respect to  $J_x$ . Following [12, Lemma 6.1], we fix  $v \in V_{\sigma} \subset V_{\pi}$ , thought of as a tangent vector at x in  $V_{\pi}$ , and we define a map

$$R_b: \mathcal{U}(L^2_{k+2}(X_b, \mathbb{C})) \to L^2_{k+2}(X_b, TX_b)$$
$$\varphi_b \mapsto \xi_b(\varphi_b, v\mid_b)$$

such that  $R_b(0) = 0$  and

1. 
$$d_0 R_b(\varphi_b) = \operatorname{grad}^{\omega_b}(\operatorname{Re}(\varphi_b)) + J_v \mid_{X_b} \operatorname{grad}^{\omega_b}(\operatorname{Im}(\varphi_b));$$
  
2.  $F^*_{\xi_b(\varphi_b, v\mid_b)} J_x \mid_{X_b} \in \mathscr{G}^c_b \cdot J_x \mid_{X_b}.$ 

Since this map is defined via the implicit function theorem, the vector field  $\xi_b(\varphi_b, v \mid_b)$  varies smoothly with b, thus defining a global vertical vector field on X (more details about this technique of using the implicit function theorem to prove smooth dependence on b are given in the proof of Proposition 2.23 below). So we can define a global map

$$R: \mathcal{U}(L^2_{k+2}(X,\mathbb{C})) \to L^2_{k+2}(X,\mathcal{V})$$
$$\varphi \mapsto \xi(\varphi,v) \quad \text{s.t.} \quad \xi(\varphi,v) \mid_{X_b} = \xi(\varphi \mid_b, v \mid_b).$$

The complex structure  $F^*_{\xi(\omega,v)}J_x$  on *X* satisfies the following properties:

1. it is compatible with  $\omega$ . Indeed

$$\omega(F^*_{\xi(\varphi,v)}J_x\cdot,\cdot) + \omega(\cdot,F^*_{\xi(\varphi,v)}J_x\cdot) = \\ \omega_F(F^*_{\xi(\varphi,v)}J_x\cdot,\cdot) + \omega_F(\cdot,F^*_{\xi(\varphi,v)}J_x\cdot) + \omega_{X,\mathcal{H}}(J_x\cdot,\cdot) + \omega_{X,\mathcal{H}}(\cdot,J_x\cdot).$$

The first two terms sum to zero because the complex structure  $F^*_{\xi(\varphi,v)}J_x$  is fibrewise compatible with the fibrewise Kähler form. The last two terms sum to zero since  $J_x \in \mathscr{J}_{\pi}$ ;

- 2. it preserves  $\pi$ , since the differential commutes with pull-back;
- 3. it satisfies property 2 above for every  $b \in B$ .

Let now  $\alpha(\varphi, v)$  be the (0, 1)-form with values in the holomorphic vertical tangent space which is the pre-image of  $F^*_{\xi(\varphi,v)}J_x$  via  $\Psi$ . Then from (2.16) it follows that  $\alpha(\varphi, v)$  fibrewise satisfies an elliptic equation of the form

$$\Box_{\mathcal{V}} T(\varphi, v, N(\alpha)) = 2\bar{\partial}_{\mathcal{V}}^* \bar{\partial}_{\mathcal{V}} \bar{\partial}_{\mathcal{V}}^* \left[ T(\varphi, v, N(\alpha)), S(\varphi, v, \alpha) \right], \qquad (2.17)$$

where T(0, v, N) = N and  $S(0, v, \alpha) = \alpha$ .

Let now  $J \in \mathscr{J}_{\pi}^{\text{int}}$  be close to I in  $L_k^2$ . The proof now goes exactly as in [12, Lemma 6.1], and we report it here for completeness. Since J is integrable, the corresponding vector-valued  $(0, 1)_I$ -form  $\alpha_J$  satisfies (2.17) for all  $(\varphi, v)$ . Consider the  $L_k^2$ -projections

$$\Pi_1: L^2_k\left(T^{0,1}_I \mathscr{J}_{\pi}\right) \to \operatorname{Im}(P_{\mathcal{V}}), \qquad \Pi_2: L^2_k\left(T^{0,1}_I \mathscr{J}_{\pi}\right) \to \widetilde{H}^1_{\mathcal{V}}$$

and consider the map  $\chi : \mathcal{U}(L^2_{k+2}(X,\mathbb{C})) \times V_{\sigma} \to \operatorname{Im}(P_{\mathcal{V}}) \times \widetilde{H}^1_{\mathcal{V}}$  defined by

$$(\varphi, v) \mapsto \left( \Pi_1 \left( F^*_{\xi(\varphi, v)} J_v \right), \Pi_2 \left( F^*_{\xi(\varphi, v)} J_v \right) \right)$$

Remark that if  $\alpha, \beta \in T_I^{0,1} \mathscr{J}_{\pi}$  satisfy (2.17) and they are such that  $(\Pi_1 \alpha, \Pi_2 \alpha) = (\Pi_1 \beta, \Pi_2 \beta)$ , then by ellipticity it follows that  $\alpha = \beta$ . The differential of the map  $\chi$  is  $d_{(0,0)}\chi(\varphi, v) = (P_V(\varphi), v)$ : it is surjective and the kernel corresponds to fibrewise holomorphy potentials, so it is finite dimensional fibrewise. Thus again by the implicit function theorem, there exist  $(\varphi, v)$  such that  $(\Pi_1(\alpha_J), \Pi_2(\alpha_J)) = \chi(\varphi, v)$ , hence by the ellipticity argument  $\alpha_J = F^*_{Y(\varphi, v)}J_x$ .

*Remark* 2.22 (Versal deformations). The proof of the relative Kuranishi Theorem guarantees the existence of *versal* deformations. Recall from Definition 1.24 that deformation  $X \to B \times V_{\pi}$  with central fibre (X, I) is called versal if any other family  $X' \to B \times V'_{\pi}$  (centred at *I*) is obtained by pullback via a map  $f : V'_{\pi} \to V_{\pi}$ , which might not be unique but whose differential is uniquely determined. This is proven in the third step of Theorem 2.21, where a single complex structure *J* is considered instead of a second family  $\{J_{t'}\}$ . The pullback is given by the exponential map  $F_{\xi}$ , where  $\xi = \xi(\varphi, v)$  is uniquely determined by the vector *v* tangent to the complex structure *J*.

**Proposition 2.23.** Possibly after shrinking  $V_{\pi}$ , we can perturb the map  $\Psi$  to

$$\Phi: V_{\pi} \to \mathscr{J}_{\pi} \tag{2.18}$$

such that

$$\operatorname{Scal}_{V}(\omega, \Phi(x)) - \widehat{S}_{b} \in C^{\infty}(E).$$

Remark that the claim holds fibrewise as a consequence of Theorem 1.23, so we just need to check that the complex structure we find on each fibre  $X_b$  varies smoothly with b. This relies on the fact that the proof involves the implicit function theorem.

#### 2.3 Deformations of fibrations

*Proof of Proposition 2.23.* For every  $b \in B$ , the Lie algebra of  $K_b$  is  $\mathfrak{t}_b = \ker \mathcal{D}_b^* \mathcal{D}_b$ , which is exactly the fibre  $E_b$  of the vector bundle E defined in §2.1. Let  $U_l$  be a small ball around the origin of  $L^{2,l}(R)$ . Define a map

$$G: B \times V_{\pi} \times U_{l} \to L^{2,l-4}(R)$$
  
(b, x, \varphi) \mapsto \pi\_{L^{2,l-4}(R)} \mapsto \mapsto al \left(\omega\_{b}, F\_{\varphi\_{b}}(\Psi(x) \mi\_{b})\right),

where  $\Psi(x) |_b$  defines an element in  $\mathscr{J}_{\pi}$  from Theorem 1.23, and the map *F* is the one defined in (1.14). The derivative along the third component of *G* at 0 of a function  $\varphi$  is given by  $P^*_{\mathcal{V}}P_{\mathcal{V}}(\varphi)$ , which is an isomorphism  $L^{2,l}(R) \to L^{2,l-4}(R)$ . By the implicit function theorem, for every b,  $\Psi$  can be perturbed to  $\Phi_b : V_b \to \mathscr{J}(X_b)$ in such a way that  $\text{Scal}(\omega_b, \Phi_b(x)) \in \mathfrak{t}_b$ , and  $\Phi_b$  varies smoothly with b. Thus we find a map

$$\Phi: V_{\pi} \to \mathscr{J}_{\pi}$$

such that  $\operatorname{Scal}_{\mathcal{V}}(\omega, \Phi(x)) - \widehat{S}_b \in C^{\infty}(E)$ .

Let us return now to considering a holomorphic submersion  $\pi_X : X \to B$  with a relatively cscK metric ( $\omega$ , I). By viewing  $\omega$  as fixed and varying the complex structure, we consider a family { $J_s$ } such that ( $\omega$ ,  $J_s$ ) are relatively Kähler metrics on  $X \to B$ . Theorem 2.21, together with Proposition 2.23, allow us to extend definition of the sections  $\mu_s$  (2.7) and  $\nu$  (2.8) of  $C^{\infty}(E)$  to the following maps.

**Definition 2.24.** Let  $\mu_{\pi}$  be the map

$$\mu_{\pi}: V_{\pi} \to C^{\infty}(E, I)$$
$$x \mapsto \operatorname{Scal}_{V}(\omega, \Phi(x)) - \widehat{S}_{b}$$

and let  $\nu_{\pi}$  be the map

$$\begin{aligned} \nu_{\pi} : \widetilde{H}^1 \to C^{\infty}(E) \\ v \mapsto \nu_{\pi}(v), \end{aligned}$$

where  $v_{\pi}(v) \mid_{b} = v_{b}(v \mid_{X_{b}})$  and  $v_{b}$  is the map defined in Definition 1.26.

*Remark* 2.25. If  $x_s \in V_{\pi}$  corresponds to  $J_s$  via the relative Kuranishi map (2.18), we have that  $\mu_{\pi}(x_s) = \mu_s$ , where  $\mu_s \in C^{\infty}(E)$  is the section defined in (2.7). Similarly, if  $v \in \widetilde{H}^1_V$  is the deformation of the family  $\{J_s\}$ , then  $\nu_{\pi}(v)$  is the section  $v \in C^{\infty}(E)$  defined in (2.8). By applying Proposition 2.23 we can perturb  $\mu_{\pi}$  to end up in  $C^{\infty}(E)$ , so we do not see the projection as in (2.7).

From the definition (2.7), the perturbation given by Proposition 2.23 and the expansion (2.9) of  $\mu_s \in C^{\infty}(E)$  it follows that, if  $v \in \widetilde{H}^1_V$  is the deformation of the family  $\{J_s\}$ ,

$$\operatorname{Scal}_{\mathcal{V}}(\omega, J_s) - \widehat{S}_b = \mu_{\pi}(x_s) = \frac{s^2}{2} \nu_{\pi}(v) + O(s^3).$$
 (2.19)

This expansion will be used in §3.1, where we derive the optimal symplectic connection equation from an expansion of the scalar curvature. Holomorphic submersions

## **Chapter 3**

## Extremal metrics on the total space

As before, let  $\widehat{\pi}$  :  $(X, \mathcal{H}) \to (B, L) \times S$  be a degeneration of a fibration  $\pi_Y$  :  $(Y, H_Y) \to B$  with central fibration  $\pi_X : (X, H_X) \to B$ , endowed with a  $\mathbb{C}^*$ -action on  $B \times S$  which lifts to  $(X, \mathcal{H})$ . Let  $\omega$  be a relatively cscK metric on X. As in §2.2.2, we can assume that  $\omega$  is relatively Kähler also on Y. It follows that the general fibrations  $X_s \to B \times \{s\}$  are all biholomorphic to  $Y \to B$ . For  $k \gg 0$ , consider the Kähler form

$$\omega_k = \omega + k\omega_B,$$

where  $\omega_B$  is a fixed Kähler metric on *B*. In this chapter, we obtain the optimal symplectic connection from the expansion of the scalar curvature in powers of *k*. We then compute the linearisation of the optimal symplectic operator and we use optimal symplectic connections to construct constant scalar curvature and extremal Kähler metrics in the class  $c_1(H_Y) + kc_1(L)$ .

### **3.1** Expansion of the scalar curvature

In this subsection, we derive an expansion of the scalar curvature  $Scal(\omega_k, J_s)$ , in powers of *s* and inverse powers of *k*, from which we deduce the optimal symplectic connection equation (2.10). Recall from [14, §4.1] that

 $\operatorname{Scal}(\omega_k, J_s) = \operatorname{Scal}_{\mathcal{V}}(\omega, J_s) + k^{-1} \left( \operatorname{Scal}(\omega_B) + \Delta_{\mathcal{V}}(\Lambda_{\omega_B} \omega_{\mathcal{H}}) + \Lambda_{\omega_B} \rho_{\mathcal{H}} \right) + O\left(k^{-2}\right).$ 

Clearly, the  $k^{-1}$  term - denoted  $T_{k^{-1}}$  - depends on *s*, so we can write

$$T_{k^{-1}}(\omega_B, \omega, J_s) = T_{k^{-1}}(\omega_B, \omega, I) + O(s)$$

**Proposition 3.1.** By choosing  $s^2 = \lambda k^{-1}$  for  $\lambda > 0$  and using the expansion (2.19) for the vertical scalar curvature we obtain

$$\operatorname{Scal}(\omega_k, J_s) = \widehat{S}_b + k^{-1} \left( \psi_B + p_E(\Delta_{\mathcal{V}}(\Lambda_{\omega_B}\omega_{\mathcal{H}}) + \Lambda_{\omega_B}\rho_{\mathcal{H}}) + \frac{\lambda}{2}\nu(v) + \psi_R \right) + O\left(k^{-3/2}\right)$$

where:

1.  $\psi_B$  is a function on the base given by

$$\psi_B = \operatorname{Scal}(\omega_B) + \int_{X/B} (\Lambda_{\omega_B} \rho_{\mathcal{H}}) \omega^m.$$

2.  $\psi_R \in C^{\infty}(R, I)$ .

*Proof.* For the first item, in [14, §4.1] based on [26, Lemma 2.3] it is shown that

$$\int_{X/B} (\Lambda_{\omega_B} \rho_{\mathcal{H}}) \omega^m = -\Lambda_{\omega_B} \alpha_{\mathrm{WP}},$$

where  $\alpha_{WP}$  is the Weil-Petersson metric defined in (1.24). Moreover, the  $k^{-1}$ -term depends only on *I* because the O(s)-part ends up in  $O(k^{-3/2})$ . Its expression is obtained following [14, §4.2].

Thus the optimal symplectic connection equation implies that the  $C^{\infty}(E)$ -part of the  $k^{-1}$ -term of the expansion of the scalar curvature vanishes. Note that  $\widehat{S}_b$  is a topological constant independent of b, since all the fibres are diffeomorphic.

*Remark* 3.2. For simplicity, we have assumed that all the manifolds involved are projective varieties. This is not a strictly necessary assumption, as everything could be carried out in the Kähler case: instead of a relative polarisation on  $X_s$ , we fix a relative Kähler class  $\alpha_s$  and on the base a Kähler class  $\beta$ . Then the constant  $\hat{S}_b$ , scalar curvature of the fibres of  $X = X_0$ , is still independent of b. Indeed locally on a chart  $U \subseteq B$ , X is diffeomorphic to  $F \times U$ , where  $(F, \alpha_F)$  is a model fibre. Then  $\alpha$  is isomorphic to the class  $p_1^* \alpha_F$ , where  $p_1$  is the projection onto the first factor  $F \times U \to F$ , so

$$\int_{X_b} \alpha_b^m = \int_F \alpha_F^{m-1} \cdot c_1(X_b).$$

Therefore the constant

$$\widehat{S}_b = \frac{c_1(X_b) \cdot \alpha_b^{m-1}}{\alpha_b^m}$$

is independent of *b*.

#### **3.2** Linearisation of the fibrewise map *v*

We restrict our attention to a single fibre, so we consider a manifold  $(M, \omega)$ , where I is a cscK complex structure and  $v \in \tilde{H}^1$  is a deformation of I. We wish to linearise the map v defined in 1.26.

Let  $\varphi_E \in \ker_{\mathbb{R}}(\mathcal{D}_0^*\mathcal{D}_0)$ . Then  $\frac{1}{2}\nabla^g \varphi_E$  is a real holomorphic vector field, where g is the Riemannian metric induced by  $\omega$  and I. Let  $\rho(t)$  be the flow of the vector field

$$\xi_{\varphi_E} := \nabla^g \varphi_E = -I \operatorname{grad}^{\omega} \varphi_E.$$

We wish to study how  $\nu(v)$  changes when changing  $\omega$  to  $\rho(t)^*\omega$ , so we must compute

$$\partial_t |_{t=0} v_t(v_t) = \partial_t |_{t=0} v_t(v) + d_v v (\partial_t |_{t=0} v_t).$$
(3.1)

In this expression,  $v_t$  is the moment map for the action of  $K^{\mathbb{C}}$  defined in 1.26 computed with respect to the Kähler form  $\rho(t)^*\omega$ , and  $v_t = \rho(t)^*v$ . Remark that  $\rho(t)$  is a 1-parameter group of diffeomorphisms in  $K^{\mathbb{C}}$  because it is the flow of a holomorphic vector field that admits a holomorphy potential.

As in §1.4, let  $\Phi : V \to \mathscr{J}$  be the Kuranishi map (2.18) which maps 0 to *I*, and  $\widetilde{H}^1$  the deformation space, which we identify with the tangent space  $T_0V$ . The map  $\Phi$  is *K*-equivariant, hence locally  $K^{\mathbb{C}}$ -equivariant. In particular

$$\Phi(\rho(t)^*x) = \rho(t)^*\Phi(x) \quad \text{for } x \in V.$$

Hence the pair  $(\omega, \rho(t)^* x)$  corresponds via  $\Phi$  to a compatible pair  $(\omega, \rho(t)^* \Phi(x))$ and this also holds for our  $v \in \tilde{H}^1$ , which is itself an element of  $V_{\pi}$ . We have that

$$\partial_t |_{t=0} \rho(t)^* v = \left( \mathcal{L}_{\xi_{\varphi_E}} \mathbf{v} \right) \Big|_0, \qquad (3.2)$$

where **v** is a vector field on  $V_{\pi}$  such that **v**  $|_0 = v$ . By abuse of notation, we will often denote this derivative by  $\mathcal{L}_{\xi_{\varphi_F}} v$ .

**Lemma 3.3.** For  $v \in \widetilde{H}^1$ ,  $v_t(v) = v(v_t)$ .

*Proof.* Again, this follows from equivariance. Consider again the moment map  $\mu(x) = S(\omega, \Phi(x))$  and denote by  $\mu_t$  the map

$$\mu_t: V \to \mathfrak{k}$$
$$x \mapsto S(\omega, \rho(t)^* \Phi(x)).$$

Because  $\Phi$  and  $\mu_t$  are (locally)  $K^{\mathbb{C}}$ -invariant, and in light of the above computation, we obtain

$$S(\omega, \rho(t)^* \Phi(x)) = S(\omega, \Phi(\rho(t)^* x)) = \mu(\rho(t)^* x) = \rho(t)^* \mu(x).$$

Now let  $v \in T_0 V = \widetilde{H}^1$ . Then

$$\mu_t(sv) = \rho(t)^* \mu(sv) = \rho(t)^* \left[ \frac{s^2}{2} \nu(v) + O(s^3) \right].$$

But also

$$\mu_t(sv)=\frac{s^2}{2}v_t(v)+O(s^3).$$

Thus  $v_t(v) = \rho(t)^* v(v) = v(\rho(t)^* v)$ , as claimed.

Using this lemma and equation (3.2), the derivative (3.1) becomes

$$\partial_t \mid_{t=0} \nu_t(v_t) = 2 \operatorname{d}_v \nu\left(\mathcal{L}_{\xi_{\varphi_E}} v\right).$$

Thus using the definition of moment map we can compute the linearisation of the map  $\nu$  as follows. Letting  $\psi \in \mathfrak{k}$ ,

$$\mathrm{d}_{v}\langle v,\psi\rangle(\mathcal{L}_{\xi_{\varphi_{E}}}v)=\Omega_{0}\left(\mathcal{L}_{\xi_{\varphi_{E}}}v,\mathcal{L}_{\eta_{\psi}}v\right),$$

where  $\eta_{\psi} = \text{grad}^{\omega}\psi$ . Recall the linearised infinitesimal action induced by  $\psi \in \mathfrak{k}$  defined in (1.21), and denoted  $A_{\psi}$ . We showed in (1.22) that

$$A_{\psi}v = -\left(\mathcal{L}_{\eta_{\psi}}\mathbf{v}\right)|_{0}.$$

Thus using the definition of  $\Omega_0$ ,

$$d_{v}\langle v,\psi\rangle\left(\mathcal{L}_{\xi_{\varphi_{E}}}v\right) = \int_{M} \langle Id_{0}\Phi\left(\mathcal{L}_{\xi_{\varphi_{E}}}v\right), d_{0}\Phi\left(\mathcal{L}_{\eta_{\psi}}v\right)\rangle_{\omega} \omega^{m}$$
$$= \int_{M} \langle d_{0}\Phi\left(\mathcal{L}_{\eta_{\varphi_{E}}}v\right), d_{0}\Phi\left(\mathcal{L}_{\eta_{\psi}}v\right)\rangle_{\omega} \omega^{m}$$
$$= \int_{M} \langle d_{0}\Phi\left(A_{\varphi_{E}}v\right), d_{0}\Phi\left(A_{\psi}v\right)\rangle_{\omega} \omega^{m},$$
(3.3)

where  $\langle \cdot, \cdot \rangle_{\omega}$  is the inner product induced by the Riemannian metric  $g(\omega, I)$ .

## 3.3 Linearisation of the optimal symplectic connection equation

Let us now return to the fibration setting. Letting  $\varphi, \psi \in C^{\infty}(E)$ , by applying (3.3) and the fact that the map  $\nu_{\pi}$  defined in 2.24 is defined fibrewise, we obtain

$$\langle \mathbf{d}_{v} \nu_{\pi}(\mathcal{L}_{\xi_{\varphi}} v), \psi \rangle = \int_{X} \langle \mathbf{d}_{0} \Phi\left(A_{\varphi} v\right), \mathbf{d}_{0} \Phi\left(A_{\psi} v\right) \rangle_{\omega_{F}} \omega_{F}^{m} \wedge \omega_{B}^{n}.$$
(3.4)

Here, the map  $A_{\psi}$  acts vertically, because it is induced by the infinitesimal action of the group of holomorphic isometries of every fibre. By using equation (3.4), we obtain the following result.

**Lemma 3.4.** Let  $\widehat{\mathcal{L}}$  be the linearisation of the equation (2.10) at a solution, composed with the projection  $p_E$ . Then

$$\langle \widehat{\mathcal{L}}(\varphi), \psi \rangle = \int_{X} \langle \mathcal{R}\varphi, \mathcal{R}\psi \rangle_{\omega_{F}+\omega_{B}} \omega_{F}^{m} \wedge \omega_{B}^{n} + \lambda \int_{X} \langle d_{0}\Phi \left(A_{\varphi}v\right), d_{0}\Phi \left(A_{\psi}v\right) \rangle_{\omega_{F}} \omega_{F}^{m} \wedge \omega_{B}^{n},$$

where  $\mathcal{R}$  is the operator (2.4), which gives the linearisation of the optimal symplectic connection equation at a solution.

From this expression it follows that  $\widehat{\mathcal{L}}$  is self adjoint.

We now study the kernel of  $\widehat{\mathcal{L}}$ . Since v is fixed in our setting, we can define the maps

$$A: C^{\infty}(E) \to \widetilde{H}^{1}_{\mathcal{V}} \quad \text{and} \quad \mathcal{A}: C^{\infty}(E) \to T^{0,1}_{I} \mathscr{J}_{\pi} \psi \mapsto A_{\psi} v \qquad \qquad \psi \mapsto d_{0} \Phi \left( A_{\psi} v \right).$$
(3.5)

**Lemma 3.5.** A function  $\psi \in C^{\infty}(E)$  is in the kernel of A if and only if  $\psi$  is a fibrewise holomorphy potential with respect to all  $J_s$ , i.e.  $\psi \in C^{\infty}(E, J_s)$ .

*Proof.* Let  $\psi \in \text{ker}A$  and take  $x_s \in V_{\pi}$  is such that  $x_0 = 0$  and  $\dot{x}_0 = v$ . Then

$$0 = A_{\psi}(v) = \left. \frac{\mathrm{d}}{\mathrm{d}t} \right|_{t=0} \mathrm{d}_{0}(x \mapsto \exp(t\psi) \cdot x)(v)$$
$$= \left. \frac{\mathrm{d}}{\mathrm{d}t} \right|_{t=0} \left. \frac{\mathrm{d}}{\mathrm{d}s} \right|_{s=0} \exp(t\psi) \cdot x_{s}$$
$$= \left. \frac{\mathrm{d}}{\mathrm{d}s} \right|_{s=0} \left. \frac{\mathrm{d}}{\mathrm{d}t} \right|_{t=0} \left( \rho^{\eta_{\psi}}(t) \right)^{*} x_{s}$$
$$= \left. \frac{\mathrm{d}}{\mathrm{d}s} \right|_{s=0} \mathcal{L}_{\eta_{\psi}} J_{x_{s}}.$$

where by  $\rho^{\eta_{\psi}}(t)$  we denote the flow of the vertical vector field  $\eta_{\psi} = \text{grad}^{\omega_F}\psi$ , and all the equalities hold fibrewise since the Hamiltonian action we consider is a fibrewise action. So  $\mathcal{L}_{\eta_{\psi}}J_{x_s}$  is fibrewise constant, i.e.

$$\left(\mathcal{L}_{\eta\psi}J_{x_s}\right)_{\mathcal{V}}=\left(\mathcal{L}_{\eta\psi}J_0\right)_{\mathcal{V}}=0$$

for all *s*. This can be rephrased as

 $\bar{\partial}_{s,\mathcal{V}}\eta_{\psi} = 0$ 

for all *s*, where  $\bar{\partial}_{s,\mathcal{V}}$  is the vertical  $\bar{\partial}$ -operator computed with respect to  $J_s$ . Notice that  $\eta_{\psi}$  is a real vector field which corresponds (under the isomorphism between real vector fields and  $(1,0)_s$  vector fields) to  $J_s \nabla^{1,0}_{s,\mathcal{V}} \psi$ , where  $\nabla_{s,\mathcal{V}} \psi$  denotes the vertical vector field which on each fibre is the Riemannian gradient with respect to the fibrewise metric induced by  $(\omega_F, J_s)$ . Since  $J_s$  is integrable,

$$\left(\mathcal{L}_{\eta_{\psi}}J_{s}\right)_{\mathcal{V}}=\left(\mathcal{L}_{J_{s}\nabla_{s,\mathcal{V}}^{1,0}\psi}J_{s}\right)_{\mathcal{V}}=\left(J_{s}\mathcal{L}_{\nabla_{s,\mathcal{V}}^{1,0}\psi}J_{s}\right)_{\mathcal{V}}=0$$

so  $\psi$  is a fibrewise holomorphic potential for  $J_s$ .

**Proposition 3.6.** The kernel of  $\widehat{\mathcal{L}}$  is given by

$$\ker \widehat{\mathcal{L}} = \left\{ \psi \in C^{\infty}(E, I) \mid \overline{\partial}_{s}(\nabla^{1,0}_{s,\mathcal{V}}\psi) = 0 \; \forall s \right\},\$$

*i.e.* the functions in the kernel are those fibrewise I-holomorphy potentials which are global holomorphy potentials with respect to all J<sub>s</sub>.

*Proof.* Since  $\Phi$  is an embedding,  $d_0\Phi$  is injective, so for  $\psi \in C^{\infty}(E)$ ,  $\psi \in \ker \widehat{\mathcal{L}}$  if and only if  $\psi \in \ker \mathcal{R}$  and  $\psi \in \ker A$ .

As seen in §2.2.1, the kernel of  $\mathcal{R}$  consists of fibrewise holomorphy potentials which are also global holomorphy potentials. Thus  $\psi \in C^{\infty}(E)$  lies in ker $\widehat{\mathcal{L}}$  if and only if

$$\bar{\partial}_B \nabla^{1,0}_{\mathcal{V}} \psi = 0$$
 and  $\bar{\partial}_{s,\mathcal{V}} \nabla^{1,0}_{s,\mathcal{V}} \psi = 0.$ 

as shown in Lemma 3.5. From these two conditions, and in light of Lemma 2.13, which implies that  $\bar{\partial}_B$  does not depend on *s*, we have

$$\bar{\partial}_s \operatorname{grad}^{\omega_F} \psi = \bar{\partial}_{s,\mathcal{V}} \operatorname{grad}^{\omega_F} \psi + \bar{\partial}_B \operatorname{grad}^{\omega_F} \psi = 0,$$

as claimed.

*Remark* 3.7. In [17, §4.1] it is explained that the kernel of the operator  $\mathcal{R}$  is given by the Lie algebra of the group Aut( $\pi_X$ ) of automorphisms of the projection, described in Definition 2.2. In our case, the kernel of the linearisation  $\widehat{\mathcal{L}}$  is the intersection

$$\ker \mathcal{L} = \operatorname{Lie}(\operatorname{Aut}(\pi_s)) \cap \operatorname{Lie}(\operatorname{Aut}(\pi_X)),$$

where we denote by  $\pi_X : X \to B$  the central fibration and we view  $\{J_s\}$  as a family of complex structures on the same underlying smooth manifolds, compatible with the projection and with  $\omega$ .

We wish to see that  $\widehat{\mathcal{L}}$  is elliptic as a differential operator on the global sections of  $E \to B$ . Let us split  $\mathcal{A}$  in (3.5) as the composition of the two operators

$$A_1: C^{\infty}(E) \to \Gamma(\mathcal{V})$$
$$\varphi \mapsto \operatorname{grad}^{\omega_F} \varphi$$

and

$$A_2: \Gamma(\mathcal{V}) \to T_I \mathscr{J}_{\pi}$$
$$\eta \mapsto -(\mathcal{L}_\eta v).$$

To give a local expression, we make use of *Riemannian* coordinates, and we denote again the vertical coordinates with the letters a, b, c, ... and the horizontal coordinates with the letters i, j, k, ..., as in §2.3. We have:

$$(A_2(\eta))^a_{\ b} = -\left(\mathcal{L}_\eta v\right)^a_{\ b} = -\eta^c \partial_c v^a_{\ b} - v^a_{\ c} \partial_b \eta^c + v^c_{\ b} \partial_c \eta^a$$
$$(A_2(\eta))^a_{\ j} = (A_2(\eta))^a_{\ c} (\omega_F)^{dc} (\omega)_{dj}$$

where the second expression follows from Lemma 2.12. Thus when  $\varphi \in C^{\infty}(E)$ and  $\eta = \omega_F^{de} \partial_e \varphi \partial_d$ ,

$$\begin{aligned} \left(\mathcal{A}(\varphi)\right)^{a}_{\ b} &= -v^{a}_{\ c}\partial_{b}(\omega^{cd}_{F}\partial_{d}\varphi) + v^{c}_{\ b}\partial_{c}(\omega^{ad}_{F}\partial_{d}\varphi) + T_{1}(\varphi) \\ \left(\mathcal{A}(\varphi)\right)^{a}_{\ j} &= -v^{a}_{\ c}\partial_{b}(\omega^{cd}_{F}\partial_{d}\varphi)(\omega_{F})^{eb}(\omega)_{ej} + v^{c}_{\ b}\partial_{c}(\omega^{ad}_{F}\partial_{d}\varphi)(\omega_{F})^{eb}(\omega)_{ej} + T'_{1}(\varphi), \end{aligned}$$

where  $T_1(\varphi)$  and  $T'_1(\varphi)$  are terms involving first order vertical derivatives of  $\varphi$ . Thus we see that  $\mathcal{A}$  is a second order differential operator, and all the derivatives of  $\varphi$  involved are vertical. The adjoint of  $A_1$  is given by

$$A_1^*(\eta) = \operatorname{div}(I\eta).$$

Indeed, we can compute the divergence with respect to any Kähler metric *g* whose Kähler form restricts vertically to  $\omega_F$ , and the result depends only on the vertical part:

$$\langle A_1 \varphi, \eta \rangle_{L^2} = \int_X g_{ac} \, \omega_F^{ab} \, \partial_b \varphi \, \eta^c \, \mathrm{d}Vol_g = \int_X -ig_{ac} g^{ab} \, (\nabla_b \varphi) \, \eta^c \, \mathrm{d}Vol_g$$
  
= 
$$\int_X -i(\nabla_c \varphi) \, \eta^c \, \mathrm{d}Vol_g = \int_X \varphi \, \nabla_c(i\eta^c) \, \mathrm{d}Vol_g = \langle \varphi, A_1^*\eta \rangle_{L^2}.$$

To compute the adjoint of  $A_2$  we make use of the following lemma.

**Lemma 3.8.** Let  $w \in \widetilde{H}^1_W$  and let  $g_F$  be the vertical Riemannian metric induced by  $(\omega_F, I)$ . Then

$$g_F(\mathcal{L}_{\eta}v,w) = g_F(w,\nabla v(\eta)) + g_F(vw - wv,\nabla \eta).$$

The proof of the lemma is obtained by computing the different quantities in Riemannian coordinates [69, §4.2].

In light of the lemma, the adjoint to  $A_2$  can be formally written as

$$A_{2}^{*}(w) = -(\nabla v)^{*}w - \nabla^{*}([v, w]).$$

If  $w = \mathcal{A}(\varphi)$ , the first term is of order 3. So we have:

$$\mathcal{A}^*\mathcal{A}(\varphi) = -\operatorname{div}\left(I\nabla^*_{\mathcal{V}}(v(\mathcal{L}_{\eta_{\varphi}}v)_{\mathcal{V}} - (\mathcal{L}_{\eta_{\varphi}}v)_{\mathcal{V}}v)\right) + \text{lower order terms.}$$

From this expression, we see that all the quantities involved are vertical. This means that, as an operator on the global sections of the vector bundle *E*, the operator

$$\mathcal{A}^*\mathcal{A}: C^\infty(E) \to C^\infty(E)$$

is of order 0. Indeed, let us denote by *r* the rank of *E* and consider a local frame  $h_1, \ldots, h_r$  of *E*. Then we can write a local section  $h = \sum_i f_i h_i$ , with  $f_i \in C^{\infty}(B)$ . Then

$$\mathcal{A}^*\mathcal{A}(h) = \sum_i f_i \mathcal{A}^*\mathcal{A}(h_i).$$

Thus, as an operator on the global sections  $C^{\infty}(E)$ , the operator  $\widehat{\mathcal{L}}$  is elliptic, since  $\mathcal{R}^*\mathcal{R}$  is from [14, §4] and  $\mathcal{R}^*\mathcal{R}$  is of lower order. We have established the following:

**Theorem 3.9.** Let  $\widehat{\mathcal{L}}$  be the linearisation of the optimal symplectic connection equation (2.10). Then  $\widehat{\mathcal{L}}$  is an elliptic operator of order two on the global sections of E which is self-adjoint and whose kernel consists of fibrewise I-holomorphy potentials which are also global  $J_s$ -holomorphy potentials for all s.

# 3.4 Automorphisms of the optimal symplectic connection equation

Let  $(X, H_X) \rightarrow B$  be a relatively cscK fibration. Consider the complex group Aut $(X, H_X)$  of automorphisms of X lifting to  $H_X$ . Its Lie algebra is given by the holomorphic vector fields which vanish somewhere, and we denote it by  $\mathfrak{h}_0$ . Recall from (2.13) the group of relative Hamiltonian isometries

 $K_{\pi} := \text{Isom}(\pi_X, \omega) = \left\{ f \in \text{Diffeo}(X) \mid f^* \omega = \omega \text{ and } \pi_X \circ f = \pi_X \right\}$ 

and from Definition 2.2 the group of relative automorphisms

$$\operatorname{Aut}(\pi_X) = \{ f \in \operatorname{Aut}(X, H_X) \mid \pi_X \circ f = \pi_X \}$$

We denote by  $\mathfrak{h}_{\pi}$  the Lie algebra of  $\operatorname{Aut}(\pi_X)$  and  $\mathfrak{t}_{\pi}$  the Lie algebra of  $K_{\pi}$ . An element in  $\mathfrak{h}_{\pi}$  is a holomorphic vector field which vanishes somewhere and whose flow lies in  $\operatorname{Aut}(\pi_X)$ , while an element of  $\mathfrak{t}_{\pi}$  is a holomorphic vector field which corresponds to a Killing vector field under the identification of the real tangent bundle  $T_{\mathbb{R}}X$  with the holomorphic tangent bundle  $T^{1,0}X$ . The following fibration version of Theorem 1.4 is a result of Dervan and Sektnan [14], [17].

**Lemma 3.10.** 1. Let  $\omega$  be an optimal symplectic connection and  $f \in Aut(\pi_X)$ . Then  $f^*\omega$  is an optimal symplectic connection.

2. Let  $\omega$  be an optimal symplectic connection. Then

$$\mathfrak{h}_{\pi}=\mathfrak{k}_{\pi}\oplus I\mathfrak{k}_{\pi}.$$

In particular, the lemma implies that  $K_{\pi}^{\mathbb{C}}$  is contained in Aut( $\pi$ ) with equality holding if ( $\omega$ , *I*) is an optimal symplectic connection.

We next prove an analogous result for the optimal symplectic connection equation (2.10) on a fibration with K-semistable fibres. Let  $(Y, H_Y) \rightarrow (B, L)$  be such a fibration admitting a degeneration to  $(X, H_X) \rightarrow (B, L)$  and let  $V_{\pi}$  be the Kuranishi space of  $\pi_X$ . Let  $(X, \mathcal{H}) \rightarrow (B, L) \times S$  be the degeneration family. The family of complex structures  $\{J_s\}$  with  $J_0 = I$  corresponds to a family  $\{y_s\}$  of points in  $V_{\pi}$ such that  $y_0$  is the origin of  $V_{\pi}$ . Let v be the tangent vector at the origin of  $V_{\pi}$  that represents the degeneration family, i.e.

$$v = \partial_s|_{s=0} y_s.$$

Consider the stabiliser of v for the action of  $K_{\pi}$ , denoted  $K_{\pi,v}$ . Then

$$K_{\pi,v} = \{ f \in K_{\pi} \mid f^*v = v \},\$$

and

$$G_{\pi,v} = (K_{\pi}^{\mathbb{C}})_v. \tag{3.6}$$

For  $f \in G_{\pi,v}$ 

$$\partial_{s}|_{s=0}y_{s} = v = f^{*}v = f^{*}(\partial_{s}|_{s=0}y_{s}) = \partial_{s}|_{s=0}(f^{*}y_{s})$$

Therefore

$$\partial_s|_{s=0}(y_s - f^* y_s) = 0,$$

so  $v = f^*v$ . So the elements of  $G_{\pi,v}$  are automorphisms of the complex structure I of the relatively cscK degeneration  $X \rightarrow B$  that preserve the projection  $\pi_X$  and are also automorphisms of the complex structures  $J_s$ . Moreover, the pull-back of the optimal symplectic connection operator via  $f \in G_{\pi,v}$  satisfies

$$f^*\left(\frac{1}{2}\nu(v) + p_E(\Theta(\omega, I))\right) = \frac{1}{2}\nu(v) + p_E(\Theta(f^*\omega, I)).$$

Indeed, since  $\nu$  is  $K_{\pi}^{\mathbb{C}}$ -equivariant,

$$f^*\nu(v) = \nu(f^*v) = \nu(v),$$

and by Lemma 3.10,

$$f^*(p_E(\Theta(\omega, I))) = p_E(\Theta(f^*\omega, I)).$$

We have proven the following.

**Lemma 3.11.** Let  $\omega$  be an optimal symplectic connection and  $f \in G_{\pi,v}$ . Then  $f^*\omega$  is an optimal symplectic connection. Moreover, if  $\varphi$  is a fibrewise I-holomorphy potential whose flow of the gradient lies in  $G_{\pi,v}$ , then  $\varphi$  is in the kernel of the linearisation  $\widehat{\mathcal{L}}$ .

Let  $\mathfrak{g}_{\pi,v}$  be the Lie algebra of  $G_{\pi,v}$ , consisting on those holomorphic vector fields whose flow lies in  $K_{\pi}^{\mathbb{C}}$  and which preserve v. In particular, preserving v means that they extend to holomorphic vector fields with respect to all  $J_s$ . Let  $\mathfrak{t}_{\pi,v}$  be the Lie algebra of  $K_{\pi,v}$ , of Killing holomorphic vector fields whose flow preserves v. We can then prove a version of Theorem 1.4 for our setting.

**Theorem 3.12.** Let  $\omega$  be an optimal symplectic connection. Then

$$\mathfrak{g}_{\pi,v} = \mathfrak{k}_{\pi,v} \oplus I\mathfrak{k}_{\pi,v}.$$

In particular  $K_{\pi,v}$  is a reductive subgroup of  $G_{\pi,v}$ .

*Proof.* From Theorem 3.9, the kernel  $\widehat{\mathcal{L}}$  of the linearisation of the optimal symplectic connection equation consists of fibrewise *I*-holomorphy potentials which are also global  $J_s$ -holomorphy potentials for all s. From the discussion above, this is in bijection with the Lie algebra  $\mathfrak{g}_{\pi,v}$ , and  $\mathfrak{t}_{\pi,v}$  corresponds to the real vector fields in  $\mathfrak{g}_{\pi,v}$ . Since  $\widehat{\mathcal{L}}$  is a real operator,  $\widehat{\mathcal{L}}(u + iv) = 0$  if and only if  $\widehat{\mathcal{L}}(u) = 0$  and  $\widehat{\mathcal{L}}(v) = 0$ .

# 3.5 Special Kähler metrics on the total space: the adiabatic limit

Let  $(X, \mathcal{H}) \to B \times S$  be a family of submersions with central fibre the fibration  $(X, H_X) \to (B, L)$  as before. In this section we construct approximate constant scalar curvature and extremal metrics on the total space of  $\pi_Y : (Y, H_Y) \to (B, L)$ , assuming the optimal symplectic connection and the extremal symplectic connection, respectively. We first construct approximate solutions in the case of a discrete automorphism group and in the presence of automorphisms, and then we perturb the approximate solutions by applying the implicit function theorem. We do so by using an *adiabatic limit*, such as in [24, 14].

We will later need to choose  $\omega_B$  appropriately, to produce cscK and extremal metrics on *Y*. To do so, we use the moduli theory of cscK manifolds explained in §1.5. Let  $\mathcal{M}^{cscK}$  be the moduli space of polarised cscK manifolds. Since our central fibration  $\pi : X \to B$  has cscK fibres, it induces a map  $q : B \to \mathcal{M}^{cscK}$ . The pull-back via q of the Weil-Petersson metric, denoted  $\alpha_{WP}$ , is a closed smooth (1, 1)-form on B, and it has the expression (1.24):

$$\alpha_{WP} = \frac{S_b}{m+1} \int_{X/B} \omega^{m+1} - \int_{X/B} \rho \wedge \omega^m, \qquad (3.7)$$

where  $\rho$  is the relative Ricci form defined in Section 2.2.1 and *m* is the dimension of the fibres. Recall that  $\alpha_{WP}$  is positive semi-definite in general.

**Definition 3.13** ([75, 24]). The Kähler metric  $\omega_B \in c_1(L)$  is

- 1. *twisted cscK* with respect to  $\alpha$  if there exists a constant  $c_B$  such that  $Scal(\omega_B) \Lambda_{\omega_B} \alpha = c_B$ ;
- 2. *twisted extremal* with respect to  $\alpha$  if  $\text{Scal}(\omega_B) \Lambda_{\omega_B} \alpha \in \ker \mathcal{D}_B$ , where  $\mathcal{D}_B$  is the Lichnerowicz operator on B.

**Definition 3.14.** The group of automorphisms of the moduli map is

$$\operatorname{Aut}(q) = \left\{ f \in \operatorname{Aut}(B, L) | q \circ f = f \right\}.$$

If we denote by  $h_B$  the twisted extremal holomorphy potential, then the linearisation of the twisted extremal operator at a solution is given by the map [35, §2], [15, §3.2]

$$\mathcal{L}_{\alpha}(\varphi) = -\mathcal{D}_{B}^{*}\mathcal{D}_{B}\varphi + \frac{1}{2}\langle \nabla \Lambda_{\omega_{B}}\alpha, \nabla \varphi \rangle + \langle i\partial\bar{\partial}\varphi, \alpha \rangle.$$
(3.8)

The kernel of this operator is given by the holomorphy potentials of those vector fields whose flow lies in Aut(q) [15, Proposition 3.5]. We prove the following results.

**Theorem 3.15.** Assume that the group  $\operatorname{Aut}(q)$  and the group  $\operatorname{Aut}(Y, H_Y)$  are discrete. Let  $\omega_B$  be twisted cscK with respect to the pull-back via q of the Weil-Petersson metric on the moduli space of cscK manifolds. Let  $\omega$  be an optimal symplectic connection on  $(Y, H_Y) \rightarrow (B, L)$ . Then for all  $k \gg 0$  there exists a constant scalar curvature Kähler metric on Y, in the class  $[\omega] + k[\omega_B]$ .

When automorphisms are present, we use extremal symplectic connections and twisted extremal metrics on the base to prove the existence of extremal metrics on the total space. Recall from Definition 2.11 that  $\omega$  is an extremal symplectic connection on  $\Upsilon$  if

$$\widehat{\mathcal{L}}\left(p_E(\Theta(\omega,I))+\frac{\lambda}{2}\nu\right)=0,$$

so that the function

$$h_1:=p_E(\Theta(\omega,I))+\frac{\lambda}{2}\nu$$

is a holomorphy potential for the complex structure of *Y*.

**Theorem 3.16.** Assume that there is an action of  $Aut(\pi_Y)$  on  $(X, \mathcal{H})$  which is equivariant with respect to the projection onto S and that all automorphisms of the moduli map q lift to  $(Y, H_Y)$ . Let  $\omega_B$  be a twisted extremal metric on B with respect to the pull-back via q of the Weil-Petersson metric on the moduli space of cscK manifolds. Let  $\omega$  be an extremal symplectic connection for on  $(Y, H_Y) \rightarrow (B, L)$ . Then for all  $k \gg 0$  there exists an extremal Kähler metric on Y, in the class  $[\omega] + k[\omega_B]$ .

#### 3.5.1 Approximate solutions in the case of discrete automorphism group

In this section we construct approximate constant scalar curvature Kähler metrics on the total space of  $\pi_s : (X_s, H_s) \rightarrow (B, L)$ , where  $(X_s, H_s)$  is a deformation of a fibration  $\pi_X : (X, H_X) \rightarrow (B, L)$  whose fibres are cscK. We make the assumptions of Theorem 3.15:

- 1. Aut( $X_s$ ,  $H_s$ ) is discrete and ( $X_s$ ,  $H_s$ ) admits an optimal symplectic connection. Thanks to Proposition 3.6, this guarantees that the operator  $\hat{\mathcal{L}}$  is invertible and also that the global Lichnerowicz operator on  $X_s$  with respect to  $\omega_k$  is invertible.
- 2. The base form  $\omega_B \in c_1(L)$  is twisted cscK with respect to the pull-back via q of the Weil-Petersson metric, as in Definition 3.13, and the group Aut(q) is discrete. As recalled in the discussion following Definition 3.14, this implies that the linearisation at a solution of the twisted cscK equation on the base is invertible;

Let  $k \gg 0$  be such that

$$\omega_k = \omega + k\omega_B$$

#### Extremal metrics on the total space

is a Kähler metric on X, and let  $s^2 = \lambda k^{-1}$  for  $\lambda > 0$ . We relate s and k as above, namely  $s^2 = \lambda k^{-1}$ , so we will sometimes denote also the corresponding complex structure by  $J_k$ . Since all the  $J_s$  are isomorphic, Theorem 3.15 still gives the existence of a cscK metric in each adiabatic class for all  $J_s$ . The adiabatic limit technique consists in constructing inductively approximated solutions, which have constant scalar curvature up to a certain order in  $k^{-1/2}$ , then using the implicit function theorem to perturb an approximate solution to a genuine solution. The following result establishes the approximate solution.

**Proposition 3.17.** With the assumptions listed above, for all  $k \gg 0$  and for each r there exist functions

$$f_{B,2}, \ldots, f_{B,r} \in C^{\infty}(B)$$
  $f_{E,2}, \ldots, f_{E,r} \in C^{\infty}(E)$   $f_{R,2}, \ldots, f_{R,r} \in C^{\infty}(R)$ 

and constants

$$\widehat{S}_2,\ldots,\widehat{S}_r$$

such that the Kähler potentials

$$h_{k,r}^B = \sum_{j=2}^r \frac{f_{B,j}}{k^{j-2}}$$
  $h_{k,r}^E = \sum_{j=2}^r \frac{f_{E,j}}{k^{(j-1)/2}}$   $h_{k,r}^R = \sum_{j=2}^r \frac{f_{R,j}}{k^{j/2}}$ 

satisfy

$$\operatorname{Scal}\left(\omega_{k}+i\partial\bar{\partial}\left(h_{k,r}^{B}+h_{k,r}^{E}+h_{k,r}^{R}\right),J_{k}\right)=\widehat{S}_{b}+\sum_{j=2}^{r}\frac{\widehat{S}_{j}}{k^{j/2}}+O\left(k^{(-r-1)/2}\right).$$

*Proof.* With the hypotheses of  $\omega$  being an optimal symplectic connection and  $\omega_B$  being a twisted cscK metric on the base, we have

$$Scal(\omega_k) = \widehat{S}_b + k^{-1} \left( c_B + \psi_{R,1} \right) + O\left( k^{-3/2} \right),$$
(3.9)

where  $\psi_{R,1} \in C^{\infty}(R)$ . In order to make the  $k^{-1}$ -term constant we add a potential  $k^{-1}f \in C^{\infty}(R)$  to  $\omega_k$ . Then

$$\operatorname{Scal}(\omega_k + k^{-1}i\partial\bar{\partial}f) = \widehat{S}_b + k^{-1}\left(c_B + \psi_{R,1} - \mathcal{D}_V^*\mathcal{D}_V f\right) + O\left(k^{-3/2}\right),$$

where the linearisation of the scalar curvature to order 0 in k coincides with (minus) the Lichnerowicz operator with respect to the complex structure I, since the scalar curvature is constant in order 0, and the higher order terms fall into  $O(k^{-3/2})$ . Since  $\mathcal{D}_V^* \mathcal{D}_V$  is a fibrewise elliptic differential operator and  $C^{\infty}(R)$  is orthogonal to its kernel, we can find a solution  $f_{R,1}$  of

$$\psi_{R,1} - \mathcal{D}_V^* \mathcal{D}_V f = constant. \tag{3.10}$$

Summing up, we have proved step r = 2 of Proposition 3.17, with  $f_{B,2} = 0 = f_{E,2}$ . We define

$$\omega_{k,2} = \omega_k + k^{-1} i \partial \partial f_{R,2}$$

such that

$$\operatorname{Scal}(\omega_{k,2}) = \widehat{S}_b + k^{-1}\widehat{S}_2 + O\left(k^{-3/2}\right).$$

To proceed with the approximate solutions, we need the linearisation of the scalar curvature at a metric ( $\omega_{k,r}$ ,  $J_k$ ).

**Lemma 3.18.** The linearisation of the scalar curvature of  $\omega_{k,r}$  satisfies

$$\mathcal{L}_{k,r} = -\mathcal{D}_{\mathcal{V}}^* \mathcal{D}_{\mathcal{V}} + k^{-1} D_1 + k^{-3/2} D_{3/2} + k^{-2} D_2 + O\left(k^{-5/2}\right),$$

where

- 1.  $D^*_{\mathcal{W}} \mathcal{D}_{\mathcal{V}}$  is the vertical Lichnerowicz operator with respect to the complex structure I;
- 2. If  $f \in C^{\infty}(B)$ ,  $D_{j+1/2}(f) = 0$  for all j;
- 3. If  $f \in C^{\infty}(B)$ ,  $D_1(f) = 0$  and

$$\int_{X/B} D_2(f) \omega^m \wedge \omega_B^n = -\mathcal{L}_\alpha(f),$$

where  $\mathcal{L}_{\alpha}$  is the linearisation of the twisted cscK equation on the base, with twisting the Weil-Petersson form  $\alpha_{WP}$ , at a solution, defined in (3.8).

4. If  $f \in C^{\infty}(E)$ , then

$$p_E \circ D_1(f) = -p_E \circ \widehat{\mathcal{L}}(f).$$

*Proof of the Lemma*. Let us distinguish the parameter *s* of the deformation of the complex structure from the parameter *k* of the polarisation. Consider the case n = 0, so that we compute the scalar curvature of the metric ( $\omega_k$ ,  $J_s$ ). Then

$$\mathcal{L}_k = \mathcal{L}_{k,0} + O(s), \tag{3.11}$$

where  $\mathcal{L}_{k,0}$  is the linearisation of the scalar curvature of  $(\omega_k, I)$ . In [14, Proposition 4.11] it is proven that

$$\mathcal{L}_{k,0} = -D_{\mathcal{V}}^* \mathcal{D}_{\mathcal{V}} + k^{-1} D_1' + k^{-2} D_2' + O\left(k^{-3}\right)$$

from which we see that the term of order zero is indeed the vertical *I*-Lichnerowicz operator. This proves claim (1). By imposing the relation  $s^2 = \lambda k^{-1/2}$  we see that the O(s)-term in (3.11) admits an expansion in powers of  $k^{-1/2}$ :

$$k^{-1}D_1'' + k^{-3/2}D_{3/2}'' + k^{-2}D_2'' + O\left(k^{-5/2}\right).$$

Claim (2) follows from the fact that the deformation of the complex structure is vertical, thus all the terms involved in the expansion of the scalar curvature coming from the deformation do not have a  $C^{\infty}(B)$ -component.

Claims (3) and (4) follow as in [14, Proposition 4.11].

The proof of Proposition 3.17 now goes by induction, using Lemma 3.18. We explain in detail steps r = 3 and r = 4. We start from the expansion

$$Scal(\omega_{k,2}) = \widehat{S}_b + k^{-1}\widehat{S}_2 + k^{-3/2}(\psi_{E,3/2} + \psi_{R,3/2}) + O(k^{-2})$$

We add a potential  $k^{-1/2} f_E$  to  $\omega_{k,2}$ . Thus we have

$$\operatorname{Scal}\left(\omega_{k,2} + k^{-1/2}i\partial\bar{\partial}f_E\right) = \widehat{S}_b + k^{-1}\widehat{S}_2 + k^{-2/3}\left(\psi_{E,3/2} + D_1(f) + \psi_{R,3/2}\right) + O\left(k^{-2}\right).$$

Using Lemma 3.18, our hypothesis on the automorphism group of  $(X_s, H_s)$  and the fact that the linearisation  $\widehat{\mathcal{L}}$  of the optimal symplectic connection equation at a solution is elliptic, as proved in Theorem 3.9, we can find a smooth  $f_{E,3}$  such that

$$\psi_{E,3/2} + p_E \circ D_1(f_{E,3}) = constant$$

This makes the  $C^{\infty}(E)$ -term constant to order  $k^{-3/2}$ . We next add a potential  $k^{-3/2} f_R \in C^{\infty}(R)$  and we obtain

$$\begin{aligned} \operatorname{Scal}\left(\omega_{k,2} + i\partial\bar{\partial}\left(k^{-1/2}f_{E,3} + k^{-3/2}f_R\right)\right) &= \widehat{S}_b + k^{-1}\widehat{S}_2 + \\ &+ k^{-3/2}\left(c_{E,3/2} + \psi'_{R,3/2} - \mathcal{D}_V^*\mathcal{D}_V f_R\right) + O\left(k^{-2}\right). \end{aligned}$$

Once again, using the fibrewise ellipticity of  $\mathcal{D}^*_{\mathcal{V}}\mathcal{D}_{\mathcal{V}}$  and the fact that  $C^{\infty}(R)$  is orthogonal to its kernel, we obtain a solution  $f_{R,3}$  of the equation

$$\psi'_{R,3/2} - \mathcal{D}^*_{\mathcal{V}} \mathcal{D}_{\mathcal{V}} f_R = constant.$$

Thus we have constructed a Kähler metric on  $X_s$  constant up to order  $k^{-3/2}$ :

$$\omega_{k,3} = \omega_{k,2} + i\partial\bar{\partial} \left( k^{-1/2} f_{E,3} + k^{-3/2} f_{R,3} \right).$$

As for the step r = 4, we explain how to deal with the  $C^{\infty}(B)$ -term. We add a potential  $f_B$  to  $\omega_{k,3}$ , which amounts to adding a potential  $k^{-1}f_B$  to  $\omega_B$ . Since the scalar curvature of the base affects the order  $k^{-1}$ -term and not the order zero term, the combined effect on the linearisation is of order  $k^{-2}$ . This allows us to write

$$\begin{aligned} \operatorname{Scal}(\omega_{k,3} + i\partial\bar{\partial}f_B) &= \widehat{S}_b + k^{-1}\widehat{S}_2 + k^{-3/2}\widehat{S}_3 + \\ &+ k^{-2}\left(\psi_{B,2} - D_2(f_B) + \psi_{E,2} + \psi_{R,2}\right) + O\left(k^{-5/2}\right). \end{aligned}$$

Thanks to Lemma 3.18 and to our hypothesis on the automorphism group of the moduli map,

$$\psi_{B,2} - p_B \circ D_2(f_B) = constant$$

admits a solution, which we denote  $f_{B,4}$ . This makes the  $C^{\infty}(B)$ -term constant to order  $k^{-2}$ .

The corrections to the  $C^{\infty}(E)$ -term and to the  $C^{\infty}(R)$ -term now work exactly as in the case r = 3.
## 3.5 Special Kähler metrics on the total space

Notice that the order is important: one can make the  $C^{\infty}(E)$ -term constant without affecting the  $C^{\infty}(B)$ -term, but it cannot work the other way around, and similarly for the  $C^{\infty}(R)$ -term.

*Remark* 3.19. The very first step of the approximate solution procedure, which the expansion (3.9), comes from the fact that in Proposition 2.23 we have modified the Kuranishi map  $\Phi$  in order to meet the requirement that  $\text{Scal}_V(\omega, \Phi(x))$  is a section of E, for  $x \in V_{\pi}$ . If we do not deform the Kuranishi map in this way, we can write the vertical scalar curvature as the sum of the projection onto  $C^{\infty}(E)$  and the projection onto  $C^{\infty}(R)$ . The  $C^{\infty}(E)$ -part is the map  $\mu_{\pi}$  defined in 2.24, while the  $C^{\infty}(R)$ -part introduces a term of order  $k^{-1/2}$  in the expansion (3.9), which then becomes

$$Scal(\omega_k) = \widehat{S}_b + k^{-1/2} \psi_{R,0} + k^{-1} (c_B + \psi_{R,1}) + O(k^{-2}).$$

We can get rid of this term by adding a potential  $k^{-1/2}i\partial\bar{\partial}\varphi_{R,0}$  to  $\omega_k$ , as in equation (3.10). Indeed, the linearisation given by Lemma 3.18 of the scalar curvature acquires an extra term  $\sqrt{k}D_{1/2}$ , which is non-zero only on  $C^{\infty}(R)$ , so it does not affect the  $C^{\infty}(E)$  and  $C^{\infty}(B)$  parts in the  $k^{-1}$ -term.

## 3.5.2 Approximate solutions in the presence of automorphisms

In this section, we allow the base and the total space to have automorphisms. As before let  $\hat{\pi} : (X, \mathcal{H}) \to (B, L) \times S$  be a degeneration of the fibration  $\pi_Y : (Y, H_Y) \to B$  to  $\pi_X : (X, H_X) \to B$ . Let  $\omega \in c_1(H)$  be a relatively cscK metric on X; since Y is a small deformation of X,  $c_1(H) = c_1(H_Y)$ , so we can assume that  $\omega$  is relatively Kähler on Y, as explained in §2.2.2.

We make the hypotheses of Theorem 3.16 concerning the groups of automorphisms  $Aut(\pi_Y)$  and Aut(q) defined in 2.2 and 3.14:

1. There is an action of  $\operatorname{Aut}(\pi_Y)$  on  $(\mathcal{X}, \mathcal{H})$  which is equivariant with respect to the projection onto *S*. This means that  $\operatorname{Aut}(\pi_Y)$  acts on each  $X_s$  as a subgroup of automorphisms of  $(X_s, H_s)$ . Since the action extends to the central fibration, this assumption allows us to view  $\operatorname{Aut}(\pi_Y)$  as a subgroup of  $\operatorname{Aut}(\pi)$ . In particular, recall from Remark 3.7 that  $\operatorname{ker} \widehat{\mathcal{L}} = \operatorname{Lie}(\operatorname{Aut}(\pi_Y)) \cap \operatorname{Lie}(\operatorname{Aut}(\pi))$ . With this assumption, we obtain

Ker 
$$\mathcal{L}$$
 = Lie(Aut( $\pi_Y$ )),

and  $h_1$  is a holomorphy potential also on *X*.

2. All automorphisms of the moduli map *q* lift to  $(Y, H_Y)$ .

The first hypothesis is motivated by the analogous definition of test configurations which are equivariant with respect to the automorphisms of the fibres, which are used to test K-polystability of polarised manifolds. Recall from Definition 3.13 that a twisted extremal metric on *B*, with twisting form the Weil-Petersson form  $\alpha_{WP}$  (3.7), satisfies the condition

$$\operatorname{Scal}(\omega_B) - \Lambda_{\omega} \alpha_{WP} = b_1 \in \operatorname{ker} \mathcal{D}_B$$
,

where  $\mathcal{D}_B$  is the Lichnerowicz operator on the base.

*Remark* 3.20. Let  $\widehat{g}$  be a lift of an automorphism of q to  $(Y, H_Y)$ . We claim that  $\widehat{g}$  lies in Aut $(X, H_X)$ . Indeed, denoting by J the complex structure of Y and I the complex structure of X, we have

$$\mathrm{d}\widehat{g}\circ J=J\circ\widehat{g}.$$

But since  $\hat{g}$  is an automorphism in the base direction, it is equivalent to say that

$$\mathrm{d}\widehat{g}\circ J_{\mathcal{H}}=J_{\mathcal{H}}\circ\widehat{g},$$

where  $J_{\mathcal{H}}$  is the horizontal part of *J*. Now,  $J_{\mathcal{H}} = I_{\mathcal{H}}$ , since the deformation of the complex structure which we are considering is only in the vertical direction. Thus  $\hat{g}$  is a lift of an automorphism of *B* to *X* as well.

**Definition 3.21.** We denote the group of automorphisms of  $(Y, H_Y)$  which are also automorphisms of  $(X, H_X)$  as Aut $(Y/X, H_Y)$ .

In light of this definition we have the inclusion  $\operatorname{Aut}(\pi_Y) \subseteq \operatorname{Aut}(Y/X, H_Y)$  and, if  $\widehat{\operatorname{Aut}}(q)$  is a lift of  $\operatorname{Aut}(q)$  to (Y), then  $\widehat{\operatorname{Aut}}(q) \subseteq \operatorname{Aut}(Y/X, H_Y)$ . Thus we can recover the following result from [14, Proposition 3.14].

**Lemma 3.22.** *Suppose that all automorphisms of q lift to* Y*. Then there is a short exact sequence* 

$$0 \rightarrow \text{Lie}(\text{Aut}(\pi_Y)) \rightarrow \text{Lie}(\text{Aut}(Y, H_Y)) \rightarrow \text{Lie}(\text{Aut}(q)) \rightarrow 0.$$

*Remark* 3.23. Let us denote by  $\xi_E$  the holomorphic vector field on Y which arises from the extremal symplectic connection condition:

$$\xi_E = J_s \nabla_{\mathcal{V}} \left( p_E(\Theta(\omega, I)) + \frac{\lambda}{2} \nu \right),$$

and  $\xi_q$  the holomorphic vector field on *B* which arises from the twisted extremal condition:

$$\xi_q = J_B \nabla_B(\operatorname{Scal}(\omega_B) - \Lambda_\omega \alpha_{WP}).$$

By our assumptions,  $\xi_E$  is a holomorphy potential on *X*, and  $\xi_q$  lifts to a holomorphic vector field on *Y* (and on *X*). Nonetheless, the holomorphy potential of  $\xi_q$  on *Y* is a function  $\tilde{b}_1$  such that

$$b_1 = k\pi^* b_1 + O(1).$$

Again from Remark 3.20,  $b_1$  is holomorphic potential for a lift of  $\xi_q$  also on X. As in [14], we need to assume the following invariance properties:  $\omega$  is invariant

under the flow of  $\xi_E$  and of the pull-back of  $\xi_q$ . In order to make this assumptions reasonable to work with, we consider a maximal torus  $T_E$  in Aut $(\pi_Y)$  which contains the flow of  $\xi_E$ , and a maximal torus  $T_q$  in Aut(B, L) which contains the flow of  $\xi_q$ . The pull back  $\widehat{T}_q$  lies in Aut $(Y/X, H_Y)$ . Then we fix a maximal torus T in Aut $(Y, H_Y)$ which contains  $T_E$  and  $T_q$ , and we require that  $\omega$  is invariant with respect to T. From Lemma 3.22, we obtain a splitting Lie $(T) = \text{Lie}(T_E) + \text{Lie}(T_q)$ , so indeed we have  $T \subset \text{Aut}(Y/X, H_Y)$  as well.

Moreover, an analogous splitting holds also for the complexification  $T^{\mathbb{C}}$ , so we can write every vector field  $\xi \in \text{Lie}(T^{\mathbb{C}})$  as  $\xi_E + \xi_q$ . If  $h_E$  is the holomorphy potential of  $\xi_E$  with respect to  $\omega$  and  $h_B$  is the holomorphy potential of  $\xi_q$  on the base with respect to  $\omega_B$ , then  $h_E + k\pi_V^* h_B$  is a holomorphy potential of  $\xi$  on Y (and on X).

Define the extremal symplectic connection operator

$$\mathcal{P}: C^{\infty}(Y,\mathbb{R}) \times C^{\infty}(E) \to C^{\infty}(Y,\mathbb{R})$$

by

$$\mathcal{P}(\varphi, h_1) = p_E \left( \Theta(\omega + i\partial \bar{\partial} \varphi, J_s) \right) + \frac{\lambda}{2} v_{\varphi} - h_1 - \frac{1}{2} \langle \nabla h_1, \nabla \varphi \rangle_{\omega_F}$$

The linearisation at  $(h_1, 0)$  applied to  $(h_1, \psi)$  is obtained, as for the extremal operator described in (1.2), as follows:

$$\widehat{\mathcal{L}}(\psi) - h_1 - \frac{1}{2} \langle \nabla h_1, \nabla \psi \rangle_{\omega_F},$$

where  $\hat{\mathcal{L}}$  is the real operator of the linearisation of the optimal symplectic connection equation described in Lemma 3.4 and the map sending  $\varphi$  to  $\langle \nabla h_1, \nabla \varphi \rangle_{\omega_F}$  is linear. We can write

$$\langle \nabla h_1, \nabla \psi \rangle_{\omega_F} = \frac{1}{2} \nabla h_1(\psi) + \frac{1}{2} i J \nabla h_1(\psi),$$

so if we assume that  $\psi$  is invariant under the torus *T*, the second term vanishes and linearisation is a real operator.

With all of these assumptions in place, we can obtain approximate solutions to the extremal equation much as in §3.5.1.

**Proposition 3.24.** Let  $(X, \mathcal{H}) \to B \times S$  be a degeneration of a smooth fibration  $\pi_Y$ :  $(Y, H_Y) \to B$  to a smooth relatively cscK fibration  $\pi_X : (X, H_X) \to B$ , equivariant with respect to Aut $(\pi_Y)$ . Let  $\omega$  be an extremal symplectic connection on  $X_s$ , invariant under the torus T described in Remark 3.23. Let  $\omega_B$  a twisted extremal metric on the base, and assume that all automorphisms of q lift to Y. Then for each r > 1 there exist functions

$$f_{B,2},\ldots,f_{B,r}\in C^{\infty}(B)^{T}, \qquad f_{E,2},\ldots,f_{E,r}\in C^{\infty}(E)^{T}, \qquad f_{R,2},\ldots,f_{R,r}\in C^{\infty}(R)^{T},$$

base holomorphy potentials

$$b_1,\ldots,b_r\in C^\infty(B)^T$$
,

fibre holomorphy potentials

$$h_1,\ldots,h_r\in C^\infty(E)^T$$

and a constant c such that, letting

$$h_{k,r}^B = \sum_{j=2}^r \frac{f_{B,j}}{k^{j-2}}, \qquad h_{k,r}^E = \sum_{j=2}^r \frac{f_{E,j}}{k^{(j-1)/2}}, \qquad h_{k,r}^R = \sum_{j=2}^r \frac{f_{R,j}}{k^{j/2}}$$

and

$$\eta_{k,r} = c + \sum_{j=1}^{r} \left( b_j k^{(-j-1)/2} + h_j k^{-j/2} \right),$$

the Kähler metric

$$\omega_{k,r} = \omega_k + i\partial\bar{\partial} \left( h_{k,r}^B + h_{k,r}^E + h_{k,r}^R \right)$$

satisfies

$$\operatorname{Scal}\left(\omega_{k,r},J_{k}\right) = \eta_{k,r} + \frac{1}{2} \langle \nabla \eta_{k,r}, \nabla \left(h_{k,r}^{B} + h_{k,r}^{E} + h_{k,r}^{R}\right) \rangle_{\omega_{k}} + O\left(k^{-r-1/2}\right).$$

## 3.5.3 Solution to the non-linear equation

In order to have genuine solutions, we perturb  $\omega_{k,r}$  to a genuine extremal metric by using a quantitative version of the implicit function theorem, as in [24, 7, 15, 14]. In particular, all the cited works rely on Fine's paper [24], though the difference with Fine's setting is that we are considering the base and the total space to have automorphisms, so the linearised operators will have a non-trivial kernel to deal with.

**Theorem 3.25** ([7, Theorem 25]). Let  $F : B_1 \to B_2$  be a differentiable map of Banach spaces such that  $D_0F$  is surjective with right-inverse P. Let

1.  $\delta' > 0$  be such that the non-linear operator  $(F - D_0 F)$  is Lipschitz in  $B_{\delta'}(0)$  with constant  $\frac{1}{2\|P\|}$ , i.e. for  $x_1, x_2 \in B_{\delta'}(0) \subseteq B_1$ , we have

$$\|(F - D_0 F)(x_1) - (F - D_0 F)(x_2)\|_{B_2} \le \frac{1}{2\|P\|} \|x_1 - x_2\|_{B_1};$$

2. 
$$\delta = \frac{\delta'}{2\|P\|}$$
.

Then for all  $y \in B_2$  such that  $||y - F(0)|| < \delta$ , there exists  $x \in B_1$  such that F(x) = y.

To apply the theorem to the extremal operator, one should bound both the right inverse of the linearisation and the non-linear operator. Denote by  $L_{0,p}^2$  the Sobolev spaces of functions on Y computed with respect to  $\omega_{k,r}$ , and remark that these do not depend on k, since the Sobolev norms are equivalent for different values of k [7, Remark 30].

## 3.5 Special Kähler metrics on the total space

Let t be the Lie algebra of T, where T is the torus of automorphisms described in Remark 3.23. Let  $\overline{t}$  be the set of holomorphy potentials whose flow lies in T. We denote by  $(L_{0,p}^2)^T$  the space of T-invariant functions in  $L_{0,p}^2$ .

For each k, r denote by  $\gamma_{k,r}$  the Kähler potential defined in Proposition 3.24, such that the approximately extremal metric  $\omega_{k,r}$  is given by  $\omega_k + i\partial \bar{\partial} \gamma_{k,r}$ . For each k, r we define the map

$$\begin{split} \tau_{k,r}: \mathfrak{t} &\to C^\infty(X,\mathbb{R}) \\ \xi &\mapsto k\pi_Y^*h_B + h_q + \frac{1}{2} \langle \nabla \gamma_{k,r}, \nabla (k\pi_Y^*h_B + h_q) \rangle_{\omega_k}, \end{split}$$

where  $h_B$  and  $h_q$  are the holomorphy potentials defined in Remark 3.23. The map  $\tau_{k,r}$  associates to a *T*-invariant holomorphic vector field the correspondent holomorphy potential with respect to  $\omega_{k,r}$ .

We apply the theorem to the operators

$$\begin{split} F_{k,r} &: (L_{0,p+4}^2)^T \times \bar{\mathfrak{t}} \to (L_{0,p}^2)^T \\ F_{k,r}(\varphi,h) &= \operatorname{Scal}(\omega_{k,r} + i\partial\bar{\partial}\varphi) - \frac{1}{2} \langle \nabla \eta_{k,r}, \nabla \gamma_{k,r} \rangle - \eta_{k,r} - \frac{1}{2} \langle \nabla (\tau_{k,r}(h)), \nabla \varphi \rangle - \tau_{k,r}(h), \end{split}$$

where  $\eta_{k,r}$  is the Kähler potential which makes  $\omega_{k,r}$  approximately extremal. The linearisation of  $F_{k,r}$  is the operator

$$G_{k,r}: (L^2_{0,p+4})^T \times \overline{\mathfrak{t}} \to (L^2_p)^T$$
  
$$(\varphi, h) \mapsto -\mathcal{D}^*_{k,r}\mathcal{D}_{k,r}(\varphi) + \frac{1}{2} \langle \nabla(\operatorname{Scal}(\omega_{k,r}) - \tau_{k,r}(h)), \nabla \varphi \rangle - \tau_{k,r}(h).$$

The proof requires two steps: the first one is to ensure that the linearisation is surjective with bounded inverse  $P_{k,r}$ . Theorem 3.25 then gives  $\delta_k$  such that if  $||F_{k,r}(0)|| < \delta_k$ , a zero of  $F_{k,r}$  exists. Since we want to find a zero for all k, the second step is to find a value of r for which the norm  $||F_{k,r}(0)||$  converges to zero quicker than  $\delta_k$ . The first step is contained in the following lemma [15, Lemma 6.6], based on [24, Lemmas 6.5, 6.6, 6.7].

**Lemma 3.26.** There exists a constant C independent of k such that  $G_{k,r}$  has a right inverse  $P_{k,r}$  such that

$$\left\|P_{k,r}\right\| \le Ck^{5/2}.$$

The second step relies on the following result [15, Lemma 6.6], which is a consequence of the mean value theorem.

**Lemma 3.27.** Let  $N_{k,r} = F_{k,r} - d_0 F_{k,r}$  be the nonlinear part of the extremal operator. Then there are constant c, C such that for all r sufficiently large, if  $f_i \in (L_{p+4}^2)^T \times \overline{t}$  for i = 1, 2 satisfy  $||f_i|| \le c$ , then

$$\left\| \mathcal{N}_{k,r}(f_1) - \mathcal{N}_{k,r}(f_2) \right\|_{L^2_p} \le C \left( \|f_1\|_{L^2_{p+4}(\omega_{k,r})} + \|f_2\|_{L^2_{p+4}(\omega_{k,r})} \right) \left\| f_1 - f_2 \right\|_{L^2_{p+4}(\omega_{k,r})}$$

## Extremal metrics on the total space

By applying the implicit function Theorem 3.25, we can now complete the proof of Theorem 3.16 as follows. Lemma 3.27 implies that  $N_{k,r}$  is Lipschitz on any ball of radius  $\rho$  sufficiently small, with Lipschitz constant  $\rho C$ . Thus the radius  $\delta'$ on which  $N_{k,r}$  is Lipschitz with constant  $(2||P||_{k,r})^{-1}$  is bounded below by some multiple of  $k^{-5/2}$ . Hence  $\delta = \delta'(2||P||)^{-1}$  is bounded below by a multiple of  $k^{-5}$ . In order to apply the implicit function theorem, it remains to bound  $F_{k,r}(0,0)$ . The point-wise bound  $F_{k,r} = O(k^{(-r-1)/2})$  is provided by Proposition 3.24. Results of Fine [24, Lemma 5.6, 5.7] can be applied directly to our situation in order to have a  $L_p^2(\omega_{k,r})$ -bound on  $F_{k,r}(0)$  of order  $k^{5-\frac{1}{2}}$ , when r > 5. Thus the hypotheses of the implicit function theorem are satisfied and  $||F_{k,r}(0)||$  converges to zero quicker than  $\delta_k$ . Therefore  $F_{k,r}$  admits a zero for all k sufficiently large, which gives the required extremal metric on Y.

## Chapter 4

# The moduli space of holomorphic submersions

We construct the moduli space of fibrations admitting optimal symplectic connections, with discrete relative automorphism group. Our main reference in the construction is Fujiki-Schumacher [31], where the moduli space of cscK manifolds in the case of discrete automorphism group is defined. Let  $Y \rightarrow B$  be a fibration that degenerates to a relatively cscK holomorphic submersion  $X \rightarrow B$ , as described in §2.3.1. We first prove that the set of deformations of  $Y \rightarrow B$  that still degenerate to a relatively cscK fibration (possibly different from  $X \rightarrow B$ ) forms a locally closed analytic subset of the relative Kuranishi space. We then prove that the solutions to the optimal symplectic connection equation (2.10) form an open set inside the locally closed subset of admissible deformations. This allows us to define a local moduli space of optimal symplectic connections. Finally, we glue the local moduli space which parametrises optimal symplectic connections.

## 4.1 **Openness of the setting**

Given a fibration  $\pi_Y : (Y, H_Y) \to (B, L)$  with analytically K-semistable fibres, we assume as in §2.2.2 that there exists a degeneration of  $\pi_Y$  to a fibration  $\pi_X : (X, H_X) \to (B, L)$  such that the fibres of  $\pi_X$  are cscK. In particular, we can consider  $Y \to B$  and  $X \to B$  as the same symplectic fibration  $\pi : (M, \omega) \to B$ , and the degeneration as a deformation of a complex structure *J* to *I*, where  $(\omega, I)$  has fibrewise constant scalar curvature. The goal of this section is to understand for which deformations *J'* of *J* we can still find a relatively cscK degeneration *X'*, and to construct such a degeneration.

We begin by working locally in *B*. Let  $\mathcal{U} \subseteq B$  be a coordinate open subset of *B*. The fibre over the origin of  $\mathcal{U}$ , denoted  $M_0$ , has a constant scalar curvature metric  $(\omega_0, I_0)$ . Let *K* be the group of Hamiltonian isometries of  $(\omega_0, I_0)$  and let  $\widetilde{H}_0^1$  be the vector space (1.15) parametrising first-order deformations of  $(M_0, \omega_0, I_0)$ . Since  $(\omega_0, I_0)$  has constant scalar curvature, the complexification of *K* is the group

$$G = \operatorname{Aut}_0(M_0, H_0).$$

We will employ Definition 1.9 of GIT-stability for the action of a reductive group on an affine space, applying the definition of stability to the vector space  $\tilde{H}_0^1$ . The complex structure  $I_0$  corresponds to the origin in the Kuranishi space  $V_0$ , which is fixed by the action of G. Therefore its orbit is closed and its stabiliser is the group G itself, so it is a polystable point. A key result for our construction is the fact that the closure of the orbit of every point in  $\tilde{H}_0^1$  contains a unique polystable orbit (Lemma 1.8).

Let  $V_0$  be the subspace of the Kuranishi space which parametrises integrable almost complex structures; it is a locally closed analytic subspace of  $\widetilde{H}_0^1$  because it is defined by the vanishing of the Nijenhuis tensor. Thus the family  $X \to B$ can be described locally over  $\mathcal{U}$  as a family  $\{x_b\}$  in  $V_0$ . By our hypothesis 2, the automorphism group of  $(M_b, H_b)$  is isomorphic to *G* for all  $b \in B$ . Therefore the points  $\{x_b \mid b \in \mathcal{U}\}$  are all fixed by the action of *G* and are hence polystable.

The family  $Y \rightarrow B$  can be described locally over  $\mathcal{U}$  as a family  $\{y_b \mid b \in \mathcal{U}\}$  of points such that for each b the closure of the G-orbit of  $y_b$  contains the polystable point  $x_b$  [12, Theorem 1.3]. By Lemma 1.8,  $x_b$  is the only polystable orbit in the closure of the orbit of  $y_b$ . Let  $V_0^+$  be the set of all semistable points that have a fixed point in the closure of their orbit. Then the map

$$F: V_0^+ \to V_0 \tag{4.1}$$

that maps a semistable point to the corresponding fixed point is well-defined.

**Lemma 4.1.** The set  $V_0^+$  is an analytic subvariety of  $V_0$  and the map (4.1) is holomorphic.

*Proof.* The space  $V_0$  is an open subset of the vector space  $\tilde{H}^1$ . Let *d* be the dimension of  $\tilde{H}^1$ , so each point  $z \in V_0$  has coordinates

$$(z_1,\ldots,z_d)$$
.

Let us begin with the case when *G* is isomorphic to  $\mathbb{C}^*$ . The fixed points of the action can be described by the vanishing of some coordinates

$$z_{i_1} = \cdots = z_{i_h} = 0$$
 where  $i_1, \ldots, i_h \in \{1, \ldots, d\}$ ,

thus they form an analytic subspace of  $V_0$ . The action of  $\mathbb{C}^*$  can be written as

$$t \cdot (z_1, \ldots, z_d) = (t^{a_1} z_1, \ldots, t^{a_d} z_d),$$

where the numbers  $a_j$  are the *weights* of the action. Then  $V_0$  splits into a sum of weight spaces  $V_0^{\text{pos}} \oplus V_0^{\text{fix}} \oplus V_0^{\text{neg}}$ , where  $\mathbb{C}^*$  acts on  $V_0^{\text{pos}}$  with positive weights, on

## 4.1 Openness of the setting

 $V_0^{\text{neg}}$  with negative weights and fixes the subspace  $V_0^{\text{fix}}$ . By the Hilbert-Mumford criterion 1.11, a semistable point that has a fixed point in the closure of its orbit is described by the following condition: if a coordinate in  $V_0^{\text{pos}}$  is nonzero, then all coordinates in  $V_0^{\text{neg}}$  vanish, and viceversa. This condition yields a set of polynomial equations that define the semistable points in  $V_0^+$  for the action of  $\mathbb{C}^*$ . Thus, the semistable points correspond to an analytic subset of  $V_0$ . The map *F* is the projection onto the set described by  $\{z_{i_1} = \ldots = z_{i_h} = 0\}$ , thus it is holomorphic.

Let now *G* be any reductive group, and let *y* be a semistable point and *x* be a polystable and fixed point in the closure of its orbit. The fixed points of *G* form a vector subspace  $V_0^{\text{fix}}$  also in this case. Moreover there exists a 1-parameter subgroup  $\lambda_x : \mathbb{C}^* \hookrightarrow G$  such that

$$\lim_{t\to 0}\lambda_x(t)\cdot y=x.$$

It follows that the map that sends y to the fixed point x in the closure of its orbit is the projection onto the vector subspace of fixed points for  $\lambda_x$ , hence it is holomorphic. Consider the composite map

$$V_0 \to V_0 \xrightarrow{\rho_1} V_0^{\text{fix}}$$
 (4.2)

where the first map is the projection onto the vector subspace of fixed points for  $\lambda_x$  and the second map is the projection onto the subspace of fixed points for the whole group *G*. The map is holomorphic because it is a composition of holomorphic projections. We prove that it coincides with the map (4.1). Let  $\tilde{x}' = \lim_{t\to 0} \lambda_x(t) \cdot y'$ . Then

$$\overline{G\cdot \widetilde{x'}} \subseteq \overline{G\cdot y'}.$$

The unique polystable orbit contained in  $\overline{G \cdot \tilde{x}'}$  is also a polystable orbit in  $\overline{G \cdot y'}$ , so it must coincide with the fixed-point orbit  $\{x'\}$ . Thus flowing along the orbit of  $\tilde{x}'$  amounts to projecting onto the subspace of fixed points, and so the map (4.2) maps any point y' to the fixed point x' in the closure of its orbit.

*Remark* 4.2. The map (4.1) is analogous to the one given by the Byałinicki-Birula decomposition [4, 46].

Let  $V_{\pi}$  be the Kuranishi space of the fibration  $X \rightarrow B$  defined in Theorem 2.21. Consider the subspace

$$V_{\pi}^{+} := \{ y \in V_{\pi} \mid y|_{X_{b}} \in V_{h}^{+} \}.$$

We remark that  $V_{\pi}^+$  depends on the complex structure of the reference fibration  $X \to B$  and its deformation  $Y \to B$ . We denote by  $\mathscr{J}_{\pi}^+$  the image of  $V_{\pi}^+$  via the relative Kuranishi map (2.18).

**Lemma 4.3.**  $V_{\pi}^+$  is a locally closed subvariety of  $V_{\pi}$ .

#### The moduli space of holomorphic submersions

*Proof.* Let  $\mathcal{U}_1$  and  $\mathcal{U}_2$  be two open coordinate subsets of B with non-empty intersection and let  $0_1$  be the origin of  $\mathcal{U}_1$  and  $0_2$  be the origin of  $\mathcal{U}_2$ . Denote by  $I_1$  and  $I_2$  the complex structures of the fibres  $X_{0_1}$  and  $X_{0_2}$ , and  $G_1$ ,  $G_2$  their groups of automorphisms. By assumption,  $G_1$  and  $G_2$  are isomorphic, and we will denote them by G. Recall that the Kuranishi space is *versal* and more specifically that it is a complete deformation space for the nearby fibres. We use the versality of the Kuranishi map to glue the spaces  $V_1^+$  and  $V_2^+$  constructed in Lemma 4.1 to a variety  $V^+$  and prove that  $V_{\pi}^+$  is obtained as the intersection of said variety with the relative Kuranishi space  $V_{\pi}$ . More precisely, versality of the Kuranishi space means that there is a map

$$\tau_{21}:V_2\to V_1,$$

not necessarily unique.

The map  $\tau_{21}$  can be taken to be *G*-equivariant. In fact, the *G*-equivariance can be traced back to the proof of Kuranishi's Theorem 1.23. The map  $\tau_{21}$  is defined using the implicit function theorem, which can be applied to a *K*-equivariant map to provide an implicit inverse function which is *K*-equivariant. Since *G* is the complexification of *K*, we obtain that  $\tau_{21}$  is *G*-equivariant. The equivariance also implies that the image of  $V_2^+$  is  $V_1^+$ , so we can restrict  $\tau_{21}$  to

$$\widetilde{\tau}_{21}: V_2^+ \to V_1^+$$

Moreover the map  $\tilde{\tau}_{21}$  has an inverse that is constructed reversing the roles of  $V_1$ and  $V_2$ , so it is an isomorphism. In fact, although the map  $\tau_{21}$  is not canonical, the restriction to  $\tilde{\tau}_{21}$  is fixed by the reference K-semistable fibration  $Y \rightarrow B$ . Each Kuranishi space  $V_b$  is a complex subspace of the vector space  $\tilde{H}_b^1$ , described as the kernel of the elliptic operator  $P_b P_b^* + (\bar{\partial}_b^* \bar{\partial}_b)^2$  (1.15). So we can use the isomorphism  $\tilde{\tau}_{21}$  to glue the spaces  $V_b^+$  to a subvariety  $V^+$  of the kernel of the fibrewise elliptic operator

$$P_{\mathcal{V}}P_{\mathcal{V}}^* + (\bar{\partial}_{\mathcal{V}}^*\bar{\partial}_{\mathcal{V}})^2$$

Therefore the intersection

$$V_{\pi}^{+} = V^{+} \cap V_{\pi}$$

is a locally closed subvariety of  $V_{\pi}$ .

The following lemma shows that we can glue the local fibration constructed in Lemma 4.1 to a global fibration over *B*.

**Lemma 4.4.** Let  $Y' = (M, \omega, J') \rightarrow B$  be a fibration with complex structure J' represented by  $y' \in V_{\pi}^+$ . Then Y' degenerates to  $X' = (M, \omega, I') \rightarrow B$  such that

- 1.  $(\omega, I')$  is relatively cscK;
- 2. the groups  $Aut(X'_h, H'_h)$  are isomorphic for all  $b \in B$ .

## 4.1 Openness of the setting

*Proof.* Let  $\mathcal{U} \subseteq B$  be an open coordinate subset. By the relative Kuranishi Theorem 1.23, for each  $b \in \mathcal{U}$  there exists a point  $y'_b \in V_0$  such that  $\Phi_0(y'_b)$  is in the same  $\mathcal{G}^c$ -orbit of  $J'_b$ . The fact that the map (4.1) is holomorphic implies that we can find polystable points  $\{x'_b\}$  such that  $\{\Phi_0(x'_b)\}$  are a holomorphic family of cscK complex structures over  $\mathcal{U}$  that are deformations of  $\{J'_b\}$ . Then we can construct a local relatively cscK fibration from the pullback diagram

$$\begin{array}{cccc} X'_{\mathcal{U}} & \longrightarrow & \mathcal{M}_{\mathcal{U}} \\ \downarrow & & \downarrow^{p_{\mathcal{U}}} \\ \mathcal{U} & & \longrightarrow & V_0 \end{array} \tag{4.3}$$

where  $p_{\mathcal{U}} : \mathcal{M}_{\mathcal{U}} \to V_0$  is Kuranishi's versal family and  $i(b) = x'_h$ .

Now we glue the local fibrations to a fibration  $X' \to B$ . Let  $\mathcal{U}_1$  and  $\mathcal{U}_2$  and G be as in Lemma 4.3. The associated Kuranishi maps are denoted respectively by  $\Phi_1 : V_1 \to \mathcal{J}$  and  $\Phi_2 : V_2 \to \mathcal{J}$ .

For  $b \in \mathcal{U}_1 \cap \mathcal{U}_2$ , consider the complex structure  $J'_b$ . Since  $J'_b$  can be regarded as a deformation of both  $I_1$  and  $I_2$ , there exist points  $y'_{b,1} \in V_1$  and  $y'_{b,2} \in V_2$  such that

$$\Phi_1(y'_{b,1}) = J'_b = \Phi_2(y'_{b,2}).$$

The diagram (4.3) produces two fibrations  $X'_1 \rightarrow \mathcal{U}_1$  and  $X'_2 \rightarrow \mathcal{U}_2$ . In order to glue the local fibrations we need to prove that there is an isomorphism

$$\phi_{12}: X_1'|_{\mathcal{U}_1 \cap \mathcal{U}_2} \to X_2'|_{\mathcal{U}_1 \cap \mathcal{U}_2}$$

and that it satisfies the cocycle condition

$$\phi_{23} \circ \phi_{12} = \phi_{13}$$

on a triple intersection  $\mathcal{U}_1 \cap \mathcal{U}_2 \cap \mathcal{U}_3$  of open subsets of *B*. In particular, the first condition produces a global compact complex manifold *X'*, while the cocycle condition implies that *X'* admits a submersion onto *B*.

To prove the existence of the isomorphism  $\phi_{12}$  we use again the fact that the Kuranishi space induces complete deformations on the nearby fibres. The map

$$\tau_{21}: V_2 \to V_1$$

is such that  $\mathcal{M}_2 = \tau_{21}^* \mathcal{M}_1$ . In particular, we have the following diagram



#### The moduli space of holomorphic submersions

If we show that the diagram is commutative, i.e.  $\tau_{21} \circ i_2 = i_1$ , then

$$X_2'|_{\mathcal{U}_1\cap\mathcal{U}_2}=i_2^*\mathcal{M}_2\simeq(\tau_{21}\circ i_2)^*\mathcal{M}_1=i_1^*\mathcal{M}_1=X_1'|_{\mathcal{U}_1\cap\mathcal{U}_2}.$$

The commutativity follows from the fact that the map  $\tau_{21}$  is *G*-equivariant. Indeed the points  $i_1(b)$  and  $i_2(b)$  are defined as the fixed-point limits of semistable orbits. Moreover, an equivariant map between two spaces on which there is an action of the same group *G* sends fixed points to fixed points and the closure of the orbit of  $i_1(b)$  to the closure of the orbit of  $i_2(b)$ .

We now show the cocycle condition. Let  $\mathcal{U}$  be a triple intersection  $\mathcal{U}_1 \cap \mathcal{U}_2 \cap \mathcal{U}_3$ and consider the following diagram



On  $V_3$  we have two families pulled-back from  $V_1$ , namely  $\tau_{31}^* \mathcal{M}_1$  and  $(\tau_{21} \circ \tau_{32})^* \mathcal{M}_1$ . They induce two distinct families on  $\mathcal{U}$ , pulled-back using  $i_3$ . Although in general it is *not* true that  $\tau_{31}$  is equal to the composition  $\tau_{21} \circ \tau_{32}$ , the commutativity of the arrows proved above implies that

$$\tau_{21} \circ \tau_{32} \circ i_3 = \tau_{31} \circ i_3.$$

Therefore the two families  $\tau_{31}^* \mathcal{M}_1$  and  $(\tau_{21} \circ \tau_{32})^* \mathcal{M}_1$  coincide.

## 4.2 Openness of the space of optimal symplectic connections

Let  $Y \to B$  be a fibration with K-semistable fibres, and assume that it degenerates to a fibration  $X \to B$  with cscK fibres, in the sense of §2.2.2. Let  $Y' \to B$  be a deformation of  $Y \to B$  in  $\mathscr{J}_{\pi}^+$ . Then  $Y' \to B$  admits a degeneration to  $X' \to B$ , whose fibres are cscK, as explained in §4.1. The goal of this section is to show that if Y admits an optimal symplectic connection then Y' also does.

We denote by *I* the complex structure of *X*, by *J* the complex structure of *Y* and we assume that  $Y \rightarrow B$  is generated by  $v_0 \in V_{\pi}$ . We also assume that  $(\omega, J)$  is an optimal symplectic connection. Let  $V_{\pi}^+$  be the subvariety of  $V_{\pi}$  which describes the family of complex structures  $\mathscr{J}_{\pi}^+$ . The following is a relative version of Proposition 1.22.

**Proposition 4.5.** For every  $\varphi \in \mathcal{K}_E(I)$  there exists  $f \in \text{Diff}_0(M)$  such that  $f^*\omega_{\varphi} = \omega$ and  $(M, \omega_{\varphi}, I) \to B$  is isomorphic to  $(M, \omega, f^*I) \to B$ . *Proof.* Let us consider a potential  $\varphi \in \mathcal{K}_E(I)$  and a path  $\{\varphi_t\}$  in  $\mathcal{K}_E(I)$  from 0 to  $\varphi$ . A result of Hallam [34, Theorem 3.3] guarantees that this path exists and that it is smooth. We define the relatively cscK metrics

$$\omega_t = \omega + 2i\partial \partial \varphi_t$$

and the Kähler metrics

$$\omega_{k,t} = \omega_t + k\omega_B,$$

where the  $\partial$ ,  $\overline{\partial}$  operators are taken with respect to the relatively cscK complex structure *I*. From Proposition 2.5, we have that  $\dot{\varphi}_t \in C^{\infty}(E(\omega_t, I)) \oplus C^{\infty}(B)$ , for all *t*. Thus the fibrewise Hamiltonian vector fields

$$\eta_t := \operatorname{grad}^{\omega_t} \dot{\varphi}_t$$

are well-defined. Consider the vertical vector fields

$$\xi_t := (\nabla^{g_t} \dot{\varphi}_t)_{\mathcal{V}} = (I \operatorname{grad}^{\omega_t} \dot{\varphi}_t)_{\mathcal{V}}.$$

Then fibrewise

$$\frac{\mathrm{d}}{\mathrm{d}t}\omega_t = -\mathcal{L}_{I\eta_t}\omega_t = \mathcal{L}_{\xi_t}\omega_t. \tag{4.4}$$

Let  $\{f_t, t \in [0, 1]\}$  be the *isotopy* of the time-dependent vector field  $\xi_t$ , i.e. the collection of diffeomorphisms of M such that

$$\frac{\mathrm{d}}{\mathrm{d}t}f_t = \xi_t(f_t), \quad f_0 = \mathrm{id}$$

Since  $\xi_t$  is vertical,  $f_t \in \text{Diffeo}(M, \pi)$ . As in the proof of Proposition 1.22, we apply the following property (1.13):

$$\frac{\mathrm{d}}{\mathrm{d}t}f_t^*\eta_t = f_t^*\left(\mathcal{L}_{\xi_t}\eta_t + \frac{\mathrm{d}\eta_t}{\mathrm{d}t}\right),$$

where  $\xi_t$  and  $f_t$  are a time-dependent vector field and its isotopy respectively. Applying it to  $\omega_t$  gives the fibrewise relation

$$\frac{\mathrm{d}}{\mathrm{d}t}f_t^*\omega_t=0,$$

which implies

$$f_t^*\omega_t = f_0^*\omega = \omega.$$

Define  $J_t = f_t^*I$ . Then, for t = 1, the two metrics  $g(\omega, f_1^*I)$  and  $g(\omega + 2i\partial \bar{\partial} \varphi, I)$  are fibrewise isometric, i.e.

$$(M, \omega, f_1^*I) \simeq (M, \omega + 2i\partial \partial \varphi, I)$$

as relative Kähler manifold with fibrewise constant scalar curvature.

Thus, given an integrable relatively cscK complex structure *I* in  $\mathcal{J}_{\pi}$ , we have a map

$$F_{\pi}: \mathcal{K}_{E}(I) \longrightarrow \mathcal{J}_{\pi}$$

$$\varphi \longmapsto f_{1}^{*}I =: F_{\pi}(\varphi, I)$$

$$(4.5)$$

which locally parametrises all integrable complex structures in the same diffeomorphism class of *I* that are fibrewise cscK with respect to the fixed  $\omega$ . Its differential at the origin is

$$d_0 F_{\pi}(\dot{\varphi}) = \left. \frac{\mathrm{d}}{\mathrm{d}t} \right|_{t=0} J_t = \left. \frac{\mathrm{d}}{\mathrm{d}t} \right|_{t=0} f_t^* I = \left. \frac{\mathrm{d}}{\mathrm{d}t} \right|_{t=0} \mathcal{L}_{\xi_t} I = \mathcal{L}_{(\mathrm{Igrad}^{\omega}\dot{\varphi})_{\mathcal{V}}} I = -\frac{1}{2} \bar{\partial} (\mathrm{grad}^{\omega} \dot{\varphi})_{\mathcal{V}},$$

where, again by Proposition 2.5,  $\dot{\phi} \in C^{\infty}(E(\omega, I)) \oplus C^{\infty}(B)$ .

Now let  $v_0 \in V_{\pi}$  be the deformation of the complex structure which generates the family  $Y \rightarrow B$  and let  $\Phi$  be the relative Kuranishi map (2.18). Then we can define the map

$$F'_{\pi} : \mathcal{K}_{E}(I) \to \tilde{H}^{1}_{\mathcal{V}}$$
$$\varphi \mapsto f_{1}^{*}v_{0} =: F'_{\pi}(\varphi, I, v_{0}).$$

Its differential at the origin computed at  $\dot{\phi} \in C^{\infty}(E(\omega, I)) \oplus C^{\infty}(B)$  is

$$d_0 F'_{\pi}(\dot{\varphi}) = \left. \frac{d}{dt} \right|_{t=0} f_t^* v_0 = \mathcal{L}_{(I \text{grad}^{\omega} \dot{\varphi})_V} v_0.$$
(4.6)

**Definition 4.6.** We denote with  $\mathcal{P}$  the set of triples  $(\varphi, x, v) \in C^{\infty}(M) \times TV_{\pi}$  such that  $\varphi \in \mathcal{K}_{E}(\Phi(x))$  and x is  $K_{\pi}^{\mathbb{C}}$ -polystable. The optimal symplectic connection operator is the map

$$\mathcal{G}: \mathcal{P} \to C^{\infty}(X)$$
$$(\varphi, x, v) \mapsto p_{E(\varphi, x)} \left( \Theta(\omega, F_{\pi}(\varphi, \Phi(x))) \right) + \frac{\lambda}{2} \nu_{\varphi, x} \left( F'_{\pi}(\varphi, \Phi(x), v) \right)$$

In this expression  $E(\varphi, x)$  is the vector bundle of fibre holomorphy potentials with respect to the Kähler structure ( $\omega$ ,  $F_{\pi}(\varphi, \Phi(x))$ ) and  $v_{\varphi,x}$  is the map (2.8) computed with respect to the complex structure  $F_{\pi}(\varphi, \Phi(x))$ .

We now compute the differential of  $\mathcal{G}$  along the  $\varphi$ -variable computed at  $(0, 0, v_0)$ . To do so, we need the following technical result on the contraction with  $\omega_k$ .

**Lemma 4.7.** Let  $\alpha$  be a covariant 2-tensor of type (1, 1). Then

$$\Lambda_{\omega_k}\alpha = \Lambda_{\mathcal{V}}\alpha + \frac{1}{k}\Lambda_{\omega_B}\alpha + O\left(k^{-2}\right),$$

where  $\Lambda_{\omega_k}$  denotes the contraction with the Kähler metric  $\omega_k$ ,  $\Lambda_V$  denotes the contraction in the vertical direction and  $\Lambda_{\omega_B}$  denotes the contraction with respect to the base metric.

*Proof.* At each point  $x \in X_b \subset X$ , the matrix  $[\omega_k]$  is block-diagonal:

$$\begin{pmatrix} [\omega_F] & 0 \\ 0 & [\omega_{k,\mathcal{H}}] \end{pmatrix}$$
.

In local coordinates around the point x

$$(\omega_k)^{p\bar{q}}\alpha_{p\bar{q}} = (\omega_F)^{ab}\alpha_{a\bar{b}} + (\omega_{k,\mathcal{H}})^{ij}\alpha_{i\bar{j}}.$$

The horizontal part of  $\omega_k$ , denoted  $\omega_{k,\mathcal{H}}$  splits as  $\omega_{\mathcal{H}} + k\omega_B$ , where  $\omega_{\mathcal{H}}$  is the horizontal part of  $\omega$ . Let  $[\omega_{k,\mathcal{H}}]$ ,  $[\omega_{\mathcal{H}}]$  and  $[\omega_B]$  be the matrices of coefficients of the two-forms  $\omega_{k,\mathcal{H}}$ ,  $\omega_{\mathcal{H}}$  and  $\omega_B$  respectively. We can write

$$[\omega_{k,\mathcal{H}}] = k \left( \frac{[\omega_{\mathcal{H}}][\omega_B]^{-1}}{k} + \mathbb{1} \right) [\omega_B],$$

where  $\mathbb{1}$  is the identity matrix and the base form  $\omega_B$  induces a Riemannian metric on the horizontal tangent bundle, so its inverse is well defined. The inverse of the matrix  $[\omega_{k,\mathcal{H}}]$  can be expanded in inverse powers of k as

$$\begin{split} [\omega_{k,\mathcal{H}}]^{-1} &= k^{-1} [\omega_B]^{-1} \left( \frac{[\omega_{\mathcal{H}}][\omega_B]^{-1}}{k} + \mathbb{1} \right)^{-1} \\ &= k^{-1} [\omega_B]^{-1} \sum_{i=0}^{\infty} \left( -\frac{[\omega_{\mathcal{H}}][\omega_B]^{-1}}{k} \right)^i \\ &= \frac{1}{k} [\omega_B]^{-1} + O(k^{-2}) \,. \end{split}$$

This implies the claim.

We then write

$$\mathcal{G}(\varphi, x, v) = \mathcal{G}_1(\varphi, x) + \mathcal{G}_2(\varphi, x, v),$$

where

$$\mathcal{G}_1(\varphi,x) = p_{E(\varphi,x)}\left(\Theta(\omega,F_\pi(\varphi,\Phi(x)))\right)$$

and

$$\mathcal{G}_2(\varphi, x, v) = \frac{\lambda}{2} v_{\varphi, x} \left( F'_{\pi}(\varphi, \Phi(x), v) \right),$$

and we split the computation into two separate lemmas.

**Lemma 4.8.** Let  $\varphi \in C^{\infty}(E(\omega, I)) \oplus C^{\infty}(B)$ . The differential along the first variable of  $\mathcal{G}_1$  is

$$D_1\mathcal{G}_1|_{(0,0,v_0)}(\varphi) = -\frac{1}{2}\mathcal{R}^*\mathcal{R}(\varphi).$$

## The moduli space of holomorphic submersions

*Proof.* Let { $J_t$ } be a family of relatively cscK complex structures compatible with  $\omega$ , such that  $J_0 = I$  and consider the scalar curvature of ( $\omega_k$ ,  $J_t$ ),

$$\operatorname{Scal}(\omega_k, J_t) = \operatorname{Scal}(\omega_k, I) + t \left. \frac{\mathrm{d}}{\mathrm{d}t} \right|_{t=0} \operatorname{Scal}(\omega_k, J_t) + O(t^2).$$
(4.7)

Let  $\alpha = \partial_{t=0} J_t$ , and define

$$Q_k(\alpha) := \left. \frac{\mathrm{d}}{\mathrm{d}s} \right|_{s=0} \operatorname{Scal}(\omega_k, J_t).$$

From Lemma 1.18 we obtain

$$Q_k(\alpha) = \operatorname{Im}\left((g_k)^{p\bar{q}} \nabla_p \nabla_a \alpha^a_{\bar{q}}\right) = \operatorname{Re}\left((\omega_k)^{p\bar{q}} \nabla_p \nabla_a \alpha^a_{\bar{q}}\right).$$

We need to compute the sub-leading order term of  $Q_k$  along the differential of the map  $F_{\pi}$  (4.5)

Let us consider  $\alpha = \overline{\partial} (\operatorname{grad}^{\omega} \varphi_E)^{1,0}_{V}$ , where  $\varphi_E \in C_E^{\infty}(X)$  is a fibrewise holomorphy potential. Then

$$Q_k(\alpha) = k^{-1} \operatorname{Re} \left( \mathcal{R}^* \mathcal{R}(\varphi_E) \right) + O\left(k^{-2}\right), \qquad (4.8)$$

where  $\mathcal{R}(\varphi_E) = \overline{\partial}_B \operatorname{grad}_V \varphi_E$  and the adjoint is computed with respect to  $\omega_F + \omega_B$ . As explained in §2.2.1, the operator  $\mathcal{R}^*\mathcal{R}$  can actually be seen as  $p_E \circ \mathcal{L}_1$  restricted to  $C^{\infty}(E(\omega, I))$ . Its kernel consists of fibrewise holomorphy potentials which are global holomorphy potentials on *X* with respect to  $\omega_k$ .

A local coordinate expression for  $\alpha$  is

$$\alpha = \bar{\partial} \left( Y_{\varphi_E} \right)_{\mathcal{V}}^{1,0} = \partial_{\bar{z}^j} \left( \omega_F^{a\bar{b}} \partial_{\bar{w}^b} \varphi_E \right) \partial_{w^a} \otimes \mathrm{d}\bar{z}^j = \nabla_{\bar{j}} \left( \omega_F^{a\bar{b}} \nabla_{\bar{b}} \varphi_E \right) \partial_{w^a} \otimes \mathrm{d}\bar{z}^j,$$

where we have used that the component of  $\alpha$  with the covariant index of vertical type vanishes, since the potential  $\varphi_E$  is a fibrewise holomorphy potential. Therefore, using Lemma 4.7,

$$\begin{aligned} Q_k(\alpha) &= k^{-1} \operatorname{Re} \left( \omega_B^{i\bar{j}} \,\nabla_i \nabla_a \nabla_{\bar{j}} \left( \omega_F^{a\bar{b}} \nabla_{\bar{b}} \varphi_E \right) \right) + O\left(k^{-2}\right) = \\ &= k^{-1} \operatorname{Re} \left( \omega_B^{i\bar{j}} \,\omega_F^{a\bar{b}} \,\nabla_i \nabla_a \nabla_{\bar{j}} \nabla_{\bar{b}} \varphi_E \right) + O\left(k^{-2}\right) = \\ &= k^{-1} \operatorname{Re} \left( \mathcal{R}^* \mathcal{R}(\varphi_E) \right) + O\left(k^{-2}\right). \end{aligned}$$

A similar computation also holds if we consider, instead of a potential in  $C^{\infty}(E(\omega, I))$ , a map  $\varphi \in C^{\infty}(E(\omega, I)) \oplus C^{\infty}(B)$ , so that  $\alpha = \overline{\partial}(Y_{\varphi})_{\mathcal{V}}^{1,0}$ . This choice amounts to considering an element in the image of the differential of the map  $F_{\pi}$  (4.5). In this case

$$Q_k(\alpha) = \operatorname{Re}(\mathcal{L}_0\varphi) + \frac{1}{k}\operatorname{Re}(\mathcal{L}_1\varphi) + O(k^{-2}),$$

where  $\mathcal{L}_0(\varphi) = \mathcal{L}_0(\varphi_B) = 0$ , since  $\varphi_B$  is constant when restricted to a fibre, so we obtain the same equation as (4.8). To obtain the correct coefficient in the claimed expression note that

$$\mathbf{d}_0 F_{\pi}(\varphi) = -\frac{1}{2}\alpha.$$

**Lemma 4.9.** Let  $\varphi \in C^{\infty}(E(\omega, I)) \oplus C^{\infty}(B)$ . The differential along the first variable of  $\mathcal{G}_2$  is

$$D_1\mathcal{G}_2|_{(0,0,v_0)}(\varphi) = -\mathcal{A}^*\mathcal{A}(\varphi).$$

*Proof.* We compute

$$\left.\frac{\mathrm{d}}{\mathrm{d}t}\right|_{t=0} v_t \left(f_t^* v_0\right)$$

where  $v_t$  is the map v computed with respect to the complex structure  $f_t^* \Phi(0)$ . Now,  $f_t$  is the isotopy of the vector field  $\xi_t = (I \operatorname{grad}^{\omega_t} \dot{\varphi}_t)_V$ , where  $\dot{\varphi}_t$  is in  $C^{\infty}(E_t) \oplus C^{\infty}(B)$ and  $\dot{\varphi}_0 = \varphi$ . In the expression of  $\xi_t$  we are fixing the complex structure I and varying the Kähler form  $\omega_t$ . In particular,  $\xi_t$  is a fibrewise holomorphic vector field with respect to I. This implies that  $f_t^*I = I$ , so  $v_t = v$ . Therefore

$$\frac{\mathrm{d}}{\mathrm{d}t}\Big|_{t=0} \nu_t \left(f_t^* v_0\right) = \mathrm{d}_{v_0} \nu \left(\frac{\mathrm{d}}{\mathrm{d}t}\Big|_{t=0} f_t^* v_0\right).$$

Using the expression (4.6) we obtain

$$\mathbf{d}_{v_0} \nu \left( \frac{\mathbf{d}}{\mathbf{d}t} \Big|_{t=0} f_t^* v_0 \right) = \mathbf{d}_{v_0} \nu \left( \mathcal{L}_{(I \operatorname{grad}^{\omega} \varphi)_{\mathcal{V}}} v_0 \right).$$

The right-hand side can be written as  $-\mathcal{A}^*\mathcal{A}(\varphi)$  following the description (3.4). The minus sign follows from the relation

$$(I {\rm grad}^{\omega} \varphi)_{\mathcal{V}} = - \nabla_{\mathcal{V}} \varphi,$$

where  $\nabla_{\mathcal{V}}$  is the vertical Riemannian gradient.

We define the operator

$$\mathcal{G}: \mathcal{P}^{2,\ell} \longrightarrow W^{2,\ell-2}(X)$$

$$(\varphi, x, v) \longmapsto p_{E(x,\varphi)} \left( \Theta(\omega, F_{\pi}(\varphi, \Phi(x))) \right) + \frac{\lambda}{2} \nu_{\varphi,x} \left( F'_{\pi}(\varphi, \Phi(x), v) \right),$$

$$(4.9)$$

where  $\mathcal{P}^{2,\ell}$  is the space defined in Definition 4.6, but the functions are considered to be in the Sobolev space  $W^{2,\ell}(X)$  instead of smooth.

**Proposition 4.10.** Let  $\pi_X : X \to B$  be a holomorphic submersion with a fibrewise cscK structure  $(\omega, I)$  and let  $\pi_Y : Y \to B$  be a deformation of  $X \to B$  with complex structure J. Let  $V_{\pi}$  be the Kuranishi space based at I and let  $v_0 \in V_{\pi}$  represent the complex structure J. Assume that

- 1.  $(\omega, J)$  is an optimal symplectic connection;
- 2. the relative automorphism group

$$\operatorname{Aut}(\pi_Y) := \left\{ f \in \operatorname{Aut}(Y, H_Y) \mid f \circ \pi = \pi \right\}$$

is discrete.

3. the group  $Aut(X_b, H_b)$  is independent of B, and will be denoted G.

Then for any small deformation v of  $v_0$  in  $V_{\pi}^+$  there exists a pair  $(x, v) \in TV_{\pi}$  such that  $(\omega, \Phi(x))$  is relatively cscK and v generates a complex structure J', and a Kähler potential  $\varphi$  such that

$$\omega + i\partial \partial \varphi$$

*is an optimal symplectic connection with respect to J', where the*  $\partial$ *,*  $\overline{\partial}$  *operators are with respect to*  $\Phi(x)$ *.* 

*Proof.* The proof consists of proving that the operator 4.9 is an elliptic operator with a trivial kernel so that we can apply the implicit function theorem. We note that this can be done even though  $V_{\pi}^+$  may be a singular complex space. Indeed, let  $X \to B \times V_{\pi}^+$  be a family of holomorphic submersions such that the fibre over  $0 \in V_{\pi}^+$  is  $X \to B$ . By the Kuranishi theorem [49, §1] we can locally consider a smooth trivialisation of the family over  $V_{\pi}^+$  such that the complex structures of the fibrations  $X_x \to B \times \{x\}$  form a smooth family  $\{J_s\}$ .

As before, let *I* denote the complex structure of *X*. As we assume that  $(\omega, I, v_0)$  is an optimal symplectic connection,  $\mathcal{G}(0, 0, v_0) = 0$ . The derivative with respect to the first component, given by Lemma 4.8 and 4.9, is

$$d_1 \mathcal{G}_{(0,0,v_0)}(\varphi) = -\mathcal{R}^* \mathcal{R}(\varphi) - \lambda \mathcal{R}^* \mathcal{R}(\varphi) = -\mathcal{L}(\varphi).$$

The hypothesis on the automorphism group implies that the kernel of the linearisation is empty, so the implicit function theorem guarantees that there exists a map

$$\{ (x,v) \in TV_{\pi} \mid x \text{ is } K_{\pi}^{\mathbb{C}} \text{-polystable} \} \to W^{2,\ell}(X)$$
$$(x,v) \mapsto \varphi(x,v)$$

such that locally around  $(0, 0, v_0)$ 

$$\mathcal{G}(\varphi(x,v),x,v) = 0. \tag{4.10}$$

The function  $\varphi(x, v)$  is smooth by the standard theory of regularity of solutions to elliptic partial differential equations, applied to the operator  $\mathcal{G}$  [3, Theorem 41]. Therefore solutions to (4.10) produce optimal symplectic connections.

## 4.3 The moduli space of optimal symplectic connections

Let  $(Y, H_Y) \rightarrow B$  be a relatively K-semistable holomorphic submersion with a degeneration to a relatively cscK fibration  $(X, H_X) \rightarrow B$ . Assume that we have a relatively Kähler metric  $(\omega, J)$  on Y that is an optimal symplectic connection. Let W be the subset of the Kuranishi space  $V_{\pi}$  that corresponds to fibrations satisfying the hypotheses of Proposition 4.10. Then W is an open subset of the locally closed subvariety  $V_{\pi}^+$  described in §4.1 by Proposition 4.10.

**Lemma 4.11** ([27, Corollary to Proposition 2]). *The group*  $Aut(\pi_Y)$  *is a subgroup of*  $Aut(Y, H_Y)$  *with finitely many connected components.* 

In particular, under our assumption Aut( $\pi_Y$ ) is a finite discrete group. Let  $V_{\pi_Y}$  be the Kuranishi space of the fibration  $\pi_Y$  and let

$$\tau: V_{\pi\gamma} \to V_{\pi}$$

be the map given by completeness of the Kuranishi space. If we denote  $\tau^{-1}W =: W_Y$ , then  $W_Y$  is a locally closed subvariety of  $V_{\pi_Y}$ .

Let  $\mathcal{Y} \to B \times W_Y$  be the Kuranishi family of fibrations which admit an optimal symplectic connection, with central fibration  $Y \to B$ . The quotient

$$W_Y/\operatorname{Aut}(\pi_Y) \tag{4.11}$$

is a local complex space and it is Hausdorff since it is the quotient of a variety by a finite group. We now explain that we can glue the quotients (4.11) to obtain a global Hausdorff moduli space  $\mathcal{M}$  of fibrations that admit an optimal symplectic connection.

*Remark* 4.12. The moduli space  $\mathcal{M}$  depends on the group  $G = \operatorname{Aut}_0(X_b, H_b)$ . In other words,  $\mathcal{M}$  parametrises all fibrations  $\pi_Y : Y \to B$  such that they have discrete relative automorphism group and such that they degenerate to a relatively cscK fibration whose fibres have G as their automorphism group.

The following result builds on [31, Proposition 6.5] and [30, Lemma 3.8].

**Lemma 4.13.** Let  $\mathcal{Y}$  and  $\mathcal{Y}'$  over  $W_Y$  be two families of fibrations that admit an optimal symplectic connection. The group of isomorphisms between  $\mathcal{Y}$  and  $\mathcal{Y}'$  that preserve the fibration structure, denoted  $\operatorname{Isom}_{W_Y}(\mathcal{Y}, \mathcal{Y}', B)$ , is proper over  $W_Y$ .

*Proof.* Let  $y_t \to \bar{y}$  be a convergent sequence in  $W_Y$  and consider a family of isomorphisms  $f_t : \mathcal{Y}_{y_t} \to \mathcal{Y}'_{y_t}$  preserving the projection onto B. Such isomorphisms are fibrewise isometries with respect to the underlying fibrewise Riemannian metrics. Therefore there exists a subsequence  $\{f_{t_k}\}$  which converges in the  $C^m$ -topology to a fibrewise  $C^m$ -isometry  $f : \mathcal{Y}_{\bar{y}} \to \mathcal{Y}'_{\bar{y}}$  [30, Lemma 3.8]. Moreover, f is a biholomorphism because it is the limit of biholomorphic maps. Therefore the convergence takes place in  $Isom_{W_Y}(\mathcal{Y}, \mathcal{Y}', B)$ , which is then proper over  $W_Y$ .

The moduli space of holomorphic submersions

The global definition of  $\mathcal{M}$  relies on the following lemma, whose proof follows from [1, XI.6].

**Lemma 4.14.** Let  $\mathcal{Y} \to B \times W_Y$  be a family of fibrations that admit an optimal symplectic connection. For any points  $y, w \in W_Y$ , the fibrations  $Y_y \to B$  and  $Y_w \to B$  are isomorphic if and only if there exists an element  $g \in \operatorname{Aut}(\pi_Y)$  such that g(w) = y.

*Proof.* We can prove the lemma for the entire relative Kuranishi space  $V_{\pi_Y}$ . The statement follows from:

- 1. For any  $y \in V_{\pi_Y}$ , the automorphism group Aut $(\pi_{Y_y})$  is contained in Aut $(\pi_Y)$ ;
- 2. For any  $y \in V_{\pi_Y}$  there exists an Aut $(\pi_{Y_y})$ -invariant open neighbourhood  $\mathcal{U}_y$  such that any isomorphism between fibres of  $\mathcal{Y}|_{\mathcal{U}_y} \to \mathcal{U}_y$  is induced by an element of Aut $(\pi_{Y_y})$ .

We begin by proving the second statement. Assume by contradiction that there exist two sequences  $\{y_n\}$  and  $\{w_n\}$  both converging to y and that there exist  $\{g_n\}$  in  $\operatorname{Aut}(\pi_{Y_y}) \setminus \operatorname{Aut}(\pi_Y)$  such that  $g_n \cdot w_n = y_n$ . Then by Lemma 4.13 there exists  $g \in \operatorname{Aut}(\pi_{Y_y})$  such that  $g_n \to g$ . Up to replacing  $g_n$  with  $g_n g^{-1}$  and  $w_n$  with  $g^{-1}w_n$  we may assume that g is the identity of  $\operatorname{Aut}(\pi_{Y_y})$ . Consider the map

$$\operatorname{Aut}(\pi_Y) \times V_{\pi_Y} \to \widetilde{\operatorname{H}}^1_{\mathcal{V}}$$
$$(\eta, y) \mapsto \eta \cdot y.$$

By Theorem 2.21, this is a local biholomorphism at (id, y). So we have  $F(id, y_n) = F(g_n, w_n)$ . Hence  $g_n = id$ , a contradiction.

The first claim then follows exactly as in [1, p.204]. We report the proof for completeness. Set

$$I = \{(y, g) \in W_Y \times \operatorname{Aut}(\pi_{y_y})\}.$$

Then *I* is equal to  $\text{Isom}(\mathcal{Y}, \mathcal{Y}, B)$ . Up to shrinking  $W_Y$  we can assume that any connected component *I'* of *I* intersects  $\{0\} \times \text{Aut}(\pi_Y)$ . Let  $\widehat{I} = \{(y, g) \in I' \mid g \in \text{Aut}(\pi_Y)\}$ . Then  $\widehat{I}$  is non-empty and Zariski-closed. It follows from 2 that  $\widehat{I}$  contains an open set. Then  $\widehat{I} = I'$ , which concludes the proof.

To glue the charts (4.11), we use the completeness of the Kuranishi space. Let  $Y_1 \rightarrow B$  be another relatively K-semistable fibration which admits an optimal symplectic connection and is close to  $Y \rightarrow B$ . Then the Kuranishi theorem gives a map

$$\tau: V_{\pi_{Y_1}} \to V_{\pi_Y}$$

such that a family  $\mathcal{Y}_1 \to B \times V_{\pi_{Y_1}}$  is isomorphic to the pull-back via  $\tau$  of  $\mathcal{Y} \to B \times V_{\pi_Y}$ . Consider the composition of this map with the inclusion  $i : W_{Y_1} \to V_{\pi_{Y_1}}$ :

$$\tau \circ i: W_{Y_1} \to V_{\pi_Y}.$$

Let  $y_1 \in W_{Y_1}$ . Then  $y_1$  represents a fibration with an optimal symplectic connection, i.e. the image of  $y_1$  via the Kuranishi map is a complex structure  $J_{y_1}$  such that  $(\omega, J_{y_1})$ is an optimal symplectic connection. Therefore, there is a representative y of  $J_{y_1}$  in  $V_{\pi_Y}$  which belongs to  $W_Y$ , so

$$\alpha := \tau \circ i : W_{Y_1} \to W_Y$$

Lemma 4.14 allows us to pass to the quotient and obtain an isomorphism

$$\widetilde{\alpha}: W_{Y_1}/\operatorname{Aut}(\pi_{Y_1}) \to W_Y/\operatorname{Aut}(\pi_Y), \tag{4.12}$$

which is uniquely determined (while  $\alpha$  itself might not be). Indeed, the inverse is constructed by reverting the roles of *Y* and *Y*<sub>1</sub>. Therefore, we can use it to glue the local charts to give  $\mathcal{M}$  the structure of a complex space.

**Proposition 4.15.** *The space* M *is a Hausdorff complex space with at most countably many connected components.* 

*Proof.* The countability follows from [30, Theorem 7.3]. We prove the Hausdorff property. Let  $\mathcal{Y} \to B \times W_Y$  and  $\mathcal{Y}_1 \to B \times W_{Y_1}$  be two families of fibrations which admit an optimal symplectic connection. Let  $y_t \to \overline{y}$  be a sequence in  $W_Y$  and  $y_{1t} \to \overline{y}_1$  be a sequence in  $W_{Y_1}$  and assume that  $Y_{y_t}$  is isomorphic to  $Y_{1,y_{1t}}$  as fibrations over *B*. Following the proof of [27, Proposition 10], we show that  $Y_{\overline{y}} \to B$  is isomorphic to  $Y_{1,\overline{y}_1}$ . Let

$$W = W_Y \times W_{Y_1},$$

and let  $\widehat{\mathcal{Y}} \to B \times \widehat{W}$  and  $\widehat{\mathcal{Y}}_1 \to B \times \widehat{W}$  be the pull-back of  $\mathcal{Y}$  and  $\mathcal{Y}_1$  using the projections of  $\widehat{W}$  onto the first and second component respectively. Consider

$$\Sigma = \{(y, y_1) \in W \mid Y_y \to B \text{ is isomorphic to } Y_{1,y_1} \to B\}.$$

It follows from the properness of Isom $(\widehat{\mathcal{Y}}, \widehat{\mathcal{Y}}_1, B)$  that  $\Sigma$  is a locally closed analytic subvariety of  $\widehat{W}$ . Therefore  $(\overline{y}, \overline{y}_1) \in \Sigma$ , which concludes the proof.

We have proven the following.

**Corollary 4.16.** There exists a Hausdorff complex space  $\mathcal{M}$  which parametrises holomorphic submersions over a fixed base admitting an optimal symplectic connection, with fixed relative automorphism group.

## 4.4 A Weil-Petersson type Kähler metric

In this section, we define a Kähler metric on the moduli space of fibrations admitting an optimal symplectic connection. We do so by describing a relative version of the theory of Weil-Petersson type metrics developed by Fujiki and Schumacher for cscK manifolds [31, Sections 8, 9]. In particular, we first define the Weil-Petersson

## The moduli space of holomorphic submersions

metric locally on the sets  $W_Y$  and subsequently we extend the definition to the charts  $W_Y/\operatorname{Aut}(\pi_Y)$  and then to the moduli space  $\mathcal{M}$ .

Consider a family of holomorphic submersions that admit an optimal symplectic connection, denoted by  $\mathcal{Y} \to B \times W_Y$ , and let  $Y \to B$  be the central fibration. Let  $\omega_t$  be also the optimal symplectic connection on  $Y_t \to B$ . For each *k* sufficiently large,  $\omega_{t,k} = \omega_t + k\omega_B$  is a Kähler form on  $Y_t$ , and we denote by  $\omega_{t,F}$  its purely vertical part. Let  $\mathcal{Y} \to W_Y$  be the composition with the second projection.

From Theorem 2.21 together with Proposition 2.23, for each  $t \in W_Y$  there is an injective map

$$d_t \Phi : T_t W_Y \hookrightarrow \widetilde{H}^1_{\mathcal{V}}(Y_t) \subseteq \Omega^{0,1}(\mathcal{V}^{1,0}_{Y_t})$$
(4.13)

that identifies a vector  $\alpha$  in the Zariski tangent space  $T_t W_Y$  with a (0, 1)-form valued in the (1, 0)-tangent bundle of  $Y_t$ , which we will also denote by  $\alpha$ . The map (4.13) is equivariant with respect to the action of Aut( $\pi_Y$ ). Therefore we can define an inner product on  $T_t W_Y$  by pulling back the  $L^2$ -product on  $\Omega^{0,1}(V_{Y_t}^{1,0})$  induced by the Hermitian metric associated to  $\omega_{t,F} + \omega_B$ . For any  $\alpha, \beta \in T_t W_Y$ , its imaginary part is given by

$$\Omega_t(\alpha,\beta) := \langle \alpha,\beta \rangle_{\omega_{t,F}+\omega_B} = \int_{Y_t} \Lambda_{\omega_{t,F}+\omega_B} \operatorname{Tr}_{\omega_{t,F}}(\alpha\overline{\beta}) \omega_t^m \wedge \omega_B^n,$$
(4.14)

where we denote by  $\Lambda$  the contraction of the covariant part and by Tr the trace of the contravariant part. We give the following definition.

**Definition 4.17.** The *relative Weil-Petersson metric* on  $W_Y$ , denoted by  $\Omega_{WP}$ , is the two-form  $\{\Omega_t\}_{t \in W_Y}$ .

Using the compatibility of the deformations with the Kähler form, we write the trace in coordinates as

$$\Lambda_{\omega_{t,F}+\omega_B} \operatorname{Tr}_{\omega_{t,F}}(\alpha\overline{\beta}) = \alpha^a_{\ \overline{q}} \overline{\beta^b_{\ \overline{p}}}(\omega_{t,F})_{a\overline{b}} (\omega_{t,F}+\omega_B)^{\overline{q}p} = \alpha^a_{\ \overline{b}} \overline{\beta^b_{\ \overline{a}}} + \Lambda_{\omega_B} \operatorname{Tr}_{\omega_{t,F}}(\alpha\overline{\beta}).$$

Therefore the integral (4.14) can be written as the sum

$$\int_{Y_t} \alpha^a_{\ \overline{b}} \overline{\beta^b}_{\ \overline{a}} \omega^m_t \wedge \omega^n_B + \int_{Y_t} \Lambda_{\omega_B} \operatorname{Tr}_{\omega_{t,F}}(\alpha \overline{\beta}) \omega^m_t \wedge \omega^n_B.$$

*Remark* 4.18. The first term can be split over  $Y_t$  as

$$\int_{B} \left( \int_{Y_{t,b}} \alpha^{a}_{\ \bar{b}} \overline{\beta^{b}}_{\ \bar{a}} \omega^{m}_{t} \right) \omega^{n}_{B}.$$
(4.15)

In particular, it vanishes when  $\alpha$  and  $\beta$  restrict to the trivial deformation on the fibres. This is the case when  $\mathcal{Y} \to B \times W_Y$  is a family of holomorphic submersions with rigid fibres, for example.

We next prove that (4.14) is closed and positive definite. Consider for each  $t \in W_Y$  a *k*-dependent inner product on  $\widetilde{H}^1_V(Y_t)$ , defined using the Kähler form  $\omega_{t,k}$  of  $Y_t$ :

$$\Omega_{k,t}(\alpha,\beta) := \int_{Y_t} \langle \alpha,\beta \rangle_{\omega_{t,k}} \omega_{t,k}^{n+m} = \int_{Y_t} \Lambda_{\omega_{t,k}} \operatorname{Tr}_{\omega_{t,k}}(\alpha\overline{\beta}) \omega_{t,k}^{n+m}.$$

The collection  $\{\Omega_{t,k}\} =: \Omega_k$  is the Weil-Petersson type Hermitian metric defined in Definition 1.30 for any family of smooth polarised varieties. Using Lemma 4.7, we can write in local holomorphic coordinates

$$\langle \alpha, \beta \rangle_{\omega_{t,k}} = \alpha^a_{\bar{q}} \overline{\beta^b}_{\bar{p}} (\omega_{t,k})_{a\bar{b}} (\omega_{t,k})^{\bar{q}p} = \alpha^a_{\bar{b}} \overline{\beta^b}_{\bar{a}} + k^{-1} \Lambda_{\omega_B} \operatorname{Tr}_{\omega_{t,F}} (\alpha \overline{\beta}) + O(k^{-2}).$$

The expansion in powers of *k* of  $\omega_{t,k}$  reads

$$\omega_{t,k}^{n+m} = k^n \omega_t^m \wedge \omega_B^n + \frac{k^{n-1}}{nm} \omega_t^{m+1} \wedge \omega_B^{n-1} + O\left(k^{n-2}\right).$$

Then

$$\Omega_{t,k}(\alpha,\beta) = k^n \int_{Y_t} \alpha^a_{\ \bar{b}} \overline{\beta^b}_{\ \bar{a}} \omega^m_t \wedge \omega^n_B + k^{n-1} \left[ \int_{Y_t} \alpha^a_{\ \bar{b}} \overline{\beta^b}_{\ \bar{a}} \omega^{m+1}_t \wedge \omega^{n-1}_B + \int_{Y_t} \Lambda_{\omega_B} \operatorname{Tr}_{\omega_{t,F}}(\alpha\overline{\beta}) \omega^m_t \wedge \omega^n_B \right] + O\left(k^{n-2}\right).$$

$$(4.16)$$

Then the two terms in the sum (4.14) are the first and third coefficients of this expansion.

We next describe a fibre integral formula for the Weil-Petersson metric on  $W_Y$ . Let  $\omega_{\mathcal{Y},k}$  be the relatively Kähler metric on  $W_Y$  such that its restriction to each  $Y_t$  is the Kähler metric  $\omega_t + k\omega_B$ , where  $\omega_t$  is an optimal symplectic connection on  $Y_t \rightarrow B$ . Let also  $\rho_{\mathcal{Y},k}$  be the curvature of the Hermitian structure induced by  $\omega_{\mathcal{Y},k}$  on the relative anticanonical bundle

$$-K_{\mathcal{Y}/W_{Y}}=\bigwedge^{n+m}\mathcal{V}_{\mathcal{Y}/W_{Y}},$$

where  $\mathcal{V}_{\mathcal{Y}/W_Y}$  denotes the vertical tangent bundle of the fibration  $\mathcal{Y} \to W_Y$ . From the fibre integral formula of Theorem (1.31), the *k*-dependent (1, 1)-form  $\Omega_k$  can be written as a fibre integral over the map  $\mathcal{Y} \to W_Y$ :

$$\Omega_k(\omega_{\mathcal{Y},k}) = -\int_{\mathcal{Y}/W_Y} \rho_{\mathcal{Y},k} \wedge \omega_{\mathcal{Y},k}^{n+m} + \frac{1}{n+m+1} \int_{\mathcal{Y}/W_Y} \operatorname{Scal}_{\mathcal{V}}(\omega_{\mathcal{Y},k}) \omega_{\mathcal{Y},k}^{n+m+1}.$$
(4.17)

By expanding  $\Omega_k(\omega_{\mathcal{Y},k})$  in powers of k, we can find a fibre integral formula for the Weil-Petersson metric (4.14). Since the base B is fixed, the relative metric  $\omega_{\mathcal{Y},k}$  can be written as

$$\omega y_{,k} = \widehat{\omega} + k \omega_B,$$

were the restriction of  $\widehat{\omega}$  to each  $Y_t$  is the optimal symplectic connection  $\omega_t$ . Then  $\rho_{\mathcal{Y},k}$  can be expanded in powers of k as

$$\rho_{\mathcal{Y},k} = i\partial\partial \log \left(k^n \widehat{\omega}^m \wedge \omega_B^n + O\left(k^{n-1}\right)\right)$$
  
=  $i\partial\bar{\partial} \log \left(\widehat{\omega}^m \wedge \omega_B^n\right) + O\left(k^{-1}\right)$   
=  $i\partial\bar{\partial} \log \det(\widehat{\omega}) + i\partial\bar{\partial} \log \det(\omega_B) + O\left(k^{-1}\right)$ 

where the second line follows from the fact that  $\widehat{\omega}^m \wedge \omega_B^n$  is a volume form. The  $k^{-1}$ -term is exact, because the two volume forms  $\omega_{\mathcal{Y},k}^{m+n}$  and  $\widehat{\omega}^m \wedge \omega_B^n$  both induce the class  $c_1(-K_{\mathcal{Y}/W_Y})$ . Moreover,  $\det(\widehat{\omega})$  is the relative determinant of  $\widehat{\omega}$  and  $i\partial\overline{\partial} \log \det(\widehat{\omega})$  is the curvature of the Hermitian metric induced by  $\widehat{\omega}$  on the relative anticanonical bundle  $-K_{\mathcal{Y}/B\times W_Y}$ . To compute the expansion of the vertical scalar curvature, Proposition 3.1 gives the expansion

$$\operatorname{Scal}(\omega_k) = \widehat{S}_b + k^{-1} \left( \operatorname{Scal}(\omega_B) - \Lambda_{\omega_B} \alpha_{\mathrm{WP}} + p_E(\Theta(\omega)) \right) + O\left( k^{-3/2} \right),$$

where  $\widehat{S}_b$  is the average scalar curvature of the fibres and  $\alpha_{WP}$  is the Weil-Petersson metric on *B* induced by the relatively cscK degeneration  $X \rightarrow B$ . Then, since  $\widehat{\omega}$  is an optimal symplectic connection when restricted to each  $Y_t$ , the vertical scalar curvature of  $\omega_{Y,k}$  admits an expansion as

$$\operatorname{Scal}_{\mathcal{V}}(\omega_{\mathcal{Y},k}) = \widehat{S}_b + k^{-1} \left( \operatorname{Scal}(\omega_B) - \Lambda_{\omega_B} \widehat{\alpha} \right) + O\left(k^{-2}\right),$$

where  $\hat{\alpha}$  is a closed two-form on *B*. Then the leading order term of (4.17) is

$$I_0 = -\int_{\mathcal{Y}/W_Y} i\partial\bar{\partial}\log\det(\widehat{\omega}) \wedge \widehat{\omega}^m \wedge \omega_B^n + \frac{1}{n+m+1}\int_{\mathcal{Y}/W_Y} \widehat{S}_b\widehat{\omega}^{m+1} \wedge \omega_B^n$$

The sub-leading order term is the sum of the four integrals

$$I_{1} = -\int_{\mathcal{Y}/W_{Y}} i\partial\bar{\partial}\log\det(\widehat{\omega}) \wedge \widehat{\omega}^{m+1} \wedge \omega_{B}^{n-1}, \qquad (4.18)$$

$$I_{2} = \frac{1}{n+m+1} \int_{\mathcal{Y}/W_{Y}} \widehat{S}_{b}\widehat{\omega}^{m+2} \wedge \omega_{B}^{n}, \qquad (4.18)$$

$$I_{3} = -\frac{1}{n} \int_{\mathcal{Y}/W_{Y}} \operatorname{Scal}(\omega_{B})\widehat{\omega}^{m+1} \wedge \omega_{B}^{n}, \qquad I_{4} = \frac{1}{n+m+1} \int_{\mathcal{Y}/W_{Y}} (\operatorname{Scal}(\omega_{B}) - \Lambda_{\omega_{B}}\widehat{\alpha}) \widehat{\omega}^{m+1} \wedge \omega_{B}^{n}.$$

We can use this expansion to prove a fibre integral formula in our setting.

**Lemma 4.19.** The Weil-Petersson metric  $\Omega_{WP}(\omega_y)$  (4.14) can be written as the fibre integral

$$I_0 + I_2 + I_3 + I_4. \tag{4.19}$$

*Proof.* It follows from [31, Lemma 8.5] applied to the family  $\mathcal{Y} \to B \times W_Y$  that the two-form defined as the collection of the integrals

$$\int_{Y_t} \alpha^a_{\ \bar{b}} \overline{\beta^b_{\ \bar{a}}} \omega^{m+1}_t \wedge \omega^{n-1}_B$$

is equal to  $I_1$ . Indeed, working locally in  $D \subset B \times W_Y$ , Fujiki and Schumacher prove that one can trivialise the family  $\mathcal{Y}$  over D as  $Y_{0,0} \times D$  such that the horizontal distribution induced by  $\widehat{\omega}$  is preserved. Then, given a family  $\{\beta_t\}$  of vertical deformations of the complex structure of  $Y_{0,0}$  which represent the family  $Y_{0,0} \times D \rightarrow$ D, [31, Lemma 8.5] gives the equality

$$i\partial\bar{\partial}\log\det(\widehat{\omega}) = \operatorname{Tr}(\partial_t\beta_t|_{t=0}\partial_t\beta_t|_{t=0}),$$

where  $\partial_t \beta_t$  is the map (4.13).

**Lemma 4.20.** The two-form  $\Omega_{WP}$  is closed and positive-definite on  $W_Y$ .

*Proof.* Since the map (4.13) is injective, the integral (4.14) is positive. To prove closedness, we show that the terms  $I_0$ ,  $I_2$ ,  $I_3$  and  $I_4$  in Lemma 4.19 are closed. The terms  $I_0$  and  $I_2$  are closed because they are the fibre integrals of a closed form. The term  $I_4$  can be written as

$$\int_{\mathcal{Y}/W_Y} (\rho_B - \widehat{\alpha}) \wedge \widehat{\omega}^{m+1} \wedge \omega_B^{n-1},$$

where  $\rho_B$  is the Ricci form of  $\omega_B$ . In particular  $(\rho_B - \hat{\alpha}) \wedge \omega_B^{n-1}$  is a top degree form on *B*, hence it is closed. So  $I_4$  is closed. Analogously,  $I_3$  is closed.

Let  $h_{WP}(W_Y)$  be the Hermitian metric on the tangent bundle to  $W_Y$  induced by the two-form  $\Omega_{WP}$ .

**Theorem 4.21.** *The Hermitian metric*  $h_{WP}(W_Y)$  *induces a global Kähler metric on the moduli space* M*.* 

*Proof.* The theorem follows from the fact that the action of the finite group  $\operatorname{Aut}(\pi_Y)$  is induced by an automorphism of the Kuranishi family, so the Weil-Petersson metric  $h_{WP}(W_Y)$  is invariant for the action of  $\operatorname{Aut}(\pi_Y)$ . Therefore it defines a metric on the quotient  $W_Y/\operatorname{Aut}(\pi_Y)$ .

*Remark* 4.22. We have defined a Weil-Petersson metric on  $W_Y$  that is independent of the adiabatic parameter k. This is reasonable from the point of view of describing the moduli space of fibrations using the optimal symplectic connection alone, which is only relatively Kähler. However, it is possible that a different kind of a Weil-Petersson type metric could be defined by taking a sequence of Kähler metrics that depend on k, where the adiabatic construction plays a bigger role.

## 4.4.1 The determinant line bundle for the Weil-Petersson metric

We construct a line bundle on  $W_Y$ , and hence on the moduli space, such that the Weil-Petersson metric represents its first Chern class. To do so, we appeal to the theory of Deligne pairings [22], [23], [5, §1]. Let  $M \rightarrow B$  be a flat, projective morphism between complex algebraic varieties of relative dimension d and consider d + 1 line bundles  $L_0, ..., L_d$  on M. The push-forward of the intersection product of  $L_0, ..., L_d$  is an isomorphism class of line bundles on B, represented by the cohomology class

$$\int_{M/B} c_1(L_0) \wedge \dots \wedge c_1(L_d).$$
(4.20)

The *Deligne pairing* of  $L_0, \ldots, L_d$ , denoted by  $\langle L_0, \ldots, L_d \rangle_{M/B}$ , is a canonical choice of a line bundle on *B* such that (4.20) is its first Chern class. The construction is symmetric, multilinear and functorial. Moreover, if  $h_0, \ldots, h_d$  are Hermitian metrics on  $L_0, \ldots, L_d$  respectively, the theory provides a metric  $\langle h_0, \ldots, h_d \rangle_{M/B}$ on  $\langle L_0, \ldots, L_d \rangle_{M/B}$ . Denoting by  $\omega_0, \ldots, \omega_d$  the curvature forms of  $h_0, \ldots, h_d$ respectively, the curvature of  $\langle h_0, \ldots, h_d \rangle_{M/B}$  is given by the fibre integral

$$\int_{M/B} \omega_0 \wedge \dots \wedge \omega_d. \tag{4.21}$$

The fibre integral formula for the Weil-Petersson metric on  $W_Y$  of Lemma 4.19 is a special case of the expression (4.21). To describe it as the curvature form of a line bundle on  $W_Y$ , we first recall the following result of Fujiki and Schumacher for the *k*-dependent Weil-Petersson metric.

**Proposition 4.23** ([31, §9]). The k-dependent Weil-Petersson type Kähler metric  $\Omega_k(\omega_{\mathcal{Y},k})$  represents the first Chern class of the line bundle

$$-\langle -K_{\mathcal{Y}/W_{Y}}, \mathcal{L}_{k}^{n+m} \rangle_{\mathcal{Y}/W_{Y}} + \frac{1}{n+m+1} \frac{-K_{Y} \cdot (H_{Y}+kL)^{n+m-1}}{(H_{Y}+kL)^{n+m}} \langle \mathcal{L}_{k}^{n+m+1} \rangle_{\mathcal{Y}/W_{Y}},$$
(4.22)

where the constant

$$\widehat{S}_Y := \frac{-K_Y \cdot (H_Y + kL)^{n+m-1}}{(H_Y + kL)^{n+m}}$$

is the average scalar curvature of Y, and hence of each  $Y_t$ , with respect to the metric  $\omega + k\omega_B$ .

We use Proposition 4.23 to prove the following.

**Proposition 4.24.** There exists a line bundle  $\mathcal{D}(Y)$  on  $W_Y$  whose first Chern class is represented by the Weil-Petersson metric (4.14).

*Proof.* Let  $\mathcal{H} \to \mathcal{Y}$  be the relatively ample line bundle induced by each relative polarisation  $H_t \to Y_t$ . More precisely, we consider the fibration  $\mathcal{Y} \to W_Y$  as the composition of

$$\mathcal{Y} \to B \times W_Y \to W_Y$$

so  $\mathcal{H}$  is relatively ample with respect to the first projection. Moreover, since the base *B* is fixed, we can pull-back *L* to  $\mathcal{Y}$  and consider it as a line bundle on  $\mathcal{Y}$ . So we can define a relatively ample line bundle  $\mathcal{L}_k$  over  $\mathcal{Y}$  as

$$\mathcal{L}_k = \mathcal{H} \otimes kL, \tag{4.23}$$

whose fibre over  $Y_t$  is  $\mathcal{L}_k|_t = H_t \otimes kL$ . Its first Chern class contains the relative metric  $\omega_{\mathcal{Y},k}$ , where  $\widehat{\omega}$  is in  $c_1(\widehat{\mathcal{H}})$ . We define a line bundle  $\mathcal{D}(Y) \to W_Y$  by using the fibre integral formula (4.19) and the expansion in powers of k of the line bundle (4.22). Expanding in k the intersection product  $\mathcal{L}_k^{n+m}$ ,  $\mathcal{L}_k^{n+m+1}$  and the expression (4.22) we obtain that the Weil-Petersson metric  $\Omega_{WP}(\omega_{\mathcal{Y},k})$  represents the first Chern class of the line bundle  $\mathcal{D}(Y)$  given as the tensor product of the line bundles

$$\begin{aligned} \mathcal{D}_{0}(Y) &= -\langle -K_{\mathcal{Y}/B \times W_{Y}}, \mathcal{H}^{m}, L^{n} \rangle_{\mathcal{Y}/W_{Y}}, \\ \mathcal{D}_{2}(Y) &= \frac{1}{n+m+1} \frac{-K_{Y/B} \cdot L^{n} \cdot H_{Y}^{m-1}}{L^{n} \cdot H_{Y}^{m}} \left( \langle \widehat{\mathcal{H}}^{m+1}, L^{n} \rangle_{\mathcal{Y}/W_{Y}} + \langle \widehat{\mathcal{H}}^{m+2}, L^{n-1} \rangle_{\mathcal{Y}/W_{Y}} \right) \\ \mathcal{D}_{3}(Y) &= -\langle -K_{B}, \widehat{\mathcal{H}}^{m+1}, L^{n-1} \rangle_{\mathcal{Y}/W_{Y}}, \\ \mathcal{D}_{4}(Y) &= \frac{1}{n+m+1} \frac{\left( -K_{Y/B} \cdot H_{Y}^{m-1} \cdot L^{n} \right) \left( H_{Y}^{m+1} \cdot L^{n-1} \right)}{H_{Y}^{m} \cdot L^{n}} \langle \widehat{\mathcal{H}}^{m+1}, L^{n} \rangle_{\mathcal{Y}/W_{Y}}, \end{aligned}$$

defined using the Deligne pairing. Indeed, by expanding the expression (4.22) in powers of k we see that the sum of the leading order term and the sub-leading order term is given by

$$\mathcal{D}_0(Y) + \mathcal{D}_1(Y) + \mathcal{D}_2(Y) + \mathcal{D}_3(Y) + \mathcal{D}_4(Y),$$

where  $\mathcal{D}_1(Y)$  is given by

$$\mathcal{D}_1(Y) = -\langle -K_{\mathcal{Y}/B \times W_Y}, \widehat{\mathcal{H}}^{m+1}, L^{n-1} \rangle_{\mathcal{Y}/W_Y}.$$

However, its first Chern class is represented by the term (4.18), which does not appear in the fibre integral formula for the relative Weil-Petersson metric of Lemma 4.19. This concludes the proof.  $\hfill \Box$ 

We have constructed a line bundle  $\mathcal{D}_Y$  on  $W_Y$  whose first Chern class is the Weil-Petersson metric  $\Omega_{WP}(\omega_{\mathcal{Y},k})$ . Let now  $p: W_Y \to \mathcal{M}$  be the composition of the maps

$$W_Y \to W_Y / \operatorname{Aut}(\pi_Y) \hookrightarrow \mathcal{M},$$

where the first map is the quotient by the group action and the second map is the inclusion of a local chart in the moduli space. It follows from Lemma 4.14 that the orbits of the Aut( $\pi_Y$ )-action on  $W_Y$  correspond to isomorphic manifolds in the family  $\mathcal{Y} \to W_Y$ . Therefore the line bundle  $\mathcal{D}(Y)$  is invariant for the action of Aut( $\pi_Y$ ) and thus descends to a line bundle  $\widetilde{\mathcal{D}}(Y)$  on the quotient  $W_Y$ /Aut( $\pi_Y$ ).

**Theorem 4.25.** There exists a line bundle  $\mathcal{F}$  on the moduli space  $\mathcal{M}$  such that its restriction to each chart  $W_{\Upsilon}/\operatorname{Aut}(\pi_{\Upsilon})$  it is isomorphic to  $\widetilde{\mathcal{D}}(\Upsilon)$ .

*Proof.* Let  $W_{Y_1}/\operatorname{Aut}(\pi_{Y_1})$  and  $W_Y/\operatorname{Aut}(\pi_Y)$  be two local charts of  $\mathcal{M}$  with non empty intersection. Then using completeness of the Kuranishi space there exists an isomorphism  $\tilde{\alpha} : W_{Y_1}/\operatorname{Aut}(\pi_{Y_1}) \to W_Y/\operatorname{Aut}(\pi_Y)$  (4.12) which preserves the relative polarisation (4.23) and the submersions onto the base B. By functoriality of the Deligne pairings, the pull-back  $\tilde{\alpha}^* \widetilde{\mathcal{D}}(Y_1)$  is then isomorphic to  $\widetilde{\mathcal{D}}(Y)$ . Therefore, on the intersection of  $W_Y/\operatorname{Aut}(\pi_Y)$  and  $W_{Y_1}/\operatorname{Aut}(\pi_{Y_1})$  there is a morphism of line bundles  $\chi : \widetilde{\mathcal{D}}(Y_1) \xrightarrow{\sim} \widetilde{\mathcal{D}}(Y)$ . Let  $\varphi_Y : \widetilde{\mathcal{D}}(Y) \to W_Y/\operatorname{Aut}(\pi_Y) \times \mathbb{C}$  and  $\varphi_{Y_1} : \widetilde{\mathcal{D}}(Y_1) \to W_{Y_1}/\operatorname{Aut}(\pi_{Y_1}) \times \mathbb{C}$  be local trivialisations and, on the intersection,

$$\psi_{Y_1Y} := \varphi_Y \circ \chi \circ \varphi_{Y_1}^{-1}.$$

The map  $\psi_{Y_1Y}$ , viewed as a function on  $\mathbb{C}$  is invertible, with inverse  $\psi_{YY_1}$ . Indeed, the map  $\chi$  is an isomorphism because  $\tilde{\alpha}$  is. The same argument proves that the cocycle condition holds.

The following corollary is a consequence of Theorem 4.25 and of [31, Proposition 1.7].

**Corollary 4.26.** Any compact analytic subspace of  $\mathcal{M}$  is projective.

## Bibliography

- [1] ARBARELLO, E., CORNALBA, M., AND GRIFFITHS, P. A. *Geometry of algebraic curves: volume* II, with a contribution by Joseph Daniel Harris. Springer, 2011. 74
- [2] BERMAN, R. J., DARVAS, T., AND LU, C. H. Regularity of weak minimizers of the K-energy and applications to properness and K-stability. *Annales scientifiques de l'École normale supérieure* (2016). i
- [3] BESSE, A. L. *Einstein manifolds*. Springer Science & Business Media, 2007. 72
- [4] BIALYNICKI-BIRULA, A. Some theorems on actions of algebraic groups. Annals of mathematics 98, 3 (1973), 480–497. v, 63
- [5] BOUCKSOM, S., HISAMOTO, T., AND JONSSON, M. Uniform K-stability and asymptotics of energy functionals in Kähler geometry. *Journal of the European Mathematical Society* 21, 9 (2019), 2905–2944. 80
- [6] BRÖNNLE, T. *Deformation constructions of extremal metrics*. PhD thesis, Imperial College London, 2011. v, 15
- [7] BRÖNNLE, T. Extremal Kähler metrics on projectivized vector bundles. Duke Mathematical Journal 164, 2 (2015), 195–233. iv, 58
- [8] CANNAS DA SILVA, A. Lectures on symplectic geometry, vol. 3575. Springer, 2001. 14, 34
- [9] CHEN, X., DONALDSON, S., AND SUN, S. Kähler-Einstein metrics on Fano manifolds. I: Approximation of metrics with cone singularities. *Journal of the American Mathematical Society 28*, 1 (2015), 183–197. i
- [10] CHEN, X., DONALDSON, S., AND SUN, S. Kähler-Einstein metrics on Fano manifolds. II: Limits with cone angle less than  $2\pi$ . *Journal of the American Mathematical Society 28*, 1 (2015), 199–234. i
- [11] CHEN, X., DONALDSON, S., AND SUN, S. Kähler-Einstein metrics on Fano manifolds. III: Limits as cone angle approaches  $2\pi$  and completion of the main proof. *Journal of the American Mathematical Society 28*, 1 (2015), 235–278. i
- [12] CHEN, X., AND SUN, S. Calabi flow, geodesic rays, and uniqueness of constant scalar curvature Kähler metrics. Annals of Mathematics (2014), 407–454. 16, 35, 36, 37, 38, 62
- [13] DERVAN, R., AND NAUMANN, P. Moduli of polarised manifolds via canonical Kähler metrics. arXiv preprint arXiv:1810.02576 (2018). i, 19, 20
- [14] DERVAN, R., AND SEKTNAN, L. M. Optimal symplectic connections on holomorphic submersions. *Communications on Pure and Applied Mathematics* (2019). ii, iii, iv, vii, 24, 25, 26, 27, 29, 41, 42, 47, 48, 50, 53, 56, 58

- [15] DERVAN, R., AND SEKTNAN, L. M. Extremal metrics on fibrations. Proceedings of the London Mathematical Society 120, 4 (2020), 587–616. 50, 58, 59
- [16] DERVAN, R., AND SEKTNAN, L. M. Moduli theory, stability of fibrations and optimal symplectic connections. *Geometry & Topology* 25, 5 (2021), 2643–2697. vi
- [17] DERVAN, R., AND SEKTNAN, L. M. Uniqueness of optimal symplectic connections. In Forum of Mathematics, Sigma (2021), vol. 9, Cambridge University Press. vi, 25, 46, 48
- [18] DONALDSON, S. K. Anti self-dual Yang-Mills connections over complex algebraic surfaces and stable vector bundles. *Proceedings of the London Mathematical Society 3*, 1 (1985), 1–26. i
- [19] DONALDSON, S. K. Remarks on gauge theory, complex geometry and 4-manifold topology. *Fields Medallists' lectures, World Scientific Series in 20th Century Mathematics* (1997), 384–403. 7, 9
- [20] DONALDSON, S. K. Symmetric spaces, Kähler geometry and Hamiltonian dynamics. Northern California Symplectic Geometry Seminar, Translations of the American Mathematical Society-Series 2 196 (1999), 13–33. 11, 13
- [21] DONALDSON, S. K. Scalar curvature and stability of toric varieties. *Journal of Differential Geometry* 62, 2 (2002), 289–349. i, i
- [22] ELKIK, R. Fibrés d'intersections et intégrales de classes de Chern. In Annales scientifiques de l'École Normale Supérieure (1989), vol. 22, pp. 195–226. 80
- [23] ELKIK, R. Métriques sur les fibrés d'intersection. Duke Mathematical Journal 61 (1990), 303–328. 80
- [24] FINE, J. Constant scalar curvature Kähler metrics on fibred complex surfaces. Journal of Differential Geometry 68, 3 (2004), 397–432. iv, iv, 50, 58, 59, 60
- [25] FINE, J. Constant scalar curvature metrics on fibred complex surfaces. PhD thesis, Imperial College London, 2004. vii
- [26] FINE, J. Fibrations with constant scalar curvature Kähler metrics and the CM-line bundle. *Mathemathical Research Letters* 14, 2 (2007), 239–247. 42
- [27] FUJIKI, A. Coarse moduli space for polarized compact Kähler manifolds. Publications of the Research Institute for Mathematical Sciences 20, 5 (1984), 977–1005. 73, 75
- [28] FUJIKI, A. Moduli space of polarized algebraic manifolds and Kähler metrics. Sugaku Expositions 5, 2 (1992), 173–191. 7, 9
- [29] FUJIKI, A., AND SCHUMACHER, G. The moduli space of Hermite-Einstein bundles on a compact Kähler manifold. *Proceedings of the Japan Academy, Series A* 63 (1987). i
- [30] FUJIKI, A., AND SCHUMACHER, G. The moduli space of Kähler structures on a real compact symplectic manifold. *Publications of the Research Institute for Mathematical Sciences* 24, 1 (1988), 141–168. 73, 75
- [31] FUJIKI, A., AND SCHUMACHER, G. The moduli space of extremal compact Kähler manifolds and generalized Weil-Petersson metrics. *Publications of the Research Institute for Mathematical Sciences* 26, 1 (1990), 101–183. i, 19, 20, 61, 73, 75, 79, 80, 82
- [32] GAUDUCHON, P. Calabi's extremal Kähler metrics: An elementary introduction. Preprint 34 (2010), 37–40. 3

- [33] GREB, D., SIBLEY, B., TOMA, M., AND WENTWORTH, R. Complex algebraic compactifications of the moduli space of Hermitian Yang-Mills connections on a projective manifold. *Geometry & Topology* 25, 4 (2021), 1719–1818. i
- [34] HALLAM, M. Geodesics in the space of relatively Kähler metrics. Journal of the London Mathematical Society 108, 3 (2023), 1036–1081. vi, vi, 24, 67
- [35] HASHIMOTO, Y. Existence of twisted constant scalar curvature Kähler metrics with a large twist. *Mathematische Zeitschrift* 292, 3 (2019), 791–803. 50
- [36] HASHIZUME, K., AND HATTORI, M. On boundedness and moduli spaces of K-stable Calabi-Yau fibrations over curves. *preprint: arXiv:2305.01244* (2023). vi
- [37] HATTORI, M. On K-stability of Calabi-Yau fibrations. arXiv preprint arXiv:2203.11460 (2022). vi
- [38] HATTORI, M. On fibration stability after Dervan-Sektnan and singularities. Osaka Journal of Mathematics 61, 2 (2024), 195–228. vi, vi
- [39] HONG, Y.-J. Constant Hermitian scalar curvature equations on ruled manifolds. *Journal* of *Differential Geometry* 53, 3 (1999), 465–516. ii, iv
- [40] HONG, Y.-J. Stability and existence of critical Kaehler metrics on ruled manifolds. *Journal of the Mathematical Society of Japan 60*, 1 (2008), 265–290. vii
- [41] HORIKAWA, E. On deformations of holomorphic maps I. Journal of the Mathematical Society of Japan 25, 3 (1973), 372–396. 32
- [42] HORIKAWA, E. On deformations of holomorphic maps II. *Journal of the Mathematical* Society of Japan 26, 4 (1974), 647–667. 32
- [43] HOSKINS, V. Moduli problems and geometric invariant theory. *Lecture notes* 2016 7 (2015), 28. 6
- [44] HUYBRECHTS, D. Complex geometry: an introduction. Springer Science & Business Media, 2006. 17, 32, 35
- [45] INOUE, E. The moduli space of Fano manifolds with Kähler-Ricci solitons. Advances in Mathematics 357 (2019), 106841. i, 17, 18, 19
- [46] JELISIEJEW, J., AND SIENKIEWICZ, Ł. Białynicki-Birula decomposition for reductive groups. Journal de Mathématiques Pures et Appliquées 131 (2019), 290–325. v, 63
- [47] KEMPF, G., AND NESS, L. The length of vectors in representation spaces. In Algebraic Geometry: Summer Meeting, Copenhagen, August 7–12, 1978 (1979), Springer, pp. 233– 243. 7
- [48] KOBAYASHI. Curvature and stability of vector bundles. *Proc. Japan Acad. 58* (1982), 158–162. i
- [49] KURANISHI, M. New proof for the existence of locally complete families of complex structures. In *Proceedings of the conference on complex analysis* (1965), Springer, pp. 142– 154. 16, 17, 72
- [50] LICHNÉROWICZ, A. Géométrie des groupes de transformations. Travaux et recherches mathématiques. Dunod, 1958. 3
- [51] LU, Z., AND SEYYEDALI, R. Extremal metrics on ruled manifolds. Advances in Mathematics 258 (2014), 127–153. vii

- [52] LÜBKE, M. Stability of einstein-hermitian vector bundles. *manuscripta mathematica* 42, 2-3 (1983), 245–257. i
- [53] MARSDEN, J., AND WEINSTEIN, A. Reduction of symplectic manifolds with symmetry. *Reports on mathematical physics* 5, 1 (1974), 121–130. 5
- [54] MATSUSHIMA, Y. Sur la structure du groupe d'homéomorphismes analytiques d'une certaine variété kaehlérinne. *Nagoya Mathematical Journal 11* (1957), 145–150. 3
- [55] McCARTHY, J. B. Canonical metrics on holomorphic fibre bundles. *arXiv preprint arXiv*:2202.11630 (2022). iv
- [56] McDuff, D., AND SALAMON, D. Introduction to symplectic topology. Oxford University Press, 2017. 23
- [57] MEYER, K. R. Symmetries and integrals in mechanics. In *Dynamical systems*. Elsevier, 1973, pp. 259–272. 5
- [58] MUMFORD, D. Stability of projective varieties. L'Enseignement mathematique, Universite de Geneve, 1977. i
- [59] MUMFORD, D., FOGARTY, J., AND KIRWAN, F. Geometric invariant theory, vol. 34. Springer Science & Business Media, 1994. 5, 7
- [60] NAGATA, M. Invariants of group in an affine ring. Journal of Mathematics of Kyoto University 3, 3 (1963), 369–378. 5
- [61] NARASIMHAN, M. S., AND SESHADRI, C. S. Stable and unitary vector bundles on a compact Riemann surface. *Annals of Mathematics* (1965), 540–567. i
- [62] NEWSTEAD, P. E. Characteristic classes of stable bundles of rank 2 over an algebraic curve. *Transactions of the American Mathematical Society* 169 (1972), 337–345. i
- [63] NEWSTEAD, P. E. Introduction to moduli problems and orbit spaces. T.I.F.R. Lecture Notes, Springer-Verlag (1978). i, 5, 6
- [64] NICOLAESCU, L. An invitation to Morse theory. Springer Science & Business Media, 2011.
   17
- [65] ORTU, A. The analytic moduli space of holomorphic submersions. *preprint arXiv*:2306.09137 (2023). viii
- [66] ORTU, A. Optimal symplectic connections and deformations of holomorphic submersions. *Advances in Mathematics* 414 (2023), 108868. viii
- [67] ORTU, A. Moment maps and stability of holomorphic submersions. *arXiv preprint arXiv*:2407.03246 (2024). 19
- [68] SCARPA, C. *The Hitchin-cscK system*. PhD thesis, SISSA International School of Advanced Studies, Trieste, 2020. Available at arXiv:2010.07728. 7
- [69] SCARPA, C., AND STOPPA, J. Scalar curvature and an infinite-dimensional hyperkähler reduction. *Asian Journal of Mathematics* (2020). 47
- [70] SCHUMACHER, G. Moduli of polarized kähler manifolds. Mathematische Annalen 269, 1 (1984), 137–144. 32
- [71] SEKTNAN, L. M., AND SPOTTI, C. Extremal metrics on the total space of destabilising test configurations. *Mathematische Annalen* 202 (2023). vii

- [72] SESHADRI, C. S. Space of unitary vector bundles on a compact Riemann surface. Annals of Mathematics (1967), 303–336. i
- [73] SJAMAAR, R., AND LERMAN, E. Stratified symplectic spaces and reduction. Annals of Mathematics (1991), 375–422. 5
- [74] STOPPA, J. K-stability of constant scalar curvature Kähler manifolds. Advances in Mathematics 221(4) (2009), 1397–1408. i
- [75] STOPPA, J. Twisted constant scalar curvature Kähler metrics and Kähler slope stability. *Journal of Differential Geometry* 83, 3 (2009), 663–691. 50
- [76] SUN, S. Degenerations and moduli spaces in Kähler geometry. In *Proceedings of the International Congress of Mathematicians: Rio de Janeiro 2018* (2018), World Scientific, pp. 993–1012. i
- [77] SZÉKELYHIDI, G. The Kähler-Ricci flow and K-polystability. American Journal of Mathematics 132, 4 (2010), 1077–1090. v, 15, 16, 17, 19
- [78] SZÉKELYHIDI, G. An introduction to extremal Kähler metrics, vol. 152. American Mathematical Soc., 2014. 2, 11
- [79] TIAN, G. Kähler-Einstein metrics with positive scalar curvature. *Inventiones Mathematicae* 130, 1 (1997), 1–37. i, 13
- [80] UHLENBECK, K., AND YAU, S.-T. On the existence of Hermitian-Yang-Mills connections in stable vector bundles. *Communications on Pure and Applied Mathematics* 39, S1 (1986), S257–S293. i
- [81] YAU, S.-T. Open problems in geometry. In *Proceedings of Symposia in Pure Mathematics* (1993), vol. 54, pp. 1–28. i